# **Integrated Corridor Management**

# Analysis, Modeling, and Simulation for the I-15 Corridor in San Diego, California

Post-Deployment Assessment Report

www.its.dot.gov/index.htm

Final Report — December 2016 FHWA-JPO-16-403





Produced by Cambridge Systematics, Inc. U.S. Department of Transportation Office of the Assistant Secretary for Research and Technology

## Notice

This document is disseminated under the sponsorship of the Department of Transportation in the interest of information exchange. The United States Government assumes no liability for its contents or use thereof. The U.S. Government is not endorsing any manufacturers, products, or services cited herein and any trade name that may appear in the work has been included only because it is essential to the contents of the work.

(Cover Image by San Diego Association of Governments.)

**Technical Report Documentation Page** 

	-					
	2. Government Accession No.	3. Recipient's Catalog No.				
FHWA-JPO-16-403	/A-JPU-16-403					
4. Title and Subtitle	5. Report Date					
Integrated Corridor Management	December 2016					
Analysis, Modeling, and Simulati	6. Performing Organization Code					
California – Post-Deployment As						
7. Author(s)	8. Performing Organization Report No.					
Vassili Alexiadis, Alice Chu						
9 Performing Organization Name And Addre	295	10. Work Unit No. (TRAIS)				
Cambridge Systematics. Inc.						
555 12 <sup>th</sup> Street Suite 1600		11. Contract or Grant No.				
Oakland California 94607		DTFH61-12-D-00042				
12. Sponsoring Agency Name and Address		13. Type of Report and Period Covered				
U.S. Department of Transportation	on	Final Report				
ITS Joint Program Office						
1200 New Jersey Avenue, SE	14. Sponsoring Agency Code					
Washington, D.C. 20590	НОР					
15. Supplementary Notes		<u> </u>				
The Government Task Manager for this task is James Colyar.						
16. Abstract						
The U.S. Department of Transportation Integrated Corridor Management (ICM) Initiative aims to advance the state of						
the practice in transportation corridor operations to manage congestion. Through the deployment of ICM at the two						
selected Demonstration Sites (Dallas, Texas and San Diego, California), this initiative thoroughly investigated and						
documented the impacts of the ICM deployments, especially in regards to improved agency coordination. Analysis,						
Modeling and Simulation (AMS) efforts assisted corridor partners to optimize their ICM deployment, and supported the						
broader evaluation effort for the entire ICM Initiative. Using AMS enabled corridor partners to identify the strategies to						
include in their ICM System that would be most effective against their specific corridor congestion issues by providing						
measureable results for multiple alternatives. The focus of this ICM Post-Deployment assessment is to investigate the						
impacts of the ICM system in its	"as denloved" state on 1-15 in San Diego, usi	ng AMS tools and techniques developed				
impacts of the row system in its as deployed state on 1-15 in San Diego, using AiviS tools and techniques developed						

and refined under both the current and previous phases of the program.

The localized ICM strategies deployed include an active decision support system, coordinated incident management, freeway coordinated ramp metering, improved multi-modal traveler information, upgrades to selected traffic signal systems, and active arterial routing. A framework of the key activities required for post-deployment AMS, namely model enhancements, model calibration and validation, cluster analysis and incident matching, and alternatives analysis, is presented. Mobility performance results for the site's peak directions of travel indicate an expected annual savings of 267,850 person hours, while expected cumulative annual variability improvements amounted to 188,816 hours.

17. Key Words		18. Distribution Statement			
Integrated Corridor Management, ICM, post-		No Restrictions.			
deployment analysis, incident, strategies, traveler					
information, pioneer corridor, analysis plan, Analysis					
Modeling Simulation (AMS) performance measures					
San Diego, California, CA, I-15					
19. Security Classif. (of this report)	20. Security Clas	ssif. (of this page)	21. No. of Pages	22. Price	
Unclassified	Unclassified		95	NA	
	1			<u>.</u>	

Form DOT F 1700.7 (8-72)

Reproduction of completed page authorized

# Acknowledgements

The following individuals provided significant support in the development of this document:

- San Diego Site Team Alex Estrella, Peter Thompson, Mike Calandra, and Matthew Juckes (contractor to San Diego Association of Governments).
- U.S. DOT Integrated Corridor Management (ICM) Team Robert Sheehan, James Colyar, Steve Mortensen, Brian Cronin and Karl Wunderlich (contractor to the U.S. Department of Transportation).
- Volpe Center Evaluation Team Lee Biernbaum and Margaret Petrella.
- ICM Evaluation Contractor Team –Michael Waisley and Murat Omay (contractors to U.S. Department of Transportation).

# **Table of Contents**

Executive Summary1
Key Findings
LESSONS LEARNED
Chapter 1. Introduction and Background4
Chapter 2. I-15 Corridor Description
Chapter 3. Integrated Corridor Management System and Response
Plans13
Chapter 4. Integrated Corridor Management Strategies
Chapter 5. Analysis, Modeling and Simulation Methodology21
MODELING COMPONENTS
MODELING INTEGRATED CORRIDOR MANAGEMENT STRATEGIES
Chapter 6. Performance Measures
- Мовіціту
RELIABILITY AND VARIABILITY OF TRAVEL TIME
OTHER MEASURES
SUMMARY OF PERFORMANCE MEASURES
Chapter 7. Post-Deployment Analysis, Modeling and Simulation
Approach
Model Enhancements
SUMMARY OF ANALYSIS SETTINGS
MODEL CALIBRATION & REASONABLENESS ASSESSMENT
Chapter 8. Alternatives Analysis and Results
ANALYSIS SCENARIOS
TRAVELER SURVEY RESULTS
ANALYSIS RESULTS
BENEFITS OF ANALYSIS, MODELING, AND SIMULATION73
References
APPENDIX A. List of Acronyms77
APPENDIX B. Performance Measure Calculation Using Simulation.79

# List of Tables

Table 1. Modeling assumptions regarding awareness, use, and compliance
to traveler information
Table 2. Summary of performance measure categories and operational
characteristics for analysis
Table 3. San Diego I-15 corridor—summary of post-deployment analysis
settings
Table 4. Reasonableness Assessment criteria and acceptance targets 39
Table 5. Count comparison for all I-15 locations above VPH threshold –
typical day40
Table 6. Summary of all clusters for all time periods and both directions,
ordered by total cluster delay impact52
Table 7. Incident matching criteria. 55
Table 8. Summary of representative days/incidents in the most
frequent/impactful clusters during the AM and PM peak periods55
Table 9. Aggregated percent error of modeled vs. observed volume data for
each cluster-representative day57
Table 10. I-15 pulse survey results – changes in travel plans based on pre-
trip information60
Table 11. I-15 pulse survey results – changes in travel plans based on en-
route information61
Table 12. Mobility performance measures – daily Person Hours Traveled 65
Table 13. Mobility performance measures – traveler breakdown when
Express Lanes were under normal operations during hypothetical
scenario
Table 14. Mobility performance measures – daily Person Miles Traveled 67
Table 15. Aggregated mobility, reliability, and variability performance results
by direction & time period
Table 16. Travel time-based analysis results71

# List of Figures

Figure 1. Flowchart. Integrated Corridor Management implementation
process5
Figure 2. Map. Location and geographic boundaries of corridor 11
Figure 3. Map. Study area I-15 corridor in San Diego, California
Figure 4. Diagram. San Diego I-15 Integrated Corridor Management system
design13
Figure 5. Diagram. San Diego I-15 Integrated Corridor Management system
inputs and outputs 14
Figure 6. Diagram. I-15 Integrated Corridor Management response
postures
Figure 7. Diagram. I-15 Decision Support System defined subsystem action
plans
Figure 8. Diagram. I-15 Decision Support System multi-modal response
plans
Figure 9. Illustration. Example response plan for northbound afternoon peak
period congestion at Rancho Bernardo Rd
Figure 10. Map. I-15 responsive signal operations arterial groups
Figure 11. Illustration. The Integrated Corridor Management Analysis,
Modeling and Simulation Methodology blends up to three classes of
modeling tools for comprehensive corridor-level modeling and
analysis
Figure 12. Map. Model network I-15 for post-deployment Analysis. Modeling.
and Simulation
Figure 13. Heatmap. Southbound I-15 AM observed (top) versus modeled
(bottom) speed contours – typical day
Figure 14. Heatmap, Northbound I-15 AM observed (top) versus modeled
(bottom) speed contours – typical day
Figure 15. Heatmap, Southbound I-15 PM observed (top) versus modeled
(bottom) speed contours – typical day 45
Figure 16. Heatmap. Northbound I-15 PM observed (top) versus modeled
(bottom) speed contours – typical day
Figure 17. Heatmap. Southbound I-15 PM observed (top) versus modeled
(bottom) speed contours – incident condition 47
Figure 18. Heatmap, Northbound I-15 PM observed (top) versus modeled
(bottom) speed contours – incident condition 48
Figure 19 Illustration Operational condition darthoard categorized by time
period and event type 56
Figure 20 Chart 1-15 pulse survey results – AM peak period travel plan
changes 62
Figure 21 Chart I-15 pulse survey results – PM peak period travel plan
changes 62
Figure 22 Chart ICM impacts on travel time reliability – huffer time 68
Figure 23 Diagram Implementable response plan for peak direction
scenario NR PM 3 70

# **Executive Summary**

The objective of the *Integrated Corridor Management (ICM)* initiative is to demonstrate how intelligent transportation systems (ITS) technologies can efficiently and proactively manage the movement of people and goods in major transportation corridors. In the context of this ICM initiative, a "corridor" refers to a largely linear geographic band defined by existing and forecasted travel patterns involving both people and goods. The corridor serves a particular travel market (or markets) that are affected by similar transportation needs and mobility issues. The corridor includes various combinations of facility type and mode, also known as networks (e.g., limited access facilities, surface arterials, transit, bicycle, pedestrian pathways, waterways, etc.) that provide similar or complementary transportation functions. Additionally, the corridor includes cross-network connections that permit the individual networks to be readily accessible from each other. The ICM initiative aims to pioneer innovative multimodal and multijurisdictional strategies and combinations of strategies that optimize existing infrastructure to help manage recurring and nonrecurring congestion in our nation's corridors.

Through the deployment of ICM at the two selected Demonstration Sites (Dallas, Texas and San Diego, California), this initiative thoroughly investigated and documented the impacts of the ICM deployments. Analysis, Modeling and Simulation (AMS) efforts assisted corridor partners to optimize their ICM deployment and supported the broader evaluation effort for the entire ICM Initiative. Using AMS enabled corridor partners to identify the strategies to include in their integrated corridor management system (ICMS) that would be most effective against their specific corridor congestion issues, by providing measureable results for multiple alternatives. A key benefit of using AMS is its ability to produce system level assessments of mobility and environmental impacts that cannot be observed directly from field data.

The focus of this ICM Post-Deployment assessment is to evaluate to what extent ICM technologies can efficiently and proactively manage the movement of goods and people in a major transportation corridor. Specifically, this project investigates the impacts of the ICM system in its "as deployed" state on Interstate 15 (I-15) in San Diego, using AMS tools and techniques developed and refined under both the current and previous phases of the program. Results from traveler behavior surveys conducted in the vicinity of the I-15 corridor by the Volpe Center were used to inform model assumptions and to enable more accurate representation of true driver behaviors on I-15.

The San Diego ICM Demonstration site corridor covers a 21 mile section of I-15 from just north of State Route (SR) 52 in the City of San Diego to SR 78 in the City of Escondido. As one of two major freeways that connect commuters and interregional goods movement between San Diego, Orange and Riverside counties and people traveling to and from Mexico, the I-15 is one of the busiest sections of freeway in the region. The Corridor study area includes the freeway, ramp metered interchanges, 20 miles of continuous Express Lanes (otherwise known as managed lanes) – 16 miles which are reversible, a bus rapid transit (BRT) line that runs on the I-15 Express Lanes, BRT stations, direct access ramps, major arterial streets, and ITS technologies.

Specific examples of practices the San Diego site team employed include:

- Provided corridor users with the operational condition of all corridor networks and components, such as travel times, incident information, and expected delays using changeable message signs (CMS), a new 511 app, and other commercial travel time information sources.
- Used a decision support system with real-time simulation, predictive algorithms, and analysis to evaluate potential congestion mitigation and select/implement the optimal combination of mitigation strategies for the corridor.
- Established, improved, and automated joint agency action plans for traveler information, traffic signal timing, ramp metering, transit and Express Lanes (or managed lanes).
- Identified means of enhancing corridor management across all networks, including shared control multi-jurisdictional coordination of field devices such as lane controls, traveler information messages, traffic signal timing plans, and transit priority.

The ICM strategies implemented in the "as-deployed" ICM system, which were replicated in the models used for post-deployment AMS include:

- Active Decision Support System;
- Coordinated incident management;
- Freeway coordinated ramp metering;
- Actionable traveler information (en-route and pre-trip via CMS, a new 511 app, and other commercial sources);
- Upgrades to selected traffic signal systems (new traffic signal coordination timings, responsive traffic signal control); and
- Alternate route wayfinding signs.

The AMS serves to assess the performance of various components of the ICM system under different operational conditions (e.g., time of day, direction of traffic, duration until the incident was cleared, etc.). Cluster analysis was used to group together workday travel characteristics between March 1, 2012 and February 28, 2013 in days where operational conditions were more similar to each other, than to those in other groups (clusters). Clusters were prioritized based on the total magnitude of expected incident impact for representative days in each cluster. The clusters in the northbound PM peak and southbound AM peak periods were of primary interest for ICM AMS because they reflect the periods of highest traffic demands. Field observed incidents that occurred in the year after ICM deployment were matched to high-impact clusters sharing similar operational conditions. Eight scenarios, representing the top eight high-impact clusters were analyzed "with" and "without" ICM. One additional hypothetical scenario was also analyzed in order to evaluate the impact of one of the managed lane strategies (i.e., opening the Express Lanes to all travelers in the event of a severe incident).

# **Key Findings**

Overall, the I-15 corridor post-deployment AMS results show consistent travel time improvements in the two peak directions as a result of ICM implementation. The weighted average of travel time beneficiaries in the northbound PM aggregated scenario is +2.65%; in the southbound PM aggregated scenario the weighted average of travel time beneficiaries is +4.01%. For the two peak directions combined (southbound AM and northbound PM), the expected daily travel time

savings are 1,403 person hours of travel; expected annual savings are 267,850 person hours of travel.

The post-deployment AMS results for the I-15 corridor generally show travel time reliability and variability improvements during both peak directions; the expected cumulative annual travel time variability improvement is 188,816 hours.

A hypothetical AMS exercise examined the potential benefit of opening the Express Lanes to all traffic during a severe incident. No travel time benefits were found in AMS resulting from this potential action for the scenario analyzed. Overall, in six out of the eight scenarios (excluding the hypothetical scenario), more travelers benefited from ICM, compared to the ones who did not.

A key benefit of using AMS is its ability to focus on system level assessments of mobility and environmental impacts that cannot be observed directly from field data; this information will be used to support of the broader Integrated Corridor Management evaluation effort. The scenarios used in alternatives analysis were chosen to fill in missing elements where observed data could not be used to make meaningful condition-to-condition comparisons, as well as to calculate unobservable systemlevel impacts.

## **Lessons Learned**

The ICM methodology encourages transportation professionals to manage the transportation corridor as a multimodal system, as opposed to managing individual assets. The San Diego ICM demonstration involved the coordination of operations along the I-15 corridor, including increased communication and coordination among partner agencies, facilitated by the deployment of an interagency-dependent decision support system (DSS). AMS aids in the broader goals of ICM Evaluation by providing a framework that can be used to quantify potential and actual benefits of localized ICM strategies. Unlike traditional corridor studies, which often focus on a specific element of a corridor, ICM AMS is a comprehensive approach that analyzes different operational conditions across time and modes and across a large enough geographic area to absorb all impacts.

One major benefit of the ICM AMS methodology is that it instigated the use of performance measures to inform and refine the response plans. This allowed AMS to provide insights through measurable results, a major factor that can help agencies determine which transportation investments are worthwhile. AMS allows agencies to "see around the corner", producing simulations of possible future conditions, allowing agencies to react proactively. AMS offers the flexibility of trying different combinations of traffic mitigation strategies, opening up an envelope of potential benefits. Transportation professionals can integrate the AMS methodology with ICM decision support systems to facilitate predictive, real-time, and scenario-based operational decision-making. Overall, this helps agencies create a better, more informed product.

# Chapter 1. Introduction and Background

The objective of the Integrated Corridor Management (ICM) initiative is to demonstrate how intelligent transportation systems (ITS) technologies can efficiently and proactively manage the movement of people and goods in major transportation corridors. In the context of this ICM initiative, a "corridor" refers to a largely linear geographic band defined by existing and forecasted travel patterns involving both people and goods. The corridor serves a particular travel market (or markets) that are affected by similar transportation needs and mobility issues. The corridor includes various combinations of facility type and mode, also known as networks (e.g., limited access facilities, surface arterials, transit, bicycle, pedestrian pathways, waterways, etc.) that provide similar or complementary transportation functions. Additionally, the corridor includes cross-network connections that permit the individual networks to be readily accessible from each other. The ICM initiative aims to pioneer innovative multimodal and multijurisdictional strategies and combinations of strategies that optimize existing infrastructure to help manage both recurring and nonrecurring congestion in our nation's corridors. There are many corridors in the country with underutilized capacity in the form of additional transit capacity-bus, rail, bus rapid transit (BRT), etc.-under saturated parallel arterials, and inefficient utilization of principal facility resources. Each of these corridors could benefit from the application of ICM technologies and strategies.

The maturation of ITS technologies, growing availability of supporting data, and emerging multiagency institutional frameworks make ICM both practical and feasible. Several freeway, arterial, and transit optimization strategies are in widespread use across the United States, with most currently managed by individual local agencies on an asset-by-asset basis. For those that are managed by a larger regional agency, the approach is still generally uncoordinated and involves little or no integration among the different resources available on the corridor. By appropriately applying ICM strategies, the agencies responsible for managing these corridors can reduce severe congestion and improve overall productivity. Furthermore, providing travelers with relevant information on transportation alternatives can encourage a redistribution of trips to less congested routes, modes, or times of day, which further reduces congestion and affords travelers a greater mobility and increased safety.

Through the deployment of ICM at the two selected Demonstration Sites (Dallas, Texas and San Diego, California), this initiative thoroughly investigated and documented the impacts of the ICM deployments, especially in regards to improved agency coordination. Getting as many corridor partners and stakeholders (e.g., roadway agencies, transit agencies, law enforcement, planning organizations, fleet operations, project evaluators, corridor travelers, etc.) involved in the design of the ICM from the very beginning adds significant value to the project—from adding precision to the design and informing travel demand modelers, to proactively addressing agency regulations. The role of Analysis, Modeling and Simulation (AMS) is to enable corridor partners to identify the strategies to include in their integrated corridor management system (ICMS) that will be most effective against their localized corridor congestion issues, by providing measureable results for multiple alternatives. The AMS methodology was applied to the ICM deployments in both Dallas and San Diego. A key benefit of using AMS is its ability to focus on system level assessments of mobility and environmental impacts

that cannot be observed directly from field data; this information will be used to support of the broader evaluation effort.

Based on the experience gained from the ICM deployments at the Dallas and San Diego Demonstration Sites, the ICM initiative developed an AMS methodology to assist corridor managers in forecasting and assessing the potential benefits and implications of ICM in their corridors of interest. The ICM AMS Guide has been incorporated into the Federal Highway Administration (FHWA) Traffic Analysis Toolbox (Volume XIII). The AMS approach is intended to be a flexible and iterative process adaptable to a wide variety of conditions, strategies, and situations. This flexibility is intended to provide practitioners with sufficient structure to enable a rigorous analysis suitable to complex strategies that at the same time is not so rigid as to limit the ability to restructure and rerun the analysis to address project contingencies as they occur. The AMS approach is designed to be implemented in conjunction with the ICM system development and design process and to provide a tool for continuous improvement of corridor performance as depicted in Figure 1. This ICM implementation process is generally representative of the Systems Engineering process followed by the ICM Demonstration Sites. Regular periodic conduct of ICM AMS also supports continuous improvement of the supporting ICM system, and the analysis tools themselves.



Figure 1. Flowchart. Integrated Corridor Management implementation process. (Source: Office of the Assistant Secretary for Research and Technology, ITS JPO.)

The United States Department of Transportation (US DOT) has published multiple reports throughout the ICM initiative which can be used as references to aid transportation professionals in implementing their own ICM projects. In addition to the subset of reports listed below which are currently available, reports covering analytical and institutional lessons learned and the broader Evaluation Report will also be published.

- "Integrated Corridor Management Analysis, Modeling and Simulation (AMS) Methodology"
- "Traffic Analysis Toolbox Volume III: Guidelines for Applying Traffic Microsimulation Modeling Software"
- "Traffic Analysis Toolbox Volume XIII: Integrated Corridor Management Analysis, Modeling, and Simulation Guide"
- "Integrated Corridor Management: Implementation Guide and Lessons Learned"

- "Operations and Maintenance Plan for the I-15, San Diego Integrated Corridor Management (ICM) Demonstration Project"
- "ICM Stage 2 Data Collection Plan for the I-15 Corridor in San Diego, California"
- "I-15 San Diego, California ICM AMS Analysis Plan"
- "U.S. 75 Dallas, Texas, ICM AMS Analysis Plan"
- "I-394 Minneapolis, Minnesota, ICM AMS Analysis Plan"
- "I-15 San Diego, California, Model Validation and Calibration Report"
- "U.S. 75 Dallas, Texas, Model Validation and Calibration Report"
- "I-394 Minneapolis, Minnesota ICM AMS Model Calibration and Validation Report"
- "Integrated Corridor Management Modeling Results Report: Dallas, Minneapolis, and San Diego"
- "Stage 3A Analysis, Modeling, and Simulation for the I-15 Corridor in San Diego, California: Final Pre-Deployment Analysis Plan"
- "Stage 3A Analysis, Modeling, and Simulation for the I-15 Corridor in San Diego, California: Pre-Deployment AMS Assessment Report"
- "Analysis, Modeling, and Simulation for the I-15 Corridor in San Diego, California: Post-Deployment Analysis Plan"
- "Integrated Corridor Management Initiative: Traveler Response Panel Survey San Diego"

As the AMS process continues in parallel with the ICM system development and design process, it is likely that new strategies, alternatives and scenarios will emerge that will need to be evaluated within the AMS process; therefore, the flexibility to foresee and account for several iterations of analysis is critical. The design process may reveal new strategies or alternatives that may need to be analyzed in the AMS, prompting modifications to the AMS structure. Likewise, the AMS process may reveal parts of the concept of operations that are unworkable or uncover opportunities that may be leveraged that result in changes to the ultimate ICM design.

The advanced analysis capabilities of the AMS approach provides practitioners with enhanced opportunities to conduct detailed alternatives analysis to identify optimal combinations of strategies and to test and refine how the strategies may be most optimally implemented. Due to the complexity and resources required of the AMS, this level of analysis is typically most appropriate in the later planning stages after the preliminary screening of alternatives has winnowed out a smaller set of strategies and alternatives to be evaluated. The AMS will often continue through the design phase—being used to fine-tune strategies in an iterative function as the realities of the design process progress or to assess the impacts of sequencing the improvements to identify the optimal deployment phasing of the strategies.

The focus of this ICM Post-Deployment assessment is to evaluate to what extent ICM technologies can efficiently and proactively manage the movement of goods and people in a major transportation corridor. Initially, the discussion on performance-driven corridor management among the participating ICM Pioneer Sites was focused on measures derived from observed data. In the AMS phase of the effort however, attention turned to producing comparable measures derived from the outputs of different traffic simulation tools. This enabled hypothetical scenarios to be modeled, testing the impacts of potential ICM strategies before implementation and therefore reducing the chance of very expensive missteps in implementation.

This project investigates the impacts of the ICM system in its "as deployed" state in 2014 and 2015 on Interstate 15 (I-15) in San Diego, using AMS tools and techniques developed and refined under both the current and previous phases of the program. Results from traveler behavior surveys conducted in the vicinity of the I-15 corridor by the Volpe Center were used to inform model assumptions and to enable more accurate representation of true driver behaviors on I-15. The results of the post-deployment AMS were then used to assess and validate the estimated impacts resulting from the ICM deployment on I-15 in San Diego.

The following is a summary of additional project objectives used to support these overall goals:

- Develop a post-deployment AMS Plan in collaboration with the ICM Demonstration Site staff to promote coordination of analysis efforts and coherent alignment of goals among this effort, the ICM Demonstration Site staff, and the ICM Evaluation team.
- Support the objective evaluation efforts of the Demonstration Site staff and enhance the ability of the modeling tools to accurately represent the deployed ICM strategies by identifying and facilitating improvements to AMS tools, techniques, and inputs.
- Manage the successful transition of modeling responsibilities from AMS Contractor to the ICM Demonstration Site staff and organizations, with workshops to promote the transfer of knowledge and technology.
- Support the integration of AMS tools and techniques into ongoing corridor management practices by the Demonstration Site staff.
- Provide technical documentation of AMS tool development, data sources, data processing methods, model calibration and validation procedures, and analysis techniques used to represent and evaluate ICM impacts.

One aspect of the ICM program is the enhancement of analytical techniques and tools to support ICM impact assessment. In an effort to advance ICM impact assessment, the main objective for the AMS team within the ICM Initiative was to refine AMS tools and strategies, assess the Pioneer Sites' data capabilities, conduct AMS for a subset of the ICM Pioneer Sites, and conduct pre- and post-demonstration evaluations using AMS tools.

The AMS methodology applied to the Dallas and San Diego Demonstration Sites were documented in FHWA's "Traffic Analysis Toolbox Volume XIII: Integrated Corridor Management Analysis, Modeling and Simulation Guide". This guide is used to assist corridor managers in forecasting and assessing the potential benefits and implications of ICM in their corridors of interest. The ICM AMS methodology is rooted firmly in the US DOT's established modeling guidelines and frameworks, as defined in the FHWA Traffic Analysis Toolbox. Unlike traditional corridor studies, which often focus on a specific element of a corridor (i.e., a freeway or freeway and frontage road during a specific time of day), ICM AMS is a comprehensive approach that analyzes different operational conditions across time and modes and across a large enough geographic area to absorb all impacts.

The following items outline the key roles of AMS in the ICM Program:

- Identifying when and where ICM strategies will be the most beneficial;
- Assists in forecasting and assessing the potential implications of ICM;
- Developing methodologies that support the process for continuous improvement;
- Supporting the ICM Evaluation;

- Enables agencies to understand system dynamics at the corridor level; and
- Developing the analytical capital within each site so that the analyses can be conducted on a regular basis to support ICM decision-making (either in planning mode or Decision Support System (DSS) mode).

The post-deployment scope of work for the AMS team includes:

- Project Management and Program Support
- Develop Analysis Plans
- Enhance Tools to Reflect As-Deployed Corridor Management
- Tool Calibration Reasonableness Assessment
- Conduct Post-Deployment Alternatives Analysis
- Post-Deployment AMS Assessment Reports and Briefings
- Support AMS Transfer to Site
- Update AMS Guide
- AMS Knowledge and Technology Transfer

This *Post-Deployment ICM AMS Assessment Report for the I-15 Corridor* outlines the core tasks associated with the realization of the project goals and objectives described earlier. The organization of this report is as follows:

- Chapter 2 provides a brief description of the I-15 Corridor in San Diego, California.
- **Chapter 3** provides an overview of the innovative I-15 ICMS and corresponding response plans.
- Chapter 4 describes the ICM strategies comprising the ICM deployment on the corridor.
- Chapter 5 describes the AMS methodology applied to the corridor.
- Chapter 6 describes the performance measures used in the AMS.
- **Chapter 7** details the post-deployment AMS approach to model enhancements and model calibration and validation.
- **Chapter 8** summarizes main findings from alternatives analysis and the significant benefits of AMS.

# **Chapter 2. I-15 Corridor Description**

The Interstate 15 (I-15) study corridor in San Diego, California, extends from State Route (SR) 52 at its southern end to SR 78 at its northern end, for an overall length of approximately 20 miles. Along this freeway corridor are arterials with the following interchanges with the freeway:

- Centre City Parkway;
- Pomerado Road;
- Rancho Bernardo Road;
- Camino Del Norte Road;
- Ted Williams Parkway (SR 56);
- Black Mountain Road; and
- Scripps Parkway.

The I-15 corridor in San Diego has been utilized as a test bed for various intelligent transportation system (ITS) strategies identified in consultation with the San Diego Association of Governments (SANDAG), the metropolitan planning organization who is responsible for transportation planning in the region and the lead for San Diego's Integrated Corridor Management (ICM) initiative, and other local stakeholders. The strategies incorporated into the deployed ICM system are described in greater detail in Chapter 4. The following sections provide an overview of the study corridor.

In San Diego, the I-15 freeway carries eight to 10 lanes of traffic and functions as an important link between the urban core of San Diego and suburban cities to the northeast, including Poway, San Diego, and Escondido, making it a heavily used commuter link between northern San Diego County and major employment centers to the south. It is one of three major north-south transportation corridors in San Diego County and is the principal inland route, serving local, regional, and interregional trips. The route is part of a major interregional goods movement corridor, as it connects Mexico to the south with Riverside County, San Bernardino County, and Las Vegas, Nevada, to the north. As of December 2011, average weekday traffic volumes ranged from 170,000 to 290,000 vehicles on the general purpose lanes of I-15, with approximately 20,000 additional vehicles using Express Lanes. The peak direction of travel is southbound in the morning and northbound in the afternoon. Public transportation along the corridor includes bus rapid transit (BRT) that runs on the I-15 Express Lanes, and local bus transit lines that run on the neighboring arterials.

Recent population and housing growth in southwestern Riverside County, one of the fastest growing areas in California, has resulted in significant interregional commuter travel into San Diego County on I-15. Likewise, increasing commercial traffic moving to and from the Otay Mesa and Inland Empire commercial vehicle operations gateways have significantly impacted traffic on the corridor. Due to geographic/land use constraints, a lack of contiguous parallel roadways, and the limited number of alternative routes, peak-period delays are further exacerbated by incidents, special events, and/or inclement weather. Future (year 2020) forecasts for the I-15 corridor indicate a 30 percent increase in

weekday traffic, which will result in even longer corridor delays and travel times. Corridor travel is anticipated to increase significantly in what are now non-peak travel directions.

Figure 2 and Figure 3 provide geographic context for the corridor and indicate the extent of the study area. Figure 2 displays three separate segments along the I-15 corridor: North Segment (pink), Middle Segment (orange) and South Segment (green), which indicates the phased construction of the Express Lanes.

In anticipation of the growing traffic congestion along the I-15 corridor, SANDAG, has continually worked with its regional partners to improve mobility along the corridor. These efforts, such as the opening of two-lane reversible Express Lanes in 2012 and continued construction of transit stations and direct access ramps (allowing buses and high-occupancy vehicles (HOV) to directly access the Express Lanes without yielding to traffic in the general purpose lanes), have been carried out in parallel with the design and implementation of the integrated corridor management system (ICMS). The Express Lanes are being operated in a 2+2 configuration except on Thursday mornings where it uses a 3 southbound + 1 northbound configuration. California Department of Transportation (Caltrans) and SANDAG are currently assessing the performance of the corridor to determine whether other weekdays warrant additional configuration changes. BRT operation in the Express Lanes helps increase traffic capacity, while dynamic variable pricing helps manage traffic flow. This ITS infrastructure, along with existing road sensors, cameras, dynamic ramp metering, adaptive control on roads parallel to the interstate, and systems which can disseminate incident information, made the I-15 corridor an ideal ICM test bed environment.



#### Figure 2. Map. Location and geographic boundaries of corridor.

(Source: San Diego I-15 Demonstration Integrated Corridor Management System PATH Report on Stage 3: Site Demonstration and Evaluation, UCB-ITS-PRR-2015-03, p. 15.)



Figure 3. Map. Study area I-15 corridor in San Diego, California. (Source: Scope and Summary: I-15 ICMS Corridor in San Diego, U.S. Department of Transportation.)

# Chapter 3. Integrated Corridor Management System and Response Plans

The San Diego Interstate (I-15) Integrated Corridor Management system (ICMS) integrated existing systems with new or updated systems. The ICMS system design can be seen in Figure 4. The ICMS consists of the following key components: 1) Data Hub – collection of external systems operated by corridor stakeholder agencies providing data to the ICMS and/or receiving control requests from the system via a standardized regional communication network called the Intermodal Transportation Management System (IMTMS); 2) Decision Support System (DSS) – tool to help system operators identify incidents and implement response plans aimed at minimizing the impacts of identified incidents on corridor operations; and 3) System Services – services to assist with data management, system management, system maintenance, and training activities (e.g., ICMS data stores, corridor performance management).



Figure 4. Diagram. San Diego I-15 Integrated Corridor Management system design. (Source: ITS 3C Summit San Diego I-15 Integrated Corridor Management System presentation, 9/16/14, unpublished.)

As shown in Figure 5, the ICMS interfaces with a variety of systems (color coded based on facility type/functionality) that are managed by different agencies, including freeway systems in turquoise – Lane Closure System, Ramp Meter Information System, Advanced Traffic Management System, Congestion Pricing System, Express Lanes Control System; arterial systems in yellow – Regional Arterial Management System; transit systems in lavender – Regional Transit Management System,

Smart Parking System); public safety in purple – Regional Event Management System; and advanced traveler information systems (ATIS) in orange – Arterial Travel Time System, Traveler Information Systems, Weather Information System.



Figure 5. Diagram. San Diego I-15 Integrated Corridor Management system inputs and outputs. (Source: San Diego I-15 Demonstration Integrated Corridor Management System PATH Report on Stage 3: Site Demonstration and Evaluation, UCB-ITS-PRR-2015-03, p. 32.)

The innovative element of the DSS lies in its ability to forecast and simulate corridor performance issues using near real-time simulation and continuous predictive analysis, promoting proactive courses of action for recurrent and non-recurrent conditions (e.g., bottlenecks, incidents) which are coordinated among *all* corridor stakeholders. While the existing IMTMS network already facilitated decision-making by enabling interagency information sharing, it did not offer the functionality needed to integrate this information into actionable traffic control strategies. The DSS filled this gap by providing improved data fusion capabilities and a new decision-making process capable of generating (automatically or semi-automatically) multimodal response plans to events affecting corridor operations. The multimodal DSS, shown in the Decision Support System component in Figure 4, or the green Network Prediction System and Real-Time Simulation System in Figure 5, integrates two tools: 1) iNET – an automatic traffic management system for field device monitoring and control, center-to-center data fusion, event management and response plan generation; and 2) Aimsun Online – for real-time traffic prediction and simulation-based evaluation of incident response or congestion management strategies. When responding to an event, the DSS continues to monitor travel conditions within the corridor and issues updated recommendations when necessary, allowing

the DSS to account for unforeseen changes in travel patterns or other events affecting corridor operations *in addition* to the original event.

Roles and responsibilities have been diligently defined for all agencies/entities involved (e.g., San Diego Association of Governments (SANDAG); California Department of Transportation (Caltrans); City Traffic Divisions of San Diego, Poway, and Escondido; Metropolitan Transit System (MTS) and North County Transit District (NCTD) transit agencies; California Highway Patrol (CHP); local first responders and law enforcement, county emergency services) for the following scenarios:

- Daily operations;
- Freeway incidents;
- Arterial incidents;
- Transit incidents;
- Special event; and
- Disaster response scenarios.

As shown in Figure 6, San Diego stakeholders organized response "postures" around a combination of demand conditions on the network (light, moderate, or heavy) and predicted event impact (low, medium, or high). Within this framework, organized as a matrix, they then determined whether they would be likely to take "conservative," "moderate" or "aggressive" measures to manage the impacts of an event. They coded their joint response plans accordingly.





The San Diego DSS is dynamic, meaning there is no "set number" of defined response plans that could be recommended by the DSS. Figure 7 and Figure 8 show how a combination of subsystem action plans are used to define an individual response plan based on agreed upon response posture responsiveness. Between the 156 alternate routes, 260 local arterial intersections, 18 ramp metered interchanges, 20 changeable message signs (CMS), five bus rapid transit (BRT) stations (with six

extra buses in the metro area for adding transit capacity when needed), 20 miles of Express Lanes (16 miles which are reversible using a movable barrier) and 30 miles of traffic-responsive 511 within the study area, this provides enough assets that can be combined to generate *billions* of different response plans. However, the DSS is limited to recommending no more than 15 response plans at any time based on asset restrictions, availability conditions, and thresholds to select "next move" relationships. It should also be noted that response plans rely on how quickly field elements can be changed – e.g., the stakeholders are required to have time to actually implement recommended signal timing plans. The San Diego ICMS is capable of changing response plans every five minutes, but it is not practical to change these so frequently since it takes approximately 20 minutes to evaluate and implement (out in the field) a response plan. Although the San Diego DSS implements response plans without requiring human intervention, it does have the ability for a transportation operator to object to a recommended response plan and prevent it from being implemented.





Express	Transit	Traveler Information • Action Plan 1	Traffic Signal Timing	Ramp	
Not used	Action Plan 1  Action Plan 2	• Action Plan 3	Action Plan 2	Action Plan 1	
• Action Plan 2		<u>,</u>	Action Plan 4	• Action Plan 2 • Action Plan 3 • Action Plan 4	

#### Figure 8. Diagram. I-15 Decision Support System multi-modal response plans. (Source: San Diego Association of Governments, 3/6/14.)

Figure 9 shows an example of a response plan which was implemented at Rancho Bernardo Rd. during the northbound afternoon peak period for an event involving major congestion levels. This response plan triggered the following CMS message: "SLOWING AT // RANCHO BERNARDO // EXPECT DELAYS", ramp metering timing adjustments for two ramp meters, as well as traffic signal coordination timings for 15 signals in the cities of San Diego and Poway.



Figure 9. Illustration. Example response plan for northbound afternoon peak period congestion at Rancho Bernardo Rd.

(Source: I-15 ICM III – PDT Meeting #59, San Diego Association of Governments, 7/15/15, p. 28, unpublished.)

# Chapter 4. Integrated Corridor Management Strategies

The San Diego Integrated Corridor Management (ICM) focuses on five primary ICM goals:

- 1. The corridor's multimodal and smart-growth approach shall improve accessibility to travel options and attain an enhanced level of mobility for corridor travelers.
- 2. The corridor's safety record shall be enhanced through an integrated multimodal approach.
- 3. The corridor's travelers shall have the informational tools to make smart travel choices within the corridor.
- 4. The corridor's institutional partners shall employ an integrated approach through a corridor-wide perspective to resolve problems.
- 5. The corridor's networks shall be managed holistically under both normal operating and incident/event conditions in a collaborative and coordinated way.

To achieve these goals, San Diego Association of Governments (SANDAG) and its partnering agencies used investments in intelligent transportation systems (ITS) to implement a "smart" transportation management system that combines road sensors, transit management strategies, video, and traveler information to reduce congestion. The smart system is expected to deliver information to commuters via the Internet and message signs, and enable managers to adjust traffic signals and ramp meters to direct travelers to high-occupancy vehicle (HOV) and Express Lanes, bus rapid transit (BRT), and other options.

Specific examples of practices the San Diego site team employed include:

- Provided corridor users with the operational condition of all corridor networks and components, such as travel times, incident information, and expected delays using changeable message signs (CMS), a new 511 app, and other commercial travel time information sources.
- Used a decision support system (DSS) with real-time simulation, predictive algorithms, and analysis to evaluate potential congestion mitigation and select/implement the optimal combination of mitigation strategies for the corridor.
- Established, improved, and automated joint agency action plans for traveler information, traffic signal timing, ramp metering, transit and Express Lanes.
- Identified means of enhancing corridor management across all networks, including shared control multi-jurisdictional coordination of field devices such as lane controls, traveler information messages, traffic signal timing plans, and transit priority.

Main components of the ICM deployment in San Diego included:

- A DSS that utilizes incoming monitoring data to assess conditions, forecast conditions up to 60 minutes in the future, and then formulate and evaluate proactively recommended response plans (including selecting from pre-approved plans) for consideration by operations personnel.
- The Intermodal Transportation Management System (IMTMS), an existing data acquisition and dissemination network within the San Diego region, which was enhanced using ICM funding to connect certain regional systems with the ICM System (ICMS).
- En-route and pre-trip traveler information using CMS, a new 511 app, and other commercial travel time information sources.
- Adjustments to ramp meter timing to support diversion to or from the freeway.
- Upgrades to selected traffic signal systems, including new traffic signal coordination timings and responsive traffic signal control on two arterial streets paralleling Interstate 15 (I-15), as well as on arterials connecting the freeway to parallel arterials (concept shown in Figure 10).
- Alternate route wayfinding signs with the aid of highway CMS to help diverted drivers return to the freeway downstream of the incident.

These components increased the number of different ICM action plan categories (e.g., traveler information, traffic signal timing, ramp metering, transit, and Express Lanes) are available to the DSS to be included in a response plan. The thresholds and performance elements behind each category triggers different action plans. The combination of action plans triggered from each category formulates a recommended response plan.

Additional infrastructure investments made outside of the ICM Initiative not only improved mobility along the I-15 corridor as an isolated system, but also worked to supplement the corridor-wide ICM strategies implemented. For example, the tracking technology that were installed on the BRT buses (gathers data on the speed, location and passenger loads) and additional traffic detectors that were installed on arterials helped to enhance the transit and arterial network information that is used for real-time simulation of traffic conditions.



Figure 10. Map. I-15 responsive signal operations arterial groups. (Source: Integrated Corridor Management Stage 3A Analysis, Modeling, and Simulation for the I-15 Corridor in San Diego, California Pre-Deployment AMS Assessment Report, FHWA-JPO-13-007, p. 56.)

# Chapter 5. Analysis, Modeling and Simulation Methodology

The Analysis, Modeling and Simulation (AMS) methodology applied to the San Diego Demonstration Site was documented in the Federal Highway Administration's (FHWA) "Traffic Analysis Toolbox Volume XIII: Integrated Corridor Management, Analysis, Modeling, and Simulation Guide", a guide designed to help corridor stakeholders implement the Integrated Corridor Management (ICM) AMS methodology successfully and effectively. This guide provides a framework for developing an effective analysis plan to support selection and application of available tools and models specifically conducive to ICM.

Every tool type represents a tradeoff between geographic scope and level of resolution (scale versus complexity). Figure 11 shows the different types of analysis tools that can be incorporated into AMS. Less detailed tool types are tractable for large networks, while more detailed tool types are restricted to smaller networks. Depending on corridor size and the types of analyses required, all tool types are potentially valuable for ICM AMS. Microscopic simulation models, for example, are effective at analyzing system optimization strategies, such as freeway ramp metering and arterial traffic signal coordination, while mesoscopic simulation models are less effective, and travel demand models do not have this analysis capability. Travel demand models are better at estimating mode shift, but microscopic and mesoscopic simulation models are better at estimating mode shift. Mesoscopic tools can estimate regional dynamic diversion of traffic, while microscopic tools can estimate route shift at a smaller geographic scale. Finally, mesoscopic simulation tools are better at analyzing traveler responses to congestion pricing. The ICM AMS offers corridor managers greater capability than is available in any single existing tool.

## **Modeling Components**

The approach used to model the Interstate 15 (I-15) ICM corridor integrated a range of proven AMS tools and ICM analysis resources to form a single coherent system that can be used for corridor planning, design, and operations. This involves combining tools with inherently different analysis resolutions— macroscopic-level simulations for modeling travel demand and microscopic-level simulations for modeling detailed driver behavior—to evaluate a variety of ICM strategies and scenarios.

For AMS of the I-15 corridor, macroscopic models were used to produce Origin-Destination (OD) trip tables that were supplied as inputs to the microscopic simulation models. The microscopic models were then used to simulate the behavior of individual drivers in response to various control strategies on the corridor and at junctions (e.g., freeway interchanges or arterial intersections), including shifts within and between travel modes. The methodology also provided methods for connecting the different analysis tools, including post-processing modules that enable analysis of benefits and costs and the measurement of performance metrics.



#### Figure 11. Illustration. The Integrated Corridor Management Analysis, Modeling and Simulation Methodology blends up to three classes of modeling tools for comprehensive corridor-level modeling and analysis.

(Source: Cambridge Systematics, Inc., September 2009.)

An overview of the various components used in the AMS framework is provided in the following sections.

#### **Travel Demand Forecasting Model**

Predicting travel demand requires analysis tools that appropriately consider destination choice, mode choice, time-of-day travel choice, and route choice, given a traffic state for each link in the network. When combined into a coherent demand modeling framework, the result can be used to predict future travel patterns from current traffic levels, forecasted household characteristics, and predicted employment characteristics.

The San Diego Association of Governments (SANDAG) regional Coordinated Travel – Regional Activity Based Modeling Platform (CT-RAMP) travel demand model (TDM) was used to develop the broader OD matrices for the I-15 Corridor, which were then disaggregated by travel analysis zones to provide finer trip modeling resolution for simulation. Parameters from the TDM were used to model mode shifts in response to congestion and to ICM strategies.

### **Microscopic Simulation Model**

Microscopic simulation models simulate the movement, behavior, and decisions of individual drivers, based on models of car-following and lane-changing and a variety of population parameters. Typically, the analytical engine (which drives the simulation) begins by adding vehicles to the network at unconnected link entrances and at mid-block locations (representing new trips from origins along that link). These new trips are generated according to a specified distribution (e.g., Poisson, Uniform), with the shape parameters for these distributions based on user-defined values and the trip tables provided to the simulation. Similarly, driver characteristics (e.g., driver aggressiveness, following distance, acceleration or deceleration profile) are assigned for each vehicle at the time it first enters the simulation network according to statistical distributions. Each generated vehicle is also given a desired destination, and its progress toward that destination is simulated in small time increments

(e.g., half a second) or "simulation steps." Microscopic simulation models generally also consider roadway characteristics—including grade, lane width, and design speed—when evaluating the movement of individual vehicles on each link, with the effects of each roadway parameter being modeled according to relationships established by past research. Once a model is built, it must then be calibrated through the adjustment of driver and roadway parameters to achieve an optimal alignment between the route choices and link capacities observed in the model and measured in the field.

The microscopic simulation engine that was used for analysis of the I-15 ICM corridor is Aimsun, developed by Transport Simulation Systems (TSS). Figure 12 shows the model network that was used in post-deployment AMS. This model is currently being used as part of the decision support system (DSS) employed in the I-15 ICM system. This software suite is capable of simulating the details of ICM traffic control strategies, such as adaptive freeway ramp metering, arterial traffic signal coordination, and managed-use lane operations. At each step in the simulation, individual vehicles may be rerouted to different paths based on network conditions (e.g., congestion) and driver characteristics (e.g., driver willingness to divert, availability of traffic information to the driver). These routing decisions are based on the evaluated generalized costs (e.g., travel time) of each potential path to the traveler's destination. Some drivers, designated as "informed," were assumed to have perfect knowledge of real-time travel information (by means of a smartphone, global positioning system (GPS) device, etc.), and dynamically routed themselves through the network based on the currently evaluated shortest time paths to their destinations. Other drivers who are not considered to have access to real-time travel information in the simulation evaluated whether to divert to alternate routes in the face of heavy congestion based on historical travel time information, which these drivers would have learned through experience.

In addition to modeling traveler choices and network conditions, the simulation can also inform appropriate actions to take in response to congestion. Because Aimsun can realistically simulate the operation of various ICM components and the effects of changes to the network (e.g., lane blockages, real-time changes to speed limits), it can be used to evaluate the impacts of different operational decisions on congestion, bottleneck performance, or other metrics.

The traffic assignment method within Aimsun allows the use of static and dynamic assignment methods based on requirements of different study types. Traffic assignment models were used to estimate the flow of traffic on a network. These models take as input a matrix of flows that indicate the volume of traffic between OD pairs. The flows for each OD pair were loaded onto the network based on the travel time or impedance of the alternative paths that could carry this traffic. For traffic simulation models, the flow on a network was modeled by representing individual vehicle movements, and subsequently the link-based performance measures were evaluated based on movements of these individual vehicles as they rest in queues, travel in free flow, or maneuver through congestion. Whether all vehicles traveling a given path reach all links on the path within a given analysis period was dependent on time-variant travel conditions in the network (Source: AIMSUN Microsimulator and Mesosimulator User's Manual.).



Figure 12. Map. Model network I-15 for post-deployment Analysis, Modeling, and Simulation. (Source: San Diego: Integrated Corridor Management System, Transport Simulation Systems.)

The key behavioral assumptions underlying the User Equilibrium (UE) assignment model are that every traveler has perfect information concerning the attributes of network alternatives, all travelers choose a route that minimizes their travel time or travel costs, and all travelers have the same valuations of network attributes. At UE, no individual travelers can unilaterally reduce their travel time by changing paths. A consequence of the UE principle is that all used paths for an OD pair have the same minimum cost. An alternative and more realistic equilibrium model is known as Stochastic User Equilibrium or SUE. This model is premised on the assumption that travelers have imperfect information about network paths and/or vary in their perceptions of network attributes. At SUE, no travelers believe that they can increase their expected utility by choosing a different path. Because of variations in traveler perceptions and also in the level of service experienced, utilized paths do not necessarily have identical generalized costs. The SUE model is consistent with the concept of applying discrete choice models for the choice of route, but with the necessary aggregation and equilibrium solution.

#### **Temporal Analytical Resolution**

Microscopic simulation models require travel demand data in the form of O-D tables to properly generate and distribute (e.g., by mode, by route) trips on the network. These tables are generated on the macroscopic level by regional travel demand models and supplied as inputs to the microsimulation engine in Aimsun. Generally, these regional models lack the temporal resolution to be suitable for use with microsimulation, with the macroscopic trip tables being aggregated into intervals of several hours each and the microscopic simulations requiring inputs on the order of 15-minute increments to achieve realistic and reasonable results. However, SANDAG has developed a travel demand model that produces trip tables in 15-minute intervals, making it suitable for use with Aimsun's microsimulation engine.

# **Modeling Integrated Corridor Management Strategies**

Modeling ICM strategies is discussed in further detail in the ensuing sections.

## **Pre-Trip Traveler Information**

Pre-trip traveler information includes any travel information accessible to the public that can be used in planning trip routes, estimating departure times, and/or choosing travel mode. Such information can be available through the 511 system, via the phone, the Internet, or public access television. The analysis captured the impacts of such information on traveler's route choice and departure times. The fraction of I-15 users (pre-ICM and post-ICM), who access such information prior to making their trip, was estimated based on findings from the Volpe Center's "Integrated Corridor Management Initiative: Traveler Response Panel Survey – San Diego" and was buttressed using data sources available in the region, such as available information on utilization of features like 511 and traffic web sites in San Diego. Subsequently, this portion of the traveling population (the "informed travelers") was identified as a particular traveler class within the model.

As shown in Table 1, 93 percent of travelers were considered to have access to pre-trip traveler information before the ICMS was deployed (pre-ICM). Despite the fact that more valuable traveler information was made available after deployment (post-ICM), a*wareness* levels remained stagnant. *Compliance* to pre-trip information increased 0.4 percent post-ICM to 8 percent.

#### **En-Route Traveler Information**

As part of the I-15 ICM system an enhanced 511 system was deployed in the region with predictive traveler information. AMS analyzed the impact of en-route information available to travelers on changes in route choice. Changes in route choice relate to real-time change in route choice of travelers based on travel time or congestion updates they receive via radio, 511, smart phones, or wireless-equipped GPS devices. This feature was incorporated into the analysis as a fixed percentage of drivers who would be likely to have this information, along with a corresponding "compliance ratio" representing travelers who would consider changing route if faced with congestion.

To facilitate AMS of traveler responses to CMS, modeling sensors were coded in the model along the route upstream of the message sign. As drivers approach the message sign, they pass through these sensors, which in turn calls up a macro that updates these drivers' route choice decisions. When the macro is activated, new routes are assigned to the percentage of drivers that divert their routes based on the posted information. Depending on the scenario or type of incident that may have occurred, compliance rates associated with each message sign varies, and hence the amount of route diversion also differs throughout the simulation runtime.

Table 1 provides a summary of modeling assumptions used in pre- and post-deployment AMS regarding awareness, use, and compliance to pre-trip and en-route traveler information for both preand post-ICM implementation. The post-deployment contents (both pre-ICM and post-ICM) in this table were refined based on the findings resulting from the traveler survey conducted by the Volpe Center in the I-15 corridor, detailed in the report titled "Integrated Corridor Management Initiative: Traveler Response Panel Survey – San Diego".

As shown in Table 1, *awareness* of en-route traveler information increased from 83 percent pre-ICM to 84 percent post-ICM. *Use* levels increased by less than one percent post-ICM, from 12.3 percent to 12.6 percent, indicating a slight increase in the relevance of en-route information, while *compliance* levels also increased by less than one percent post-ICM, from 10.2 percent to 10.6 percent.

Note: "Use" does not necessarily result in an action, unless the proposed mode-route option is more attractive than the "historical route," based on the diversion rules. Therefore, "use" reflects an upper bound on the percent of travelers who might divert as a response to the information, with the actual percentage dependent on the attractiveness of the new route and referred to as "compliance."

Table 1. Modeling assumptions regarding awareness, use, and compliance to traveler
information.

	<b>Pre-Trip</b> (Agency websites, 511, public access TV, local radio, etc.)			En-Route (CMS, radio, 511, GPS devices, etc.)			
	Awareness	Use	Compliance	Awareness	Use	Compliance	
Pre-Deployment: "Awareness" percentages were based on expected adoption rates of GPS devices and Personal Digital Assistants. The pre-deployment AMS tool (TransModeler) did not require "use" and "compliance" rates.							
Pre-ICM	5%	N/A	N/A	5%	N/A	N/A	
Post-ICM	30%	N/A	N/A	30%	N/A	N/A	
Post-Deployment: Based on overall findings from panel surveys of I-15 corridor users, conducted by the Volpe Center (not pulse surveys).							
Pre-ICM	93%	8.2%	7.6%	83%	12.3%	10.2%	
Post-ICM	93%	8.6%	8%	84%	12.6%	10.6%	

(Source: Integrated Corridor Management Analysis, Modeling, and Simulation for the I-15 Corridor in San Diego, California Post-Deployment Analysis Plan, FHWA-JPO-16-393, p. 29 and Integrated Corridor Management Initiative: Traveler Response Panel Survey San Diego - Draft, Volpe National Transportation Systems Center, July 2016.)

#### **Ramp Metering**

The I-15 freeway currently has a number of ramps that are metered in both the northbound and southbound directions. The meters operate on a local occupancy-based algorithm working off the San Diego Ramp Metering Software. The analysis modeled a corridor-coordinated ramp metering algorithm implemented under the Intermodal Transportation Management System (IMTMS) framework for both pre- and post-deployment ICM.

## **Traffic Signal Timing**

In the San Diego ICM network, most of the signalized intersections are operated with actuated signal control whose parameters are defined by different cities and authorities (i.e., Caltrans, Cities of San Diego, Poway, San Marcos, and Escondido). In the model, signal timing plans are operating according to a fixed time-of-day schedule for the base scenario. For each intersection, there are approximately eight different coordinated control plans and one available non-coordinated one.

When a response plan is triggered due to a congestion event, some intersections along the alternative routes switch control plans to provide additional green time to support the increase of traffic. This causes the iNET system to send Aimsun the details for each response plan (including ramp metering, signal timing changes and diverging route). Each diversion route has a predefined set of signal changes that are activated in the model via a traffic management action that starts when the response plan is activated and lasts for the duration of the congestion event.

# **Chapter 6. Performance Measures**

This chapter provides an overview of the performance measures used in the Analysis, Modeling and Simulation (AMS) of Integrated Corridor Management (ICM) strategies for the I-15 Corridor.

The performance measures analyzed by the AMS team focused on the following key areas.

## **Mobility**

Mobility describes how well the corridor moves people and freight. The mobility performance measures are readily forecast by the AMS tools used. Three primary types of measures were used to quantify mobility in the Interstate 15 (I-15) Corridor, including:

- **Travel time**—This is defined as the average travel time for the entire length of the corridor or segment within the corridor by facility type (e.g., mainline, high-occupancy vehicle (HOV) lanes, and surface streets), mode, link, individual traveler, and by direction of travel. Travel times were computed for each peak period analyzed.
- Delay—This is defined as the total observed travel time less the travel time under uncongested conditions, and is reported both in terms of vehicle-hours and person-hours of delay. Delays were calculated for freeway mainline and HOV facilities, transit, and surface streets, for all travelers individually and cumulatively, in all analysis scenarios.
- **Throughput**—Throughput is measured by comparing the total number of vehicles entering the network and reaching their destination within the simulation time period. The measure ensures that the throughput of the entire system can be utilized as a performance measure for all the scenarios. The corresponding Vehicle Miles Traveled (VMT), Person Miles Traveled (PMT), Vehicle Hours Traveled (VHT), and Person Hours Traveled (PHT) were reported as a macroscopic measure of the general mobility of the corridor.

## **Reliability and Variability of Travel Time**

Reliability and variability capture the relative predictability of the public's travel time. Unlike mobility, which measures how many people are moving at what rate, the reliability and variability measures focus on how much mobility varies from day to day. Travel time reliability was reported in terms of changes in the Buffer Time and Planning Time Index, while travel time variability was reported in terms of changes in the standard deviation of average travel time. The Planning Time Index is a ratio of the 95<sup>th</sup> percent peak period travel time to the free flow travel time. A value of 2.50 indicates that for a trip that takes 30 minutes in light traffic, a person should budget 75 minutes (30 minutes x 2.50) to ensure on-time arrival 95 percent of the time. The Buffer Time represents the additional time (or time cushion) that travelers must add to their average travel time when planning trips to ensure on-time arrival. Since a microscopic model was used (Aimsun), the AMS Team used post-processors to calculate the impacts on the reliability and variability of travel time. Appendix B describes the methodology that was used in calculating reliability and variability impacts.

# **Other Measures**

#### **Emissions and Fuel Consumption**

The I-15 Corridor AMS also produced model outputs for use by the Evaluation Contractor to estimate emissions and fuel consumption, associated with the deployment of ICM strategies. The emissions analysis methodology incorporated reference values to identify the emissions and fuel consumption rates based on variables, such as facility type, vehicle mix, speed ranges, and acceleration ranges. The emissions and fuel consumption rates were based on available sources. Emissions that are principal pollutants of concern include nitrogen oxides (NO<sub>x</sub>), particulate matter (PM), hydrocarbons (HC), volatile organic compounds (VOCs), carbon monoxide (CO), sulfur dioxide (SO<sub>2</sub>), hazardous air pollutants (toxics), and greenhouse gases (CO<sub>2</sub>). Emissions are generally measured in terms of kilograms of output and computed by pollutant, mode, and facility type. Fuel consumption is typically computed by fuel type, mode, and facility type. Fuel consumption is generally measured in terms of gallons of fuel consumed. The broader Evaluation Report will contain the results on the specific measures used to evaluate the impact of ICM strategies on emissions and fuel consumption.

#### **Cost Estimation**

For the identified ICM strategies and based on input by the Evaluation Contractor, planning-level cost estimates will be prepared by the Evaluation team for life-cycle costs (capital, operating, and maintenance costs) and therefore, were not a part of this analysis. Typically, analyzed scenarios representing different operating conditions will be combined together, weighted by the probability of occurrence to arrive at a total annual benefit, net annual benefit, and benefit-cost. Please refer to the full Evaluation Report (scheduled to be published in 2017) for the final benefit-cost assessment.

## Safety

Although safety is an important performance measure to consider, currently, available safety analysis methodologies are not sensitive to ICM strategies. At best, available safety analysis methods rely on crude measures, such as a volume-to-capacity ratio (V/C), and cannot take into account ICM effects on smoothing traffic flow. Clearly, this is an area deserving of new research and as such, no explicit safety analysis was conducted as part of this effort.

## **Summary of Performance Measures**

Table 2 provides a summary of the mobility, reliability, and variability performance measures used to analyze the impacts of ICM. Performance measures which are typically used in evaluating emissions, fuel consumption, and cost estimation are listed.
Category	Performance Measure
	Travel time: average travel time
Mobility	Delay: vehicle-hours of delay, person-hours of delay
	Throughput: vehicle miles traveled, person miles traveled, vehicle hours traveled, person hours traveled
Travel Time Reliability	Buffer Time, Planning Time Index
Travel Time Variability	Changes in the standard deviation of average travel time
Emissions	Kilograms of Nitrogen oxides (NO <sub>x</sub> ), particulate matter (PM), hydrocarbons (HC), volatile organic compounds (VOCs), carbon monoxide (CO), sulfur dioxide (SO <sub>2</sub> ), hazardous air pollutants (toxics), and greenhouse gases (CO <sub>2</sub> )
Fuel Consumption	Gallons consumed for each fuel type
Cost Estimation	Infrastructure costs and incremental costs for capital costs, operating costs, and maintenance costs
<b>Operational Characteris</b>	stics
Facility Type	Mainline, High-Occupancy Vehicle (HOV) lanes, Surface Streets
Mode Type	Drive, Transit
Direction of Travel	Northbound, Southbound
Time of Day	AM peak period, PM peak period
Scenarios	With-ICM, without ICM

Table 2. Summary of performance measure categories and operational characteristics for
analysis.

(Source: Cambridge Systematics, Inc., 2016)

# Chapter 7. Post-Deployment Analysis, Modeling and Simulation Approach

Pre-deployment Analysis, Modeling and Simulation (AMS) activities were associated with AMS support prior to the deployment and activation of Integrated Corridor Management (ICM) systems. Pre-Deployment AMS activities focused on the expected impacts and benefits of ICM associated with "as planned" ICM strategies prior to deployment. Pre-Deployment AMS activities were intended to both refine and prepare AMS capabilities to represent the "as planned" ICM strategies and to inform an ICM evaluation regarding the type, location, and intensity of potential benefits.

Post-Deployment AMS activities focus on identifying impacts and benefits of the "as-deployed" ICM system. The "as-deployed" ICM strategies may differ from "as-planned" ICM strategies. The differences could include ICM strategies that were not successfully deployed, ICM strategies that were deployed differently from planned because of technical issues, and ICM strategies that were deployed differently to take advantage of enhancements or impacts not anticipated pre-deployment. Further, Post-Deployment AMS activities should take full advantage of site-specific traveler behavior and response characterization efforts conducted by the ICM Evaluation team. This includes the refinement of parameters and methods in tools to most accurately reflect traveler behavior in response to ICM strategies.

This chapter describes the post-deployment AMS activities that supported the ICM system for the Interstate 15 (I-15) corridor. During post-deployment AMS, the tools and methodologies developed in previous AMS efforts were revisited and further evaluated in order to improve the capability of the site-specific tools to represent and evaluate the ICM system. The key objectives of post-deployment AMS include the following:

- Identify and facilitate further enhancements to tools, data, and methods developed from previous AMS activities.
- Conduct modeling analysis using enhanced tools in order to assess the impacts of the ICM strategies deployed in the corridor.
- Provide guidance for the site's ICM deployment and support for the integration of the AMS tools and methods developed with their ongoing corridor management practices.
- Support Demonstration Site-Specific ICM Demonstration Evaluation efforts.
- Manage the successful transition of modeling leadership responsibilities from the AMS contractor to the ICM Demonstration site staff and organizations.
- Provide technical documentation of ICM AMS tool development, data collection and analysis, model calibration and validation methods, and analytical methods deployed to both represent and evaluate ICM impacts.

To achieve these objectives, post-deployment AMS included the following tasks in order to evaluate the impacts and readiness of the deployed ICM system. Subsequent sections provide further detail on each of the following tasks:

- Enhance tools to reflect as-deployed corridor management. Adjust tools and methods to differentiate the "as-deployed with-ICM" and "without-ICM" alternatives in analytical tools— this was accomplished by modifying model inputs, assumptions, and analytical approaches to reflect as deployed ICM strategies and observed traffic conditions.
- Conduct post-deployment alternatives analysis using most impactful scenarios from cluster analysis and incident matching.

# **Model Enhancements**

This section describes the task items related to coordination and support of the alteration of tool inputs, analytical methodology, and enhancements to analytical software to reflect post-deployment corridor management technologies and strategies. The AMS team coordinated with the I-15 ICM team and the Evaluation team to confirm, refine, and validate the parameters and assumptions that serve as the basis for modeling traveler responses and impacts related to ICM strategies currently present in the models used in the real-time decision support efforts. San Diego Association of Governments (SANDAG) and local stakeholders reviewed the model parameter assumptions to ensure that they sufficiently capture travel characteristics for the corridor and system response times according to the capabilities of their transportation management systems.

Post-deployment AMS work captured the nature of the as-deployed system, including a good representation of traveler responses to ICM strategies, based on site-specific measurements of traveler responses and reactions, conducted in other parts of the ICM program. The AMS team coordinated with both the ICM Demonstration Site and the Evaluation team to clearly identify whether the deployed capability matches the assumptions made for modeling and simulation.

# **Analysis Tool for Post-Deployment AMS**

Early in the AMS process, a decision was needed regarding whether to conduct AMS in: 1) the simulation platform (Aimsun) used in real-time decision support in the I-15 ICM and currently incorporated into the ICM management software, or 2) in the simulation platform used in predeployment AMS (TransModeler). Factors considered included:

- The Aimsun model needed modifications to allow it to: 1) meet certain model validation benchmarks, 2) represent the full peak periods, instead of hourly traffic conditions, 3) conduct real-time mode shift analysis, and 4) conduct real-time analysis of parking demand and capacity. On the positive side the Aimsun model: 1) included all current ICM strategies already coded in the model; 2) had archived data associated with different operational conditions (incidents, high demand, etc.); 3) was available for both the AM and PM peak periods; and 4) its use would ensure better consistency with the ICM evaluation effort as both efforts would rely on the same datasets.
- The TransModeler model was already calibrated but it used 2003 data for its baseline (effects of the recent economic recession may not have been properly accounted for), and still needed to be made consistent with current travel demand and ICM deployment data, and focused on the AM peak only.

Upon discussion with the I-15 project team (including SANDAG and their partners and contractors) the Aimsun platform was selected as the modeling tool to be used in post-deployment AMS. Choosing to switch to the Aimsun Online platform was a catalyst that transformed agency practice in the San Diego region, becoming the standard tool used by transportation agencies in the area.

### Ensure I-15 ICM System is Accurately Represented in the Model

To more precisely model the operation of the ICM system and evaluate its benefits, the AMS team used the ICM test system and operational model to check and test the operation of the response plans within the system. This allows the Aimsun Online model to estimate impacts of the response plans and approximate the decision support system (DSS) process. It was important to use the test system as the implementation of response plans has to date been limited. In addition to calibrating the model for a typical day the San Diego I-15 team also conducted a reasonableness assessment for an incident day. The selected real incident occurred on October 16th, 2014 and blocked two to three lanes on northbound I-15 south of Pomerado Boulevard interchange near Lake Hodges between 5:30 PM and 7:00 PM. This day was chosen because it had a major I-15 incident that met the criteria that the system is expected to respond to and show a significant benefit through the rerouting of traffic combined with the managing of the arterials signals, ramps meters, and variable message signs.

### The I-15 Model Better Represents the Congestion Pricing System

Along the I-15 corridor the Express Lanes can be used for free by high-occupancy vehicles (HOV), and single-occupancy vehicles (SOV) vehicles with transponders pay a variable toll that is generated by the Congestion Pricing System (CPS). In assigning vehicle types the I-15 model relies on the vehicle mix defined by the SANDAG regional model where vehicles are categorized as SOV Toll and SOV Nontoll, as well as HOV Toll and HOV nontoll. The CPS generates the toll rate per mile that applies to SOV vehicles and is used in the drivers' deciding whether the cost of the toll is greater than the cost of lost time when using the general-purpose lanes.

Improvements and changes were needed as it was observed that the regional model underestimated the number of CPS users and so the AMS team conducted sensitivity tests on the percentage of SOV Toll vehicles by adjusting the travel demands of the offline AM and PM peak period models. The sensitivity tests compared the mainline and CPS volumes greater than 2,000 vehicles per hour along the corridor under congested conditions where the variable toll was greater than the minimum value. The tests and modifications resulted in an increase in the number of SOV Toll vehicles and a decrease in the SOV Nontoll vehicles. With these changes, the peak directions of the CPS system were matched with a higher level of accuracy.

# The I-15 AMS Tool Enhancement to Allow for the Creation of Multi-Hour Models

The I-15 ICM Online system was running simulations for a 60-minute duration. For the AMS analysis the simulations need to run for a longer duration to be able to analyze the impacts over the complete peak. As part of the ICM system, a tool was used to extract simulations run as part of the online system, either for the predictions or for the evaluations of response plans to be used for offline review and analysis. This tool was modified to allow the modeler to enter the time range and date of the offline simulation model to be generated. Using the stored input data files from the ICM system, the tool collects the adjusted 15-minute demand files, network device status files (signals, ramps, signs...)

and all event data over the duration of the model. This tool is an integral part of being able to correctly analyze the impacts and benefits of the suggested or implemented response plans over the longer durations. This tool was used to generate the initial models used for the Post-Deployment AMS Reasonableness Assessment described in the Model Calibration & Reasonableness Assessment section of Chapter 7.

### **Traveler Information Sensitivity Analysis**

One of the major efforts to enhance the post-deployment model was refining model parameters and structure by modifying traveler information availability to better represent route diversion and mode shift based on Volpe Center surveys. The Volpe Center gathered behavioral data for travelers in the area of the I-15 ICM project through panel surveys. The ICM Evaluation team collected and analyzed field data for the post-deployment period, and the AMS team modeled different operating scenarios (with and without ICM) using post-deployment data as well. Collaboration between efforts was needed to ensure that any major events (i.e., incidents) that occur on the corridor were properly captured/analyzed by all three teams. Furthermore, traveler information parameters and assumptions were collected by both the Volpe Center travel surveys and by the ICM evaluation effort.

The following Volpe survey measures were used in the AMS analysis:

- Percent travelers who made a travel change based on pre-trip information (percent of travelers who changed time of departure, route, mode, destination, or decided not to make trip); and
- Percent travelers who made a change to their trip (en-route) based on information (percent of travelers who changed route, mode, and destination).

A panel survey approach was selected whereby the same individuals are surveyed both before and after the deployment of ICM. Since the impacts of ICM are expected to be greatest during incident conditions, the methodology included a series of "pulse" surveys that were administered immediately following incidents in each of the corridors. This enabled the measurement of trip-specific behavior during incident conditions.

The study population included "regular" users of the main facility in the I-15 corridor. In order to qualify for the survey, individuals had to travel on the facility three or more weekdays per week in either the AM peak period (6-10 AM) and/or the PM peak period (3-7 PM). The study population was constrained in this way for two reasons:

- 1. Regular users are familiar with the performance of the facility and are likely to be more sensitive to any changes in corridor performance.
- 2. In order to successfully conduct the pulse surveys, the Volpe Center required a panel of travelers who are regularly on the facility (particularly at congested times of the day, such as the AM and PM peak), so that they maximize the pool of respondents who are eligible to be pulsed for any given incident (and thus increase the likelihood of obtaining responses to the pulse surveys).

The purpose of the survey was to measure the impacts of ICM on travelers in each of the corridors. More specifically, the survey addresses:

- Changes in peak period travel behavior (mode, route, timing, frequency, etc.) due to conditions in the corridor and due to improved traveler information;
- Changes in satisfaction regarding travel/trip experiences in the corridor;

- Ability of travelers to detect improvement in the quality of service in the corridor;
- Changes in awareness of traveler information sources;
- Changes in reported utilization of (frequency, method, timing, etc.) traveler information sources; and
- Changes in satisfaction regarding traveler information/sources.

These measures were identified for comparable incidents in the pre- and post-ICM periods (and when a response plan was implemented in the post-ICM period). The "Post-Deployment" section of Table 1 outlines the model parameter values derived for awareness and use, based on real data collected from traveler surveys.

### **Other Model Enhancements**

Additional model enhancements were conducted as follows:

- Network Changes—Three main types of changes were made to the network for both offline and online use, including: a) Geometric changes including edits to intersections and new Direct Access Ramps (DAR); b) Transit updates including new bus routes or new bus services; and c) Signal updates to make sure that any new signals or signals added to the Regional Arterial Management System were included.
- Demand Changes—The following steps were involved with the update to the travel demands using more recent travel demand data. Step 1) Review and update of detection to account for the updates of the external systems. Step 2) New data collection to update the historical patterns for the Monday, Tuesday/Thursday, Wednesday and Friday day types. Step 3) Creation of patterns for the different day types using various types of filtering mechanisms. Step 4) Creating and training the models. Step 5) Generation of demand matrices for each typical day.
- Online System Files—For microscopic simulations, it was necessary to upload a path assignment file and an initial state file to start each simulation. The initial state files were created at the same time that the dynamic adjustments were run. Every 15 minutes a dynamic adjustment was run and a new initial state was saved after 15 minutes of simulation, so that the following simulation for the dynamic adjustment could start with this new initial state created with the adjusted matrices.
- Post-Processors—Currently the Aimsun model does not have the ability to calculate impacts on the reliability of travel time, and includes internal processors to calculate impacts on vehicular emissions and fuel consumption based on European standards. The model was enhanced so that it could calculate travel time reliability impacts as well as produce estimates of emissions and fuel consumption impacts based on California standards for San Diego. The AMS team provided post-processors which produced inputs to the Evaluation Contractor's travel time reliability impacts, as well as estimates of emissions and fuel consumption impacts.

# **Summary of Analysis Settings**

A summary of the ICM strategies implemented in the "as-deployed" ICM system, which were replicated in the models used for post-deployment AMS include:

- Active Decision Support System;
- Coordinated incident management;
- Freeway coordinated ramp metering;
- Actionable traveler information (en-route and pre-trip via changeable message signs (CMS), a new 511 app, and other commercial sources);
- Upgrades to selected traffic signal systems (new traffic signal coordination timings, responsive traffic signal control); and
- Alternate route wayfinding signs.

Based on the Volpe Center traveler surveys, the "Post-Deployment" section of Table 1 presents the parameters that were used in the AMS related to the travelers' awareness, use and compliance to traveler information:

- "Awareness" represents the portion of travelers who have access to information. For awareness the AMS used the percentages from the Volpe Center's baseline/endline surveys, and they were both in the mid 80 to 90-percent range.
- "Use" represents a traveler's intent to take action, but does not necessarily result in an action, unless the proposed mode-route option is more attractive than the "historical route," based on the model's diversion rules. Therefore, "use" reflects an upper bound on the percent of travelers who might divert as a response to the information, with the actual percentage dependent on the attractiveness of the new route and referred to as "Compliance." For better linearity of model functions (non-jumpiness across steps) the model uses this convention, where "compliance"="awareness" \* "use".
- This AMS effort (as reported in the "Post-Deployment" section of Table 1) used the compliance numbers reported in the pulse summary surveys provided by the Volpe Center.

Table 3 shows the analysis settings used for conducting Post-Deployment AMS for the San Diego I-15 corridor.

Parameter	Value	Comment
Analysis year	2015	The analysis year was derived from the anticipated completion of design, testing, and deployment of ICM.
Time period of analysis	AM peak period (6 AM to 10 AM) PM peak period (3 PM to 7 PM)	Several incidents and bottleneck events that occurred in days representative of different clusters, and for which response plans were activated were selected to represent AM and PM peak periods. Also, one hypothetical scenario was selected for analysis, including opening all Express Lanes to all traffic during a major incident.
Simulation period	4 hours in each peak period	6-10 AM and 3-7 PM were selected to represent the AM and PM analysis periods.
Freeway incident locations and durations	Based on cluster analysis and presented in Table 8	These locations experienced incidents, offered the potential for route diversion, had a response plan activated, and had a high impact on corridor travel.

### Table 3. San Diego I-15 corridor—summary of post-deployment analysis settings.

(Source: Integrated Corridor Management Analysis, Modeling, and Simulation for the I-15 Corridor in San Diego, California Post-Deployment Analysis Plan, FHWA-JPO-16-393, p. 29.)

# Model Calibration & Reasonableness Assessment

Accurate calibration is a necessary step for proper simulation modeling. This is especially critical for the real-time modeling done in I-15's ICM system, which requires models to be accurate on a daily basis in order for the ICMS to function properly. Before modeling ICM strategies, model calibration ensures that base scenarios represent reality, creating confidence in the scenario comparison. Each simulation software program has a set of user-adjustable parameters that enable the practitioner to calibrate the software to better match specific local conditions. Calibration improves the ability of the model to accurately reproduce local traffic conditions. The calibration efforts conducted as part of the ICM initiative exceeded standard calibration efforts by introducing innovative methods such as having specific calibration criteria for incident days and transit. The key steps in model calibration include:

- Identification of necessary model calibration targets;
- Selection of the appropriate calibration parameter values to best match locally measured street, highway, freeway, and intersection capacities;
- Selection of the calibration parameter values that best reproduce current route choice patterns; and
- Calibration of the overall model against overall system performance measures, such as travel time, delay, and queues.

Available data on bottleneck locations, traffic flows, and travel times were used for calibrating the simulation model for the analysis of the I-15 corridor. The I-15 Corridor calibration strategy was based on the three-step strategy recommended in the Federal Highway Administration (FHWA) Guidelines for Applying Traffic Microsimulation Modeling Software (Source: Dowling, R., A. Skabardonis, and V. Alexiadis, *Traffic Analysis Toolbox Volume III: Guidelines for Applying Traffic Microsimulation Modeling Software*, U.S. DOT-HRT-04-040, Federal Highway Administration, July 2004.):

- **Capacity calibration.** An initial calibration was performed to identify the values for the capacity adjustment parameters that cause the model to best reproduce observed traffic capacities in the field. A global calibration was first performed, followed by link-specific fine-tuning. The capacity calibration for the I-15 Corridor was performed utilizing volume data collected from the Caltrans Performance Measurement System (PeMS) database for the year 2003 between the periods of September to November.
- **Route choice calibration.** Because the I-15 Corridor includes parallel arterial streets, route choice calibration plays a significant role in the overall calibration effort. After capacity calibration, this second calibration process was performed with the route choice parameters. A global calibration was first performed, followed by link-specific fine-tuning.
- System performance calibration. Finally, the overall model estimates of system performance (travel times and queues) was compared to the field measurements for travel times and queues. Fine-tuning adjustments are made to enable the model to better match the field measurements.

### Post-Deployment AMS Tool Reasonableness Assessment

Full recalibration of the model system was not expected to be required in the Post-Deployment AMS Phase. However, a Reasonableness Assessment was conducted, where the model inputs and parameters were modified as necessary so that the model can reasonably match Post-Deployment field conditions, including location, extent, and severity of bottlenecks. The objective of the Reasonableness Assessment was to review the post-deployment simulation model, and modify the model inputs accordingly in order to ensure that the model sufficiently replicates and simulates observed travel conditions and congestion patterns on the field during the post-deployment stage of ICM.

### Methodology

The Reasonableness Assessment Methodology involved the comparison of the I-15 year 2014 model volumes, travel times, and speeds (including bottleneck locations) with field observed data in 2014. In order to perform this assessment, the methodology included four steps, as detailed in the following sections.

#### Step 1. Data Collection

The first step in the Reasonableness Assessment was to obtain the necessary data inputs. The data inputs for this assessment of the I-15 corridor included volumes observed in the field and speeds from the following sources: Freeway General Purpose, Mainline Lanes; Freeway Express Lanes – HOV and Single-Occupancy Vehicles (tolled); On and Off Freeway Ramps; and, Arterials.

The data used corresponded to a typical day and to a day with an incident. In calculating the typical day, data were used from a day that was clear of incidents during the peak periods. In the initial calibration of the I-15 model, as part of the implementation process, both static counts and live data from the external systems were used as limited arterial data were available. In the recalibration effort, further arterial data was added to the external data and was included as part of this Reasonableness Assessment. The data sources included Caltrans Advanced Traffic Management System (ATMS), the Ramp Metering Information System (RMIS), and the Regional Arterial Management System (RAMS). Since the last update to the on-line model, a number of

improvements had been made to the ATMS stations and hence this effort included data from the stations currently in use and the stations that have been updated since the last model update using PeMS data. Where available, PeMS data were used and whenever an issue occurred, external system data were used, filling the gaps with pattern data; this gave a higher level of fidelity to the volume and bottleneck checks compared to using only the current online data set.

#### Step 2. Reasonableness Assessment Criteria

The Reasonableness Assessment methodology employed similar elements of the model calibration criteria detailed in the United States Department of Transportation (US DOT) "Guidelines for Applying Traffic Microsimulation Modeling Software", including two types of data comparisons:

- 1. Volume Comparison. The first part determined whether the 2014 I-15 post deployment model reasonably replicates observed volume data for 2014. The criteria for comparing hourly flows between model and observed values are summarized in Table 4. Note that peak periods are defined as 6:00 AM to 9:59 AM for the AM peak period and 3:00 PM to 7:00 PM for the PM peak period.
- 2. Travel Speeds and Bottlenecks. The reasonableness assessment of the model's speeds was based on a visual audit that compared speed contour diagrams from detector data from a typical weekday with modeled speed data, as summarized in Table 5. The speed contour diagrams depict typical weekday speeds along the I-15 corridor during the AM, Mid-Day and PM peak periods as defined above.

#### Table 4. Reasonableness Assessment criteria and acceptance targets.

Calibration Criteria and Measures	Calibration Acceptance Targets
Traffic flows within 15% of observed volumes for links with peak-period volumes greater than 2,000 vph	For 85% of cases for links with peak-period volumes greater than 2,000 vph
Sum of all link flows	Within 5% of sum of all link counts
Visual Audits Individual Link Speeds: Visually Acceptable Speed-Flow Relationship	To analyst's satisfaction
Visual Audits Bottlenecks: Visually Acceptable Queuing	To analyst's satisfaction

(Source: Integrated Corridor Management I-15 San Diego, California Analysis Plan, FHWA-JPO-10-039, p. 38.)

#### Step 3. Model vs. Observed Data Comparison

The third step of the Reasonableness Assessment involved comparing the 2014 model outputs/ performance measures against field data along the I-15 Corridor. The criteria established in Step 2 were then utilized to determine whether the model results adequately replicated the field data.

### Step 4. Travel Demand and Network Adjustments

Based on the results of the initial comparison conducted in Step 3, additional work was needed in order to adjust the overall utilization of the Express Lanes in the northbound direction during the peak hour. The main step in this was to adjust the travel demand distribution of vehicles that can access the Express Lanes as HOVs or SOVs with a toll responder. After these changes were made a better fit to the real values was achieved.

### Step 5. Incident Day Model Assessment

For an incident day, the following criteria were used within the context of the model calibration reasonableness assessment:

- **Freeway bottleneck locations.** Should be on a modeled segment that is consistent with the location, design, and attributes of the representative roadway section.
- Duration of incident-related congestion. Duration where observable within 25 percent.
- Extent of queue propagation. Should be within 20 percent.

### Results – Typical Day

### Link Count Comparisons

A total of 86 freeway mainline stations and seven Express Lanes stations in the AM peak period, 89 mainline stations and seven Express Lanes stations during the PM peak period and 70 mainline stations in the MD peak period had over 8,000 vehicles (equivalent of 2,000 vph). None of the available arterial stations meet the 8,000-vehicle threshold.

Table 5 shows the link count comparison for all I-15 locations that met the thresholds for the typical day. For a typical day with no incident, the sum of all link counts fall within the five percent acceptance target.

### Table 5. Count comparison for all I-15 locations above VPH threshold – typical day.

Мо	del	Obse	erved	Differe	ences	Percen	t Error
6-10 AM	3-7 PM	6-10 AM	3-7 PM	6-10 AM	3-7 PM	6-10 AM	3-7 PM
2,407,128	2,625,769	2,407,567	2,613,164	439	12,604	0.0%	0.5%

(Source: San Diego I-15—Post-Deployment Analysis, Modeling and Simulation (AMS) Reasonableness Assessment and Tool Modification Technical Memorandum—Final, Cambridge Systematics, Texas A&M Transportation Institute, and Southern Methodist University, December 2015, p. 4, unpublished.)

The summary of link count reasonableness assessment results for a typical, no incident day include:

- 91 of the 93 links (97 percent) met the 15 percent comparison criterion described in Table 5 for the AM peak Criterion 1 is met for the AM Peak.
- 91 of the 96 links (94 percent) meet the 15 percent comparison criterion described in Table 5 for the PM peak period- Criterion 1 is met for the PM Peak.

- 69 of the 70 links (98 percent) meet the 15 percent comparison criterion described in Table 5 for the MD peak period Criterion 1 is met for the MD Peak.
- The sum of all model link flows across all periods 6,881,464 while the sum of observed link counts is 6,879,777. These volume sums are well within 5 percent and thus Criterion 2 is met for the three combined periods.
- The sum of all model link flows in the AM peak period is 2,407,128 while the sum of observed link counts is 2,407,567. These volume sums are within 5 percent and thus Criterion 2 is met for the AM peak period.
- The sum of all model link flows in the PM peak period is 2,625,769 while the sum of observed link counts is 2,613,164. These volume sums are within 5 percent and thus Criterion 2 is for the PM peak period.
- The sum of all model link flows in the Inter peak period is 1,848,567 while the sum of observed link counts is 1,859,046. These volume sums are within 5 percent and thus Criterion 2 is for the MD peak period.
- For all the peak periods none of the arterial counts meet the required 2,000 veh/hr, thus there is no criterion to meet. Although there are differences between observed and modeled arterial volumes these counts are all included with the model sums for each period and hence the general flow of traffic along freeways and arterials meets Criterion 2.

### **Delay, Speed and Bottleneck Comparisons**

Another component of the reasonableness assessment criteria listed in Table 4 is the visual audit of model speeds and bottlenecks. Modeled versus field-observed speeds and bottlenecks can be compared using speed contour diagrams. Figure 13 through Figure 16 compare the speed contours of southbound and northbound I-15 during a typical day, generated using detector speed data from the average of PeMS for the general-purpose lanes over the February to May 2015 period (Tuesdays and Thursdays only), and the offline simulation outputs.

Comparisons of the detector and model speed contour plots show that the model was able to represent the bottleneck temporal and spatial extents for both southbound and northbound I-15 sufficiently realistically. The comparison shows that recurring congestion exists along the freeway during the AM peak in the southbound direction and during the PM peak in the northbound direction. Modeled congestion is within acceptable thresholds for observed temporal and spatial extents of observed congestion on the I-15 freeway. One exception is the observed queuing in the observed speeds in the southbound direction during PM conditions for a Typical Day, as well as an Incident day. This congestion was not represented in the model as it was caused by a reoccurring condition downstream from the study area and is the representation of spillback from that condition.

#### Results – Incident Day

In addition to assessing the model's reasonableness for a typical day, the San Diego AMS Team also conducted a reasonableness assessment for an incident day. A PM multiple lane northbound incident for October 16<sup>th</sup>, 2014 was used, and the incident data were determined using a combination of the Aimsun Online input data and the Caltrans incident report. The incident included three right lanes closed out of a total of five lanes. The duration of the incident was from 5:30 PM for approximately one hour and 45 minutes.

Figure 17 and Figure 18 show observed and modeled speed contours for that incident day. Comparisons of the detector and modeled speed contour plots show that the model was able to sufficiently represent the bottleneck temporal and spatial extents for both northbound and southbound I-15 PM peak directions during an incident. The model also realistically captured the diversion to the arterials as some travelers looked for a faster option.

			为为为为为方方方方,所有一种种植体化和化学的分子与为不可能的建筑建筑建筑建筑的资源建筑和发展和力学和优化的人,也不可能在化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化			而方法 其於 其 法 我 27 万方 但你的 经公司 法 化化合物 建筑的 经利益 化合体化合体 化合体化合体化合体化合体化合体化合体化合体化合体化合体化合体化合体化合体化合体 化合体化合体 化合体化合体化合体化合体化合体化合体 化合体化合体 化合体化合体化合体 化合体化合体 化合体		用用剂剂剂剂剂剂 建分子方式 计描述语言 化过程	ллжллиллллиллинаасассассассассассасасасы улахуу жижую соссоссассасы жижижи жижи жижи жи	为 11 关注的 22 子 7 子 7 子 7 子 7 子 7 子 7 子 7 子 7 子 7	指示: 17 27 27 27 27 27 27 27 27 27 27 27 27 27	***************************************	加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加	用为为你的经验的原始,我们不能是这些我们有的。""你是这些我们的,我们是这些我们的是这些我们的这些我们的是不是不是不是不是不是这些我们的,我们就是不是这些我们的,我们就是不是这些我们的,我们就是不是这些 化乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基	7777777777777777784746664444446644444ccccc0394774247778782424513666666666666666666666666666666666666		非有为为非常有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有	34 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7		R R R R R B B B B B B B B B B B B B B B	推测力力力力力力力力力力力和保健的保持保持的保持的保持的保持和保持的保持的保持的保持的保持的保持的保持的保持的保持的保持的保持的保持的保持的	N         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ         Λ	17.77777777777777777777777777777777777	***************************************	为其其为为为为为为为为外的原因。""这些是是一些不是是不是有些的情况。""这些是是一些是不是不是不是不是不是不是不是不是不是不是不是不是不是不是不是不是不是	╖┍┍┍┍╖╖┍┍┍┍┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙┙	7777777%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%		
10.00 ##	64 63	4	Q #	61 6	r es	64 12	65	67 65	D	IRE	CT	101	10	F F	LO	W	64	70 6	13 64	66	66	6	65	64		ы G7	74		
20.000 20.000 20.000 20.000 20.000 20.000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.00000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.0000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.00000 20.000000 20.000000 20.00000000		N 7 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4				四月月四月四月四月四月四月四日。 化合物 医白色的 化分子 化分子 化分子分子 化合合合合合合合合合合合合合合合合合合合合合合合合	***************************************	方法考试到方用石具具有几并有的方法的工艺方方法活动的特殊方式方式方用的注意的建立的最优的方式方法方式方法方式方式方式方式方式方式方式方式方式方式方式方式方式方式方式方	为此方为此有力力只有力力力力力力力力力力力的现在,能够有效的动力力力能够有效的经常和最优化的。		有非常改善非非常没有没有不能有能力。因为有些最优的体质的实际。如果是这些实际的。如果是不是不能有的。	***************************************	用用外有方法 计利益 建设定 计可设计 化合合体 化合体化合体 化合体		有77万米东方东方东省方方计均方均从均均均均均均均均均均均均均均均均均均均均均均均均均均均均均均均均	药浆剂 预加预加预加预加 医碘化白色色 经保证 经营业价格 经价格 经公司 法法 化乙基乙基乙基乙基乙基乙基乙基乙基乙基 化合物酶酶 化合物酶酶	カ 市 市 不有 方 方 方 方 方 方 で ひ し ひ し ひ し ひ し ひ い ひ い ひ い ひ い ひ い ひ い		行行行推進行進行運動推進方法推進方法方式方式方式加減加約方法減低方行方法以低低的研究目前也是自我做到最多的分子消化的消化方法	有非非非非非非非非非非非非非非能能能能能能能能能能能能能能能能能能能能能能能能	为我为什方我有到为对方为有到大方,有大方方,更有有些变化的变化的原因。但是有有些有些有些有些有的。我们是有一个人们的是不是有一个人们的,我们就是有一个人们的,我们就是不是有一个人们的,我们就是不是有一个人	2.6.774.6.6.4.5.4.5.4.5.2.4.0.0.4.0.0.7.4.4.7.7.7.7.7.7.7.7.7.7.7	带来力打发 有有方为方法 有名有有有有有有有有有有有有有有有有的。他们都能是这些有有有有有有有有有有有有有有有有有有有有有有有有有有	P. 无行的方法是无方方方方方法并且有利用方式方式的方式方方用的方面是能够是为方式方式方式方式方式方式方式方式方式方式方式方式方式方式方式方式方式方式方式	7 形形形 77 形乃 开北东西和 21 行为 21 没的 命令 经基础 建丝 包含 经收益 建合物 医胆管 化合物 化合物 计计算机 计计算机 计计算机 化合物	7 \$ \$ 777 \$ 75 5 5 5 5 7 3 4 7 2 7 5 7 7 3 2 1 3 7 7 7 7 7 2 0 0 0 0 4 1 1 4 0 6 6 0 0 4 0 7 7 6 7 6 7 6 7 7 7 7 1 7 7 7 6 6 7 7 7 7	行拖行力分为所拖方当面为为外利利用的利力力力的加工社力为分力设备工具研究内的工具力的工力及工具设计具有外方式没有限分支利	行为乃其方殊方殊方殊方法方法为其利益加强过分和药品力以及效效效力已没效应药及可用药加强力效应并药用用药的利用药物用药物和药物和	Speed (MPH) 15 20 25 30 35 40 45 55 55 60 65 70

# Figure 13. Heatmap. Southbound I-15 AM observed (top) versus modeled (bottom) speed contours – typical day.

(Source: San Diego I-15—Post-Deployment Analysis, Modeling and Simulation (AMS) Reasonableness Assessment and Tool Modification Technical Memorandum—Final, Cambridge Systematics, Texas A&M Transportation Institute, and Southern Methodist University, December 2015, p. 11, unpublished.)

| DIRECTION OF FLOW                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Speed<br>(MPH)<br>520<br>25<br>30<br>35<br>40 |                   | 3.体体结核的体结的体结的体结的体结的的体的的体积为物的物的体积的体积的体积的体积的体积的体积的体积的体积的体积体结构体结构体结构体结构体。<br>1.体体结核的体结体体的体结的体结的体积的体积的体积的体积的体积的体积的体积的体积的体积的体积和体验体和体验体体和 | *************************************** | 。 然后接近然站行出的的的行动标签这份标准的标款的保持的保持的保持的保持可以有达达的公司有达的自己和社会过过这计划的的以且因用,在这个人们在这些不是不是不是不是不是不是不是不能能在这些的。 | 2为为方方方方方方方方方方方方方方方方方方方方方方方方方方方方方方方方方方方 |                   | 5. 细胞酸脂为为力力力加强化体验检验检验检验检验检验检验检验检验检检验检检检检检检检检检检检检检检检检检 |  | 17777717177777777777777777777777777777 | 为为先先为为为为为为为为为为为为为为为,不是不是不是不是不是不是不是不是不是不是不是不是 |  |  |  |  |  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | <pre>Speed<br/>Content of the second seco</pre>                                                                                                                                                                                                                                                                                                                                                                                              
                                                                                                                                                                                                                             |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------|-------------------|-------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------|------------------------------------------------------------------------------------------------|----------------------------------------|-------------------|-------------------------------------------------------|--|----------------------------------------|----------------------------------------------|--|--|--|--|--|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| 77777777777777777777777777777777777777                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                               | DIRECTION OF FLOW |                                                                                                                                     |                                         | DIRECTION OF FLOW                                                                              |                                        | DIRECTION OF FLOW |                                                       |  |                                        |                                              |  |  |  |  |  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | N         1         4         N         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         10         <                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | | | | | | | | | | | | | | | | | | | | | | | | | | | | | | |
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 2:15 AAA<br>3:25 AAA<br>3:20 AAA<br>4:20 AAA<br>4:2 | \$255 AND<br>\$210 AND<br>\$235 AND<br>\$245 AND<br>\$245 AND<br>\$245 AND<br>\$255 A |
| SDAMA         73         73         73         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74         74 |                                               |                   |                                                                                                                                     |                                         |                                                                                                |                                        |                   |                                                       |  |                                        |                                              |  |  |  |  |  | Samak         No.         No.< | Samak         No         No        No         No | Scheek         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N </td <td>Spann         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N<td>Spann         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N<td>Spanne         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N<td>Spanne         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N<td>Spanne         T         T         B         T         B         T         B         T         B         T         B         T         B         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D<td>Signal         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi</td><td>Solaw         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P<td>SCALE         73         71         6           SCALE         73         71         6           SCALE         73         71         6         6           SCALE         73         71         6         6           SCALE         73         71         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         <t< td=""><td>SSSAM         73         71           SSSAM         73         73           SSSAM         73         60           SSSAM         60         60</td><td>5:00 AMR           5:00 AMR           <t< td=""><th></th><td></td></t<></td></t<></td></td></td></td></td></td></td> | Spann         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N <td>Spann         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N<td>Spanne         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N<td>Spanne         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N<td>Spanne         T         T         B         T         B         T         B         T         B         T         B         T         B         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D<td>Signal         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi</td><td>Solaw         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P<td>SCALE         73         71         6           SCALE         73         71         6           SCALE         73         71         6         6           SCALE         73         71         6         6           SCALE         73         71         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         <t< td=""><td>SSSAM         73         71           SSSAM         73         73           SSSAM         73         60           SSSAM         60         60</td><td>5:00 AMR           5:00 AMR           <t< td=""><th></th><td></td></t<></td></t<></td></td></td></td></td></td> | Spann         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N <td>Spanne         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N<td>Spanne         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N<td>Spanne         T         T         B         T         B         T         B         T         B         T         B         T         B         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D<td>Signal         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi</td><td>Solaw         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P<td>SCALE         73         71         6           SCALE         73         71         6           SCALE         73         71         6         6           SCALE         73         71         6         6           SCALE         73         71         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         <t< td=""><td>SSSAM         73         71           SSSAM         73         73           SSSAM         73         60           SSSAM         60         60</td><td>5:00 AMR           5:00 AMR           <t< td=""><th></th><td></td></t<></td></t<></td></td></td></td></td> | Spanne         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N <td>Spanne         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N<td>Spanne         T         T         B         T         B         T         B         T         B         T         B         T         B         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D<td>Signal         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi</td><td>Solaw         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P<td>SCALE         73         71         6           SCALE         73         71         6           SCALE         73         71         6         6           SCALE         73         71         6         6           SCALE         73         71         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         <t< td=""><td>SSSAM         73         71           SSSAM         73         73           SSSAM         73         60           SSSAM         60         60</td><td>5:00 AMR           5:00 AMR           <t< td=""><th></th><td></td></t<></td></t<></td></td></td></td> | Spanne         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N         N <td>Spanne         T         T         B         T         B         T         B         T         B         T         B         T         B         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D<td>Signal         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi</td><td>Solaw         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P<td>SCALE         73         71         6           SCALE         73         71         6           SCALE         73         71         6         6           SCALE         73         71         6         6           SCALE         73         71         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         <t< td=""><td>SSSAM         73         71           SSSAM         73         73           SSSAM         73         60           SSSAM         60         60</td><td>5:00 AMR           5:00 AMR           <t< td=""><th></th><td></td></t<></td></t<></td></td></td> | Spanne         T         T         B         T         B         T         B         T         B         T         B         T         B         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D         D <td>Signal         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi</td> <td>Solaw         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P<td>SCALE         73         71         6           SCALE         73         71         6           SCALE         73         71         6         6           SCALE         73         71         6         6           SCALE         73         71         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         <t< td=""><td>SSSAM         73         71           SSSAM         73         73           SSSAM         73         60           SSSAM         60         60</td><td>5:00 AMR           5:00 AMR           <t< td=""><th></th><td></td></t<></td></t<></td></td> | Signal         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi         Pi         Pi         Pi         Pi         Pi           Signal         Pi         Pi | Solaw         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P         P <td>SCALE         73         71         6           SCALE         73         71         6           SCALE         73         71         6         6           SCALE         73         71         6         6           SCALE         73         71         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         <t< td=""><td>SSSAM         73         71           SSSAM         73         73           SSSAM         73         60           SSSAM         60         60</td><td>5:00 AMR           5:00 AMR           <t< td=""><th></th><td></td></t<></td></t<></td> | SCALE         73         71         6           SCALE         73         71         6           SCALE         73         71         6         6           SCALE         73         71         6         6           SCALE         73         71         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6         6 <t< td=""><td>SSSAM         73         71           SSSAM         73         73           SSSAM         73         60           SSSAM         60         60</td><td>5:00 AMR           5:00 AMR           <t< td=""><th></th><td></td></t<></td></t<> | SSSAM         73         71           SSSAM         73         73           SSSAM         73         60           SSSAM         60         60 | 5:00 AMR           5:00 AMR <t< td=""><th></th><td></td></t<> |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |

# Figure 14. Heatmap. Northbound I-15 AM observed (top) versus modeled (bottom) speed contours – typical day.

(Source: San Diego I-15—Post-Deployment Analysis, Modeling and Simulation (AMS) Reasonableness Assessment and Tool Modification Technical Memorandum—Final, Cambridge Systematics, Texas A&M Transportation Institute, and Southern Methodist University, December 2015, p. 12, unpublished.)

|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 计不可能 化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化      | 每日外的外的外的外的外的的的有些不是有不是有不是有不是有不是有不是有不是有不是有不是不不不不不不不不不不                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
***************************************                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 43 化合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合                                                      | 的 达 教 机体 化 仁 仁 仁 仁 仁 仁 仁 仁 仁 仁 仁 仁 桂 桂 桂 桂 桂 桂 桂                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 经经验价值 法律证 的复数分子 化化合物 化合物 化合物 化合物 化合物 化合物 化合物 化合物 化合物 化合 | Si S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 结晶体体的存存存存存存存在存在体体体体体体体体体体体体体体体体的体力为为为为为为为为为为为力力力力力力为力力为为为为为为为为                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 的政治政策等的政治和政治政治政治政治政治政治保障保護委員会的原因的保障部署的政治政治政治政治政治政治政治政治政治政治政治政治政治                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 660000年後後後後後後後後後後後後後後後後後後後後後後後後後後の約約約約約約約約約約                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 科林县代结结结结结核核核核核核结结合合体的 计计算机保健保计计编辑器组织的的物物的的物物和物的力物的力物物的的物物的的的力        | n an                           | 统体行行行行行使体体体体体体体体体体体体体体体体体体体体体体体体体体的功能的方式方式为力力力力力力力力力力力力力力力力                        | 经保持股口的口口口口的有效的过去式和过去分词的 建化合物 化化合物 化合物的合物合物合物合物合物合物合物合物合物合物合物合物 化合物合物 | 科科达达公式和林林林林林林林林林林林林林和林和学校和学校和学校和教教和教教教教教教教教教教教                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 的品质结核的化化存在存在存在存在存在,在在在有有有有有有有有有有有有有有有有有有有有有有有有                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 结核结核结核结核结核结核结核结核结核结核结核结核结核结核结核结核结核结核结核                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 13155555555555555555555555555555555555 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 5%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%                                                    | 13月13日秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋秋                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 外的的利用的有的的复数形式的复数形式的复数形式的复数形式的复数形式的复数形式的复数形式的复数形                                              | <b>结结结体的体体的体体的体体的体体的体体的存在的分子的分子的分子的分子的分子的分子的一种分子的体体的体体的有效性的物体的物质的分子的</b> | 4 # # # # # # # # # # # # # # # # # # #                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                             | 20002888888888888888888888888888888888                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 为为为为为为为为为方方方方方方方方方方方方方方方方方方方方方方方方方方方方 |                                                                         |                                                                                                                |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------|--------------------------------------------------------------------|------------------------------------------------------------------------------------|----------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|--------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------|-------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |   
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                         | C                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | DIR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | EC                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | TI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | ٥N                                                                   | 1 C                                                                | )F                                                                                 | FL                                                                   | -0'                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | W                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                              |                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                                                                         |                                                                                                                |
| 205.00 PM         345           230.00 PM         70.0           240.00 PM         70.0           255.00 PM         70.0           330.00 PM         70.0           340.00 PM         70.0      < | 7184<br>71977777777777777777777777777777777777 | 74.0         74.7           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.7         74.8           74.8         74.8           74.7         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.8         74.8           74.9         74.8           74.8         74.8           74.9         74.8           74.9         74.8           74.9         74.8           74.9         74.9           74.9         74.9           74.9         74.9           74.9         74.9           74.9         74.9           74.9 <td>36.1         40.5           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         43.4           31.3         43.4           31.3         43.4           31.4         41.4           31.4         44.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.4         43.4           31.5         43.4           31.6         43.4           31.7         43.4           31.8<td>1 03.1<br/>5 3 52.2<br/>5 77.2<br/>5 77.2<br/>6 77.2<br/>6 77.2<br/>6 77.2<br/>7 9 54.4<br/>7 9 54.4<br/>7 9 54.4<br/>7 9 54.4<br/>9 900 7<br/>7 9 54.4<br/>9 900 7<br/>7 9 54.4<br/>9 900 7<br/>7 9 54.4<br/>9 900 7<br/>9 94.2<br/>9 900 7<br/>9 94.4<br/>9 94.4<br/>94.4<br/>9 94.4<br/>9 94</td><td>415<br/>463<br/>995<br/>915<br/>464<br/>464<br/>464<br/>464<br/>464<br/>464<br/>464<br/>464<br/>464<br/>46</td><td>2         41.3           2         50.7           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           4         54.5           4         54.5           4         54.6           5         54.7           6         54.7           6         54.7           7         64.4           6         53.3           6         54.3           6         54.3           6         54.2           7         64.4           8         54.2           6         64.2           7         64.4           8         54.2           8         64.2</td><td>200 (044) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (0</td><td>R4 の 20 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2</td><td>7 73,8<br/>9 73,4<br/>9 73,4<br/>9 73,4<br/>9 73,4<br/>9 73,4<br/>1 74,0<br/>1 74,0</td><td>21.9.3         21.9.3           21.9.4         21.9.4           21.9.4         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4          
21.9.7</td><td>2211337<br/>X70.24<br/>X70.24<br/>X70.74<br/>X70.74<br/>X70.74<br/>X70.74<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X</td><td>34.74<br/>74.74<br/>74.77<br/>74.77<br/>74.77<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75</td><td>22.3<br/>77.7<br/>77.8<br/>77.7<br/>77.8<br/>77.1<br/>77.8<br/>77.1<br/>77.1</td><td>34.1<br/>71.8<br/>71.8<br/>71.7<br/>71.7<br/>71.7<br/>71.7<br/>71.7<br/>71</td><td>72.5<br/>34.0<br/>34.2<br/>37.3<br/>37.3<br/>37.3<br/>37.3<br/>37.3<br/>37.3<br/>37.3<br/>37</td><td>76.77.77.77.77.77.77.77.77.77.77.77.77.7</td><td>603<br/>71.4<br/>805<br/>67.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>8</td><td>34.3<br/>75.4<br/>75.4<br/>75.4<br/>75.4<br/>75.4<br/>75.4<br/>75.4<br/>75.4</td><td>68.93<br/>5
8.93<br/>58.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59</td><td>7.4.4         2           7.22.4         6           7.23.5         6           7.23.5         6           7.24.6         6           7.25.7         6           7.26.8         6           7.27.1         6           7.28.7         7.29           7.29.7         7.21           7.20.7         7.20           7.21.7         7.21           7.22.7         7.21           7.23.7         7.22           7.23.7         8           7.24.7         7.22           7.22.7         8           7.23.8         7.22           8         7.24           9.21.7         7.22           9.21.7         7.22           9.21.8         8           9.21.9         7.22           9.21.9         7.22           9.22         7.22           9.23.7         7.22           9.24.8         8           9.25.9         7.22           9.21.9         7.22           9.22.9         7.23           9.23.9         7.23           9.24.9         7.23</td><td>3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         6           4.41         6           4.43         6           4.43         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         7           7         7           7.8         7           7.7         7           7.7</td><td></td><td>86.6           86.7           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.1           97.2           97.3           97.3           97.3           97.3           97.3           97.4           97.5           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.8           97.9           97.9           97.9</td><td>7132<br/>6033<br/>6034<br/>6037<br/>7100<br/>7100<br/>7100<br/>7100<br/>7100<br/>7100<br/>7100<br/>7</td><td>71.3           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.27           71.41           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27<td>75.1<br/>74.9<br/>74.9<br/>74.9<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0</td><td>343<br/>738<br/>738<br/>737<br/>737<br/>737<br/>737<br/>737<br/>737<br/>737<br/>73</td><td>14.1           17.3           17.3           17.3           17.3           17.3           17.3           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4</td><td>5.0 5 7 7 7 7 7 7 7 6 6 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9</td><td>3.5         71           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.5         71.7           3.6         71.7           3.7         71.7           3.8         71.7           3.9         71.7           3.10         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.11         71.7           3.11         71.7           3.11         71.7</td><td></td><td>344<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>7</td><td>Speed<br/>(MPH)<br/>15<br/>20<br/>25<br/>30<br/>35<br/>30<br/>35<br/>40<br/>45<br/>50<br/>55<br/>50<br/>65<br/>60<br/>65<br/>70</td></td></td> | 36.1         40.5           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         42.4           31.3         43.4           31.3         43.4           31.3         43.4           31.4         41.4           31.4         44.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.3         43.4           31.4         43.4           31.5         43.4           31.6         43.4           31.7         43.4           31.8 <td>1 03.1<br/>5 3 52.2<br/>5 77.2<br/>5 77.2<br/>6 77.2<br/>6 77.2<br/>6 77.2<br/>7 9 54.4<br/>7 9 54.4<br/>7 9 54.4<br/>7 9 54.4<br/>9 900 7<br/>7 9 54.4<br/>9 900 7<br/>7 9 54.4<br/>9 900 7<br/>7 9 54.4<br/>9 900 7<br/>9 94.2<br/>9 900 7<br/>9 94.4<br/>9 94.4<br/>94.4<br/>9 94.4<br/>9 94</td> <td>415<br/>463<br/>995<br/>915<br/>464<br/>464<br/>464<br/>464<br/>464<br/>464<br/>464<br/>464<br/>464<br/>46</td> <td>2         41.3           2         50.7           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4      
    3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           4         54.5           4         54.5           4         54.6           5         54.7           6         54.7           6         54.7           7         64.4           6         53.3           6         54.3           6         54.3           6         54.2           7         64.4           8         54.2           6         64.2           7         64.4           8         54.2           8         64.2</td> <td>200 (044) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (0</td> <td>R4 の 20 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2</td> <td>7 73,8<br/>9 73,4<br/>9 73,4<br/>9 73,4<br/>9 73,4<br/>9 73,4<br/>1 74,0<br/>1 74,0</td> <td>21.9.3         21.9.3           21.9.4         21.9.4           21.9.4         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7</td> <td>2211337<br/>X70.24<br/>X70.24<br/>X70.74<br/>X70.74<br/>X70.74<br/>X70.74<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X70.70<br/>X</td>
<td>34.74<br/>74.74<br/>74.77<br/>74.77<br/>74.77<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.74<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75<br/>75.75</td> <td>22.3<br/>77.7<br/>77.8<br/>77.7<br/>77.8<br/>77.1<br/>77.8<br/>77.1<br/>77.1</td> <td>34.1<br/>71.8<br/>71.8<br/>71.7<br/>71.7<br/>71.7<br/>71.7<br/>71.7<br/>71</td> <td>72.5<br/>34.0<br/>34.2<br/>37.3<br/>37.3<br/>37.3<br/>37.3<br/>37.3<br/>37.3<br/>37.3<br/>37</td> <td>76.77.77.77.77.77.77.77.77.77.77.77.77.7</td> <td>603<br/>71.4<br/>805<br/>67.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>807.4<br/>8</td> <td>34.3<br/>75.4<br/>75.4<br/>75.4<br/>75.4<br/>75.4<br/>75.4<br/>75.4<br/>75.4</td> <td>68.93<br/>58.93<br/>58.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59.97<br/>59</td> <td>7.4.4         2           7.22.4         6           7.23.5         6           7.23.5         6           7.24.6         6           7.25.7         6           7.26.8         6           7.27.1         6           7.28.7         7.29           7.29.7         7.21           7.20.7         7.20          
7.21.7         7.21           7.22.7         7.21           7.23.7         7.22           7.23.7         8           7.24.7         7.22           7.22.7         8           7.23.8         7.22           8         7.24           9.21.7         7.22           9.21.7         7.22           9.21.8         8           9.21.9         7.22           9.21.9         7.22           9.22         7.22           9.23.7         7.22           9.24.8         8           9.25.9         7.22           9.21.9         7.22           9.22.9         7.23           9.23.9         7.23           9.24.9         7.23</td> <td>3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         6           4.41         6           4.43         6           4.43         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         7           7         7           7.8         7           7.7         7           7.7</td> <td></td> <td>86.6           86.7           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.1           97.2           97.3           97.3           97.3           97.3           97.3           97.4           97.5           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.8           97.9           97.9           97.9</td> <td>7132<br/>6033<br/>6034<br/>6037<br/>7100<br/>7100<br/>7100<br/>7100<br/>7100<br/>7100<br/>7100<br/>7</td> <td>71.3           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.27           71.41           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27<td>75.1<br/>74.9<br/>74.9<br/>74.9<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0</td><td>343<br/>738<br/>738<br/>737<br/>737<br/>737<br/>737<br/>737<br/>737<br/>737<br/>73</td><td>14.1           17.3           17.3           17.3           17.3           17.3           17.3           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4</td><td>5.0 5 7 7 7 7 7 7 7 6 6 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9</td><td>3.5         71           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.5         71.7           3.6         71.7           3.7         71.7           3.8         71.7           3.9         71.7           3.10         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.11         71.7           3.11         71.7           3.11         71.7</td><td></td><td>344<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>7</td><td>Speed<br/>(MPH)<br/>15<br/>20<br/>25<br/>30<br/>35<br/>30<br/>35<br/>40<br/>45<br/>50<br/>55<br/>50<br/>65<br/>60<br/>65<br/>70</td></td> | 1 03.1<br>5 3 52.2<br>5 77.2<br>5 77.2<br>6 77.2<br>6 77.2<br>6 77.2<br>7 9 54.4<br>7 9 54.4<br>7 9 54.4<br>7 9 54.4<br>9 900 7<br>7 9 54.4<br>9 900 7<br>7 9 54.4<br>9 900 7<br>7 9 54.4<br>9 900 7<br>9 94.2<br>9 900 7<br>9 94.4<br>9 94.4<br>94.4<br>9 94.4<br>9 94 | 415<br>463<br>995<br>915<br>464<br>464<br>464<br>464<br>464<br>464<br>464<br>464<br>464<br>46 | 2         41.3           2         50.7           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           3         54.4           4         54.5           4         54.5           4         54.6           5         54.7           6         54.7           6         54.7           7         64.4           6         53.3           6         54.3           6         54.3           6         54.2           7         64.4           8         54.2           6         64.2           7         64.4           8         54.2           8         64.2 | 200 (044) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (045) (0 | R4 の 20 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2               | 7 73,8<br>9 73,4<br>9 73,4<br>9 73,4<br>9 73,4<br>9 73,4<br>1 74,0<br>1 74,0 | 21.9.3         21.9.3           21.9.4         21.9.4           21.9.4         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7         21.9.4           21.9.7 |
2211337<br>X70.24<br>X70.24<br>X70.74<br>X70.74<br>X70.74<br>X70.74<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X70.70<br>X | 34.74<br>74.74<br>74.77<br>74.77<br>74.77<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.74<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75<br>75.75 | 22.3<br>77.7<br>77.8<br>77.7<br>77.8<br>77.1<br>77.8<br>77.1<br>77.1 | 34.1<br>71.8<br>71.8<br>71.7<br>71.7<br>71.7<br>71.7<br>71.7<br>71 | 72.5<br>34.0<br>34.2<br>37.3<br>37.3<br>37.3<br>37.3<br>37.3<br>37.3<br>37.3<br>37 | 76.77.77.77.77.77.77.77.77.77.77.77.77.7                             | 603<br>71.4<br>805<br>67.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>807.4<br>8 | 34.3<br>75.4<br>75.4<br>75.4<br>75.4<br>75.4<br>75.4<br>75.4<br>75.4 |
68.93<br>58.93<br>58.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59.97<br>59 | 7.4.4         2           7.22.4         6           7.23.5         6           7.23.5         6           7.24.6         6           7.25.7         6           7.26.8         6           7.27.1         6           7.28.7         7.29           7.29.7         7.21           7.20.7         7.20           7.21.7         7.21           7.22.7         7.21           7.23.7         7.22           7.23.7         8           7.24.7         7.22           7.22.7         8           7.23.8         7.22           8         7.24           9.21.7         7.22           9.21.7         7.22           9.21.8         8           9.21.9         7.22           9.21.9         7.22           9.22         7.22           9.23.7         7.22           9.24.8         8           9.25.9         7.22           9.21.9         7.22           9.22.9         7.23           9.23.9         7.23           9.24.9         7.23 | 3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         7           3.8         6           4.41         6           4.43         6           4.43         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         6           6.8         7           7         7           7.8         7           7.7         7           7.7 |                                        | 86.6           86.7           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.0           97.1           97.2           97.3           97.3           97.3           97.3           97.3           97.4           97.5           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.7           97.8           97.8           97.9           97.9           97.9 | 7132<br>6033<br>6034<br>6037<br>7100<br>7100<br>7100<br>7100<br>7100<br>7100<br>7100<br>7 | 71.3           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.25           72.27           71.41           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27           72.27 <td>75.1<br/>74.9<br/>74.9<br/>74.9<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0<br/>75.0</td> <td>343<br/>738<br/>738<br/>737<br/>737<br/>737<br/>737<br/>737<br/>737<br/>737<br/>73</td> <td>14.1           17.3           17.3           17.3           17.3           17.3           17.3           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4</td> <td>5.0 5 7 7 7 7 7 7 7 6 6 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9</td> <td>3.5         71           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.5         71.7           3.6         71.7           3.7         71.7           3.8         71.7           3.9         71.7           3.10         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.11         71.7           3.11         71.7           3.11         71.7</td> <td></td> <td>344<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>745<br/>7</td> <td>Speed<br/>(MPH)<br/>15<br/>20<br/>25<br/>30<br/>35<br/>30<br/>35<br/>40<br/>45<br/>50<br/>55<br/>50<br/>65<br/>60<br/>65<br/>70</td> | 75.1<br>74.9<br>74.9<br>74.9<br>75.0<br>75.0<br>75.0<br>75.0<br>75.0<br>75.0<br>75.0<br>75.0 | 343<br>738<br>738<br>737<br>737<br>737<br>737<br>737<br>737<br>737<br>73 | 14.1           17.3           17.3           17.3           17.3           17.3           17.3           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4           17.4 | 5.0 5 7 7 7 7 7 7 7 6 6 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 | 3.5         71           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.4         41           3.5         71.7           3.6         71.7           3.7         71.7           3.8         71.7           3.9         71.7           3.10         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.12         71.7           3.13         71.7           3.14         71.7           3.15         71.7           3.11         71.7           3.11         71.7           3.11         71.7
          3.11         71.7 |                                       | 344<br>745<br>745<br>745<br>745<br>745<br>745<br>745<br>745<br>745<br>7 | Speed<br>(MPH)<br>15<br>20<br>25<br>30<br>35<br>30<br>35<br>40<br>45<br>50<br>55<br>50<br>65<br>60<br>65<br>70 |

# Figure 15. Heatmap. Southbound I-15 PM observed (top) versus modeled (bottom) speed contours – typical day.

(Source: San Diego I-15—Post-Deployment Analysis, Modeling and Simulation (AMS) Reasonableness Assessment and Tool Modification Technical Memorandum—Final, Cambridge Systematics, Texas A&M Transportation Institute, and Southern Methodist University, December 2015, p. 13, unpublished.)

200 hills 220 hi	力物剂的药物药物的盐酸盐盐酸盐酸盐酸盐酸盐酸盐酸盐医酸盐酸盐酸盐酸盐酸盐酸盐酸盐酸盐酸盐酸盐酸盐	***************************************	***************************************	经有价价的 化合合物 化合合物 化合合物 化合物合物合物合物 化加加加加加加加加加加加加加	的复数形式 化化合物 化合物 化合物 化化合物 化化合物 计分子 化合物化合物 化加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加加	64 A A A A A A A A A A A A A A A A A A A	化非常不能能能加强 化化化化化 化化化化化化化化化化化化化化化化化化化化化化化化化化化化	化化物物的物物的物物物物物物的复数 化分子子 计计算机 化分子分子 化合金化合金化合金化合金化合金化合金化合金化合金化物物物	상당한 동료 속은 유럽 유럽 유럽 유럽 유럽 유럽 가슴이 있는 것을 같은 것을 수 있는 것을 가지 않는 것을 같은 것을 하는 것을 같은 것을 알았다. 것을 알았는 것을 많은 것을 같은 것을 같 것을 같	医外部 医外部 化化学学 化化学学 化化学学 化化学学 化化学学 化化学 化化学 化化学 化	***************************************	林林 经合约保存额 网络 网络拉拉 化分子分子 建建合金 计算机 化分子 建氯化合金 化合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合合		***************************************	非非非非常有效的 化合物 化合物合物 化合物 化分子 化合物			# # # # # # # # # # # # # # # # # # #	经经过保证保证 经有利利 化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化	林林然后我的最优级的现在分词 22 中部的 网络的 网络印刷银印刷银印刷银石印刷银银印刷银银印刷制品的复数 化硫酸化硫酸化硫酸化硫酸化硫酸	林泉县市市部委员会部委员会的资格的公司 化合金	经财务利益的利益利益 网络拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉拉	86.85.65.65.85.85.64.44.45.81.87.61.63.62.62.62.62.62.62.62.62.62.62.62.62.62.	86 6 6 7 7 7 7 8 8 9 9 1 9 1 9 1 9 1 9 1 9 1 9 1 9 1 9	的特别的特别的特别的特别来到了计算法通道的长行,在这些是有一些有些有些有些无法无法无法的保存的现在分词 计计数据的过程设计过程设计过程设计计算的		***************************************	***************************************	0.000000000000000000000000000000000000	00 00 00 00 00 00 00 00 00 00 00 00 00			医外肠膜炎 网络拉拉拉 化化化化化化化化化化化物 化化物物 化化物物物 化化物物物 化化化化物化化化化化化	
											DI	RE	C	ΓIC	N	OF	F	LO	W															
105         105           105         106           105         106           105         106           105         106           105         106           105         106           105         106           105         106           105         106           105         106           105         106           105         106           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105         107           105	27 77 77 77 10 44 45 46 46 46 47 77 77 77 77 77 77 77 77 77 77 77 77	212124666699646855544498416056534896743972227404899298687882128921898686982122782727860122222222242		544 610 555 557 557 557 557 557 557 557 557 55	的什么你的外球,我们就是这些不是一个是一个是一个是一个是一个是一个是一个是一个是一个是一个是一个是一个是一个是			加力推加時期就最終的。但是了公司的自分自动自己的保险地位,就是这些分别就是不是这些自己都已经加热的保险和优化的自己的保健的人力力推	11.71 71 72 77 77 77 77 77 77 77 77 77 77 77 77	9 1 1 1 1 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	从27万方为和34万27九27月21月7月38日66月16日996日4507日6596日48日44099日43日15798日4599日4550日458日459日777元6777%34343	илиматили и поли по	***************************************	¥5¥78¥753¥2735227322752242222222222222222222222222		77.77.77.77.8644 42.864 42.31.25.27.27.29.33.39.43.49.57.52.42.42.44.46.49.44.32.36.32.44.45.35.55.57.53.55.83.44.55.46.44.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.40.45.36.86.46.45.36.86.36.36.36.36.36.36.36.36.36.36.36.36.36	57 00 Mi 44 47 70 70 71 71 71 72 72 72 72 72 72 72 72 72 72 72 72 72	66 67 67 69 72 77 76 64 66 55 66 66 57 77 76 64 66 55 66 66 77 77 76 64 66 55 66 66 66 77 77 76 64 66 55 66 66 66 77 76 66 76 66 77 76 66 76 66 77 76 66 76 66 77 76 66 76 66 77 76 66 76 66 77 76 76	51 53 53 53 53 53 54 55 55 55 55 55 55 55 55 55	17 # 18 # 18 19 19 19 19 19 19 19 19 19 19 19 19 19	25 行头为 74 为 24 为为为为为为为为为为为为为 25 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 7 7 7 7	27.77.14.66.87.77.27.27.77.16.67.61.69.94.67.61.67.67.67.67.67.67.67.67.67.67.77.77.77.	维朗姆格尔舒斯格林和利利格拉和利伯的领地分为分分分为中针针导导导力为为推进能力和方量并是54444456439390的新研究的利益已经分分	为为的最存储最低的最好的价格或以分价。 化化分子分子分子分子分子分子分子分子分子分子分子分子分子分子的合合物为有限的存储在的方面的	使多生气的的变形的现在分词 建设化学 化化化物 化过程性 化过程性 化过程分 化过程分组合 经过程序的 医结核体 医马达林 医子外腺 医外周的 化乙基 医马克尔氏 化乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基	99997110714771107112000000000000000000000000	12 72 73 73 73 77 72 72 72 72 72 72 72 72 72 72 72 72	22734777777722777227776666666672777777887082277777687072277788727222228727777777777			<sup>3</sup> <sup></sup>	落地力也有有方利和有方利的有利的利利利利的利利的利利的现在分词的利利的现在分词的利利的现在分词有有方利的方利的方利的有效的有利的	其外外不是其外的其外和其外,并且不是有利用,以且在方力可以在这些方力方力在方力方式在有利用,并有利用的有利的方式和方式和方式和方式	Speed (MPH) 15 20 25 30 35 40 45 50 55 60 65 60 65 70

# Figure 16. Heatmap. Northbound I-15 PM observed (top) versus modeled (bottom) speed contours – typical day.

(Source: San Diego I-15—Post-Deployment Analysis, Modeling and Simulation (AMS) Reasonableness Assessment and Tool Modification Technical Memorandum—Final, Cambridge Systematics, Texas A&M Transportation Institute, and Southern Methodist University, December 2015, p. 14, unpublished.)

| $\begin{array}{c} 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 $                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | "这些情情的"你是这时好我就的好好也是你能好好好好!"他的想知我的儿子的想想是我的是我们的情况就是你们们不能的我的的是你们都                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 24433444677884466744444666744444866877446674869884877897788466664644666746748846674466674484667448466                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
7、100011712以下171711月,建築建築部分に建築建築は1718基金建築合成の10001170111月11日、171211月11日、171211月11日、171211月11日、171211月11日、171211月11日、171211月11日、171211月11日、171211月11日、171211月11日、171211月11日、17121日、17121日、17121日、17121日、17121日、17121日、17121日、17121日、17121日、17121日、17121日、17121日                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | "我就了我了我们了了了了你就是我们?我我就是我们的我的是我的的吗?""你们是我们的我们的,我们们有什么?""你是你们的我们的吗?""你们们有什么?""你们是我们的,我们们有什么?""你们是我们的,我们们                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 5.44013440043100330033043454840988888888888888500000455858888888888                                                                                                                       | S体好好好好好好能够做和的体体体的的的形态的的的物质和和和对方方方方方式的和且不知知和的的。例如如何有力力的和力力力力和力力力力                                                                                                                                             | 10444555646554746727764564544647787477774777777777777777777                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1992年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997年4月1日<br>1997<br>1997<br>1997<br>1997<br>1997<br>1997<br>1997<br>19 | ***************************************                                                    | s 46 46 46 47 47 47 47 97 99 49 49 49 47 70 70 70 70 70 70 71 72 73 44 48 40 44 44 44 45 47 49 49 47 47 49 49 47 70 49 49 70 71 72 73 44 48 40 44 44 44 45 47 49 49 47 47 49 49 47 70 49 49 70 71 72 73 44 48 40 44 44 45 45 47 49 47 47 49 49 47 70 49 49 71 72 73                                  | 3. 单心的 化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化                                                                                                                                                                                                           | 2.47亿纪纪体统结婚组织为地方为地地为地地方为师师方式的结婚为力力力力力力力力的地地地和地力地的维修体结结地地地地力力力力力力                                                                     | 3.日花花时打打铺模好打块铺铺铺铺的的和和的月光模模铺和和和为月月月月月月月的和和和那种的装饰铺铺的和砂的的的的用罐的和砂                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 【林氏科林林林林林林林林林林林林林林林林林林林林林林林林林林林林林林林林林林林                                                                                           | 14.07年纪时结婚的保佑派的结婚的为婚师为为为几九为为九为为九九万九万万万万为婚姻的结核维持结核保持能力为师的为九为师为为为                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                      | 3.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4                                                                                                                                                                               | 6.我们就好好都就好好就像你的你的没有不可能的你的你的你们不能不能不能不能不能的就能好你不可能的你的你们不能不能不能不能的                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 可以有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有有                                                                                                                      | 5.化行畅畅依然行行行政制行处理保备确保创建体研的极行处理和形式方式力力及使保健行行物确保保健研究为方式方式力力力力为为                                                                                                                                  
                                                                                         | 7.000000000000000000000000000000000000                                                                                                                                                       | 1996-04-04-04-04-04-04-04-04-04-04-04-04-04-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 34444444444444444444444444444444444444                                                                   | · 外外的外的外的外的。 一個一個一個一個一個一個一個人的一個人的一個人的一個人的一個人的一個人的一個人的                                                       | 19.化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 3.打打的的影响的是有的现在分词有有有有有有有有有的。我们是有些有些有些的有效。我们是不是有的有的。我们有有多少的。                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                | 中于了先来来在来来了了来在来来了这些在这些你们没有对了这些你的。我们都是有些自己的人们的,我们都是有些有些人们的情况,我们就是有些有些的。" 化乙酰基 化化化乙酰基 化化化乙酰基 化化化乙酰基 化化化乙酰基 化化化乙酰基 化分子子 计分子子 化分子子 化分子子 化分子子 化分子子 化分子子 化分子子                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |   |                                                                |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	----------------------------------------------------------------------------------------------------------	-------------------------------------------------------------------------------------------------------------	-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
3.05.00 Ph	81	ME	N3	415	64.7 59
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 5.4 004                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 03                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 613                                                                                                                                                                                       | 59.4<br>59.5                                                                                                                                                                                                 | 204                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 20.4 71:1<br>249 91                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                            | DIR                                                                                                                                                                                                                                                                                                  | 2EC                                                                                                                                                                                                                                                  |                                                                                                                                      | 10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | V C                                                                                                                               | DF                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | FL                                                                                                   | _0                                                                                                                                                                                                                    | W                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 67.1                                                                                                                                                         | 717                                                                                                                                                                                                                                                                                     |     
                                                                                                                                                                                        | 60.4 J                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 47 62                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 88.0<br>2010 - 2010                                                                                      | 743                                                                                                         | 221                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | <u>N5</u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 69.0                                                                                                                                                                                                                                                                           | 65.8                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 21.3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 0.0 223                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |   |                                                                |
| 111.000 MM<br>22500 MM<br>23500 MM<br>23500 MM<br>23500 MM<br>24500 MM<br>24500 MM<br>25500 MM<br>25500 MM<br>25500 MM<br>21500 MM<br>21500 MM<br>25500 MM<br>2550 | 75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84<br>75.84 | 744<br>744<br>745<br>745<br>745<br>745<br>745<br>745<br>745<br>745                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 73.7<br>74.7<br>74.9<br>72.8<br>72.8<br>72.8<br>72.8<br>72.8<br>72.8<br>72.8<br>72.8                                                                                                                                       | 911<br>911<br>911<br>911<br>911<br>911<br>911<br>911                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 1935         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | Joint         State           Joint <th>63.4         68.3           4         68.3           2         68.6           9         67.5           2         66.6           3         67.5           2         66.6           3         67.5           3         66.7           3         66.7           3         72.1           3         72.2       
   3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           4         64.6           6         64.5           6         64.5           6         64.5           7         64.5           7         70.2           8         64.0           6         65.2</th> <th>611<br/>611<br/>653<br/>664<br/>664<br/>664<br/>664<br/>665<br/>665<br/>665<br/>665<br/>665<br/>665</th> <th>57.3<br/>58.3<br/>60.6<br/>60.0<br/>60.0<br/>60.1<br/>60.0<br/>60.1<br/>60.2<br/>58.6<br/>61.2<br/>60.2<br/>60.6<br/>61.6<br/>61.2<br/>60.6<br/>61.6<br/>61.2<br/>60.6<br/>61.6<br/>61.6<br/>61.6<br/>61.6<br/>61.6<br/>61.6<br/>61.6</th> <th>807<br/>407<br/>408<br/>408<br/>408<br/>408<br/>408<br/>408<br/>408<br/>408</th> <th>Model         Table           Model         71.3           Model         71.3           Model         71.3           Model         71.3           Model         71.4           Model         71.4     &lt;</th> <th>710<br/>710<br/>710<br/>710<br/>710<br/>710<br/>710<br/>710<br/>710<br/>710</th> <th>73.6<br/>73.6<br/>73.6<br/>73.6<br/>74.1<br/>73.2<br/>74.1<br/>73.7<br/>73.3<br/>73.9<br/>73.3<br/>73.9<br/>73.3<br/>73.9<br/>74.0<br/>74.7<br/>74.0<br/>74.7<br/>74.0<br/>74.7<br/>74.0<br/>74.7<br/>74.0<br/>74.7<br/>74.0<br/>74.7<br/>74.0<br/>74.7<br/>74.0<br/>74.7<br/>74.1<br/>74.1<br/>74.1<br/>74.1<br/>74.1<br/>74.1<br/>74.1</th> <th>489<br/>47.4<br/>72.1<br/>72.3<br/>72.3<br/>72.4<br/>465.8<br/>465.8<br/>465.8<br/>47.8<br/>72.6<br/>72.6<br/>72.6<br/>72.6<br/>72.6<br/>72.6<br/>72.6<br/>72.6</th> <th>1248<br/>1288<br/>1445<br/>1445<br/>1445<br/>1445<br/>1445<br/>1445<br/>1445<br/>14</th> <th>11.4<br/>00.3<br/>11.1<br/>12.1<br/>10.5<br/>12.1<br/>10.5<br/>12.1<br/>10.5<br/>12.1<br/>10.5<br/>12.1<br/>10.5<br/>12.1<br/>10.5<br/>12.1<br/>10.5<br/>12.1<br/>10.5<br/>12.1<br/>10.5<br/>12.1<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>12.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5<br/>10.5</th> <th>714<br/>716<br/>718<br/>718<br/>718<br/>718<br/>718<br/>718<br/>718<br/>718<br/>718<br/>718</th> <th>619<br/>709<br/>723<br/>716<br/>710<br/>700<br/>701<br/>705<br/>717<br/>705<br/>717<br/>705<br/>717<br/>717<br/>717<br/>717<br/>717<br/>717<br/>717<br/>717<br/>717<br/>71</th> <th>224<br/>315<br/>724<br/>315<br/>712<br/>712<br/>712<br/>712<br/>712<br/>712<br/>712<br/>712<br/>712<br/>712</th> <th>013<br/>013<br/>013<br/>013<br/>013<br/>015<br/>014<br/>017<br/>015<br/>017<br/>015<br/>017<br/>015<br/>017<br/>015<br/>017<br/>015<br/>017<br/>015<br/>017<br/>015<br/>017<br/>015<br/>017<br/>017<br/>017<br/>017<br/>017<br/>017<br/>017<br/>017</th> <th>17.17<br/>18.14<br/>19.12<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19.18<br/>19</th> <th>005<br/>644<br/>645<br/>646<br/>645<br/>645<br/>645<br/>645<br/>64</th> <th>713<br/>7023<br/>7024<br/>7123<br/>722<br/>722<br/>722<br/>722<br/>722<br/>722<br/>722<br/>721<br/>724<br/>725<br/>724<br/>726<br/>727<br/>721<br/>724<br/>725<br/>724<br/>725<br/>724<br/>725<br/>724<br/>725<br/>724<br/>725<br/>725<br/>724<br/>725<br/>725<br/>725<br/>725<br/>725<br/>725<br/>725<br/>725<br/>725<br/>725</th> <th>0070<br/>0770<br/>0770<br/>0770<br/>0770<br/>0770<br/>0770<br/>077</th> <th>1013         1           662         1           662         1           662         1           705         1           868         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705</th> <th>5.4         3.5         3.6           4.4         3.6         3.7           4.4         3.7         3.7           4.4         3.7         3.7         
 5.4         1.6         5.7           5.5         5.6         5.7           5.5         5.5         5.5           5.5         5.6         5.7           5.5         5.6         5.5           5.6         5.7         5.6           5.5         5.6         5.5           5.6         5.5         5.6           5.5         5.6         5.5           5.6         5.5         5.6           5.5         5.6         5.5           5.6         5.5         5.6           5.7         6.6         5.5           5.1         5.1         5.1           5.1         5.5         5.1           5.5         5.5         5.5</th> <th>5 7.1.1<br/>6 76.4<br/>70.2<br/>2 62.5<br/>2 63.7<br/>2 63.7<br/>2 64.6<br/>7 7.1.8<br/>8 64.2<br/>7 64.0<br/>7 7 7.1.8<br/>8 64.2<br/>7 7 7 7.1.8<br/>9 7.1.0<br/>9 7.1.0</th> <th>724<br/>724<br/>724<br/>724<br/>727<br/>727<br/>727<br/>727<br/>727<br/>727</th> <th>7248<br/>7429<br/>7157<br/>711,<br/>712,<br/>713,<br/>714,<br/>714,<br/>714,<br/>714,<br/>714,<br/>714,<br/>714,<br/>714</th> <th>12.14<br/>71.71<br/>71.71<br/>70.80<br/>71.71<br/>71.74<br/>71.72<br/>72.72<br/>71.71<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72<br/>71.72</th> <th>72,44<br/>71,27<br/>71,25<br/>71,25<br/>71,27<br/>71,27<br/>72,21<br/>71,27<br/>72,21<br/>71,27<br/>72,27<br/>72,27<br/>72,27<br/>73,21<br/>73,11<br/>73,11<br/>73,11<br/>73,11<br/>73,11<br/>73,11<br/>73,11<br/>73,11<br/>73,11<br/>73,12<br/>73,12<br/>73,12<br/>73,12<br/>73,12<br/>73,12<br/>73,12<br/>73,12<br/>73,12<br/>73,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12<br/>74,12,</th> <th>213<br/>735<br/>735<br/>735<br/>735<br/>737<br/>737<br/>856<br/>80<br/>80<br/>754<br/>735<br/>736<br/>80<br/>754<br/>735<br/>736<br/>80<br/>754<br/>735<br/>735<br/>736<br/>80<br/>754<br/>735<br/>735<br/>735<br/>736<br/>80<br/>754<br/>735<br/>735<br/>735<br/>735<br/>735<br/>735<br/>735<br/>735<br/>735<br/>735</th> <th>006         1           006         1           008         0           008         0           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011</th> <th>17         2           17         2           17         2           1811         1           1872         2           1872         2           1872         2           1872         2           1872         2           1872         2           1872         2           1872         2           1873         2           1874         2           1873         2           1873         2           1873         2           1873         2           1874         2           1875         2           1874         2           1874         2           1874         2           1874         2           1875         3           1884         3           1884         3           1884         3           1975         3           1975         3           1975         3           1975         3           1975         3           1975         3</th> <th>2.3         2.9           2.3         2.9           2.3         2.9           2.3         7.2.9           2.8         7.2.8           2.8         7.2.8           2.1.1       
 7.2.9           2.3.3         7.2.9           7.3.3         7.2.9           7.2.8         7.2.8           7.3.9         7.2.0           7.2.0         7.2.8           7.0         7.3.4           7.2.7         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.1.8         7.5.7           7.4.8         7.5.7           7.4.8         7.5.7           7.4.1         7.5.2           7.4.8         7.5.7           7.4.8         7.5.7           7.4.8         7.5.4           7.5.7         7.4.4           7.5.7         7.4.4           7.5.7         7.4.4</th> <th>S</th> <th>peed<br/>MPH)</th> | 63.4         68.3           4         68.3           2         68.6           9         67.5           2         66.6           3         67.5           2         66.6           3         67.5           3         66.7           3         66.7           3         72.1           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           3         72.2           4         64.6           6         64.5           6         64.5           6         64.5           7         64.5           7         70.2           8         64.0           6         65.2 | 611<br>611<br>653<br>664<br>664<br>664<br>664<br>665<br>665<br>665<br>665<br>665<br>665                                                                                                   | 57.3<br>58.3<br>60.6<br>60.0<br>60.0<br>60.1<br>60.0<br>60.1<br>60.2<br>58.6<br>61.2<br>60.2<br>60.6<br>61.6<br>61.2<br>60.6<br>61.6<br>61.2<br>60.6<br>61.6<br>61.6<br>61.6<br>61.6<br>61.6<br>61.6<br>61.6 | 807<br>407<br>408<br>408<br>408<br>408<br>408<br>408<br>408<br>408                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Model         Table           Model         71.3           Model         71.3           Model         71.3           Model         71.3           Model         71.4           Model         71.4     <                                                                                                                                                                                                                                   | 710<br>710<br>710<br>710<br>710<br>710<br>710<br>710<br>710<br>710                         | 73.6<br>73.6<br>73.6<br>73.6<br>74.1<br>73.2<br>74.1<br>73.7<br>73.3<br>73.9<br>73.3<br>73.9<br>73.3<br>73.9<br>74.0<br>74.7<br>74.0<br>74.7<br>74.0<br>74.7<br>74.0<br>74.7<br>74.0<br>74.7<br>74.0<br>74.7<br>74.0<br>74.7<br>74.0<br>74.7<br>74.1<br>74.1<br>74.1<br>74.1<br>74.1<br>74.1<br>74.1 | 489<br>47.4<br>72.1<br>72.3<br>72.3<br>72.4<br>465.8<br>465.8<br>465.8<br>47.8<br>72.6<br>72.6<br>72.6<br>72.6<br>72.6<br>72.6<br>72.6<br>72.6                                                                                                       | 1248<br>1288<br>1445<br>1445<br>1445<br>1445<br>1445<br>1445<br>1445<br>14                                                           | 11.4<br>00.3<br>11.1<br>12.1<br>10.5<br>12.1<br>10.5<br>12.1<br>10.5<br>12.1<br>10.5<br>12.1<br>10.5<br>12.1<br>10.5<br>12.1<br>10.5<br>12.1<br>10.5<br>12.1<br>10.5<br>12.1<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>12.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5<br>10.5 | 714<br>716<br>718<br>718<br>718<br>718<br>718<br>718<br>718<br>718<br>718<br>718                                                  | 619<br>709<br>723<br>716<br>710<br>700<br>701<br>705<br>717<br>705<br>717<br>705<br>717<br>717<br>717<br>717<br>717<br>717<br>717<br>717<br>717<br>71                                                                                                                                                                                                                                                                                                                                                                                                                                                | 224<br>315<br>724<br>315<br>712<br>712<br>712<br>712<br>712<br>712<br>712<br>712<br>712<br>712       | 013<br>013<br>013<br>013<br>013<br>015<br>014<br>017<br>015<br>017<br>015<br>017<br>015<br>017<br>015<br>017<br>015<br>017<br>015<br>017<br>015<br>017<br>015<br>017<br>017<br>017<br>017<br>017<br>017<br>017<br>017 |
17.17<br>18.14<br>19.12<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19.18<br>19 | 005<br>644<br>645<br>646<br>645<br>645<br>645<br>645<br>64                                                                                                   | 713<br>7023<br>7024<br>7123<br>722<br>722<br>722<br>722<br>722<br>722<br>722<br>721<br>724<br>725<br>724<br>726<br>727<br>721<br>724<br>725<br>724<br>725<br>724<br>725<br>724<br>725<br>724<br>725<br>725<br>724<br>725<br>725<br>725<br>725<br>725<br>725<br>725<br>725<br>725<br>725 | 0070<br>0770<br>0770<br>0770<br>0770<br>0770<br>0770<br>077                                                                                                                                  | 1013         1           662         1           662         1           662         1           705         1           868         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705         1           705                    | 5.4         3.5         3.6           4.4         3.6         3.7           4.4         3.7         3.7           4.4         3.7         3.7           5.4         1.6         5.7           5.5         5.6         5.7           5.5         5.5         5.5           5.5         5.6         5.7           5.5         5.6         5.5           5.6         5.7         5.6           5.5         5.6         5.5           5.6         5.5         5.6           5.5         5.6         5.5           5.6         5.5         5.6           5.5         5.6         5.5           5.6         5.5         5.6           5.7         6.6         5.5           5.1         5.1         5.1           5.1         5.5         5.1           5.5         5.5         5.5 | 5 7.1.1<br>6 76.4<br>70.2<br>2 62.5<br>2 63.7<br>2 63.7<br>2 64.6<br>7 7.1.8<br>8 64.2<br>7 64.0<br>7 7 7.1.8<br>8 64.2<br>7 7 7 7.1.8<br>9 7.1.0<br>9 7.1.0                                                                                                                                                                                                                                    | 724<br>724<br>724<br>724<br>727<br>727<br>727<br>727<br>727<br>727                                       | 7248<br>7429<br>7157<br>711,<br>712,<br>713,<br>714,<br>714,<br>714,<br>714,<br>714,<br>714,<br>714,<br>714 | 12.14<br>71.71<br>71.71<br>70.80<br>71.71<br>71.74<br>71.72<br>72.72<br>71.71<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72<br>71.72 |
72,44<br>71,27<br>71,25<br>71,25<br>71,27<br>71,27<br>72,21<br>71,27<br>72,21<br>71,27<br>72,27<br>72,27<br>72,27<br>73,21<br>73,11<br>73,11<br>73,11<br>73,11<br>73,11<br>73,11<br>73,11<br>73,11<br>73,11<br>73,12<br>73,12<br>73,12<br>73,12<br>73,12<br>73,12<br>73,12<br>73,12<br>73,12<br>73,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12<br>74,12, | 213<br>735<br>735<br>735<br>735<br>737<br>737<br>856<br>80<br>80<br>754<br>735<br>736<br>80<br>754<br>735<br>736<br>80<br>754<br>735<br>735<br>736<br>80<br>754<br>735<br>735<br>735<br>736<br>80<br>754<br>735<br>735<br>735<br>735<br>735<br>735<br>735<br>735<br>735<br>735 | 006         1           006         1           008         0           008         0           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011         1           011 | 17         2           17         2           17         2           1811         1           1872         2           1872         2           1872         2           1872         2           1872         2           1872         2           1872         2           1872         2           1873         2           1874         2           1873         2           1873         2           1873         2           1873         2           1874         2           1875         2           1874         2           1874         2           1874         2           1874         2           1875         3           1884         3           1884         3           1884         3           1975         3           1975         3           1975         3           1975         3           1975         3           1975         3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 2.3         2.9           2.3         2.9           2.3         2.9           2.3         7.2.9           2.8         7.2.8           2.8         7.2.8           2.1.1         7.2.9           2.3.3         7.2.9           7.3.3         7.2.9           7.2.8         7.2.8           7.3.9         7.2.0           7.2.0         7.2.8           7.0         7.3.4           7.2.7         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.2.9         7.4.4           7.1.8         7.5.7           7.4.8         7.5.7           7.4.8         7.5.7           7.4.1         7.5.2           7.4.8         7.5.7           7.4.8         7.5.7           7.4.8         7.5.4           7.5.7         7.4.4           7.5.7         7.4.4           7.5.7         7.4.4                                                               | S | peed<br>MPH)                                                   |
| 6 00:00 PP<br>6 00:00 PP<br>6 30:00 PP<br>6 30:00 PP<br>6 32:00 PP<br>6 35:00 PP<br>7 05:00 PP<br>7 35:00 PP<br>7 35:0                                                                     | A80<br>359<br>7531<br>763<br>763<br>763<br>763<br>763<br>763<br>763<br>763<br>763<br>763                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 24.6<br>24.7<br>24.7<br>24.7<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>24.1<br>25.5<br>25.5<br>24.1<br>25.5<br>25.5<br>25.5<br>25.5<br>25.5<br>25.5<br>25.5<br>25 | 73.1<br>73.2<br>74.7<br>74.0<br>73.7<br>74.3<br>74.3<br>74.3<br>74.3<br>75.5<br>75.3<br>75.5<br>75.3<br>75.9<br>76.0<br>75.9<br>76.0<br>75.9<br>76.0<br>75.2<br>76.2<br>76.2<br>76.2<br>76.2<br>76.2<br>76.2<br>76.2<br>76 | 41.6<br>43.6<br>43.6<br>52.9<br>54.4<br>55.4<br>55.4<br>55.4<br>55.4<br>55.2<br>55.4<br>55.4<br>55.2<br>55.4<br>55.2<br>55.4<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.4<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2<br>55.2 | 61.0 S 61.7 S 64.3 6 6 64.3 6 6 64.3 6 6 65.4 S 66.5 S 59.7 S 64.6 S 59.7 S 64.6 S 66.1 6 6 61.6 6 6 6 6 6 7 72.9 6 772.9 6 772.9 6 772.9 6 772.9 71.5 6 77 71.5 7 71.5 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.6 7 773.0 8 7 773.6 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 8 7 773.0 | 8.1         85.3           8.1         55.8           8.1         55.8           8.4         64.8           8.0         66.8           8.1         66.8           8.1         66.8           8.1         66.8           8.1         66.8           8.1         66.8           8.1         66.8           8.1         66.8  
        8.2         90.3           8.4         66.8           8.5         71.6           8.5         71.6           8.5         71.6           8.5         71.4           8.5         71.4           8.5         71.4           8.5         71.4           8.5         71.4           8.6         71.4           8.7         71.4           8.8         71.4           8.9         71.4           8.1         71.4           8.2         71.4           8.3         71.4           8.4         71.4           8.5         71.4           8.5         71.4           8.6         71.4 <td>1         66.1           66.9         68.4           0         68.4           0         64.6           0         64.6           0         64.7           0         64.8           0         64.7           0         64.8           0         64.8           0         70.0           0         71.0           0         74.0           0         74.0           0         74.0           0         74.0           0         74.0           1         71.0           4         71.7           7         73.4           7         73.4           7         73.2           6         71.8           7         73.3</td> <td>654<br/>645<br/>645<br/>608<br/>672<br/>635<br/>653<br/>671<br/>668<br/>702<br/>668<br/>702<br/>668<br/>702<br/>668<br/>702<br/>668<br/>702<br/>671<br/>708<br/>716<br/>727<br/>708<br/>749<br/>738<br/>754<br/>754</td> <td>62.4<br/>61.2<br/>64.0<br/>65.3<br/>62.4<br/>60.5<br/>61.8<br/>62.9<br/>61.7<br/>66.0<br/>65.4<br/>67.7<br/>67.3<br/>68.4<br/>65.7<br/>72.6<br/>73.6<br/>72.8<br/>70.6<br/>72.8<br/>70.6</td> <td>701 1<br/>401 39<br/>71 5<br/>72 2<br/>71 0<br/>71 8<br/>70 0<br/>71 8<br/>70 0<br/>72 2<br/>70 0<br/>71 8<br/>70 0<br/>72 2<br/>70 0<br/>71 8<br/>70 0<br/>70 0<br/>70</td> <td>71.4         71.2         74.3           71.2         74.3         71.7           71.2         71.3         71.7           71.2         71.3         71.7           71.2         71.3         71.3           70.8         72.2         71.4           70.9         71.3         71.4           71.3         71.4         71.3           71.3         71.4         71.4           72.5         74.4         76.0           72.5         74.4         76.0           72.5         74.4         76.0           72.5         74.4         76.0           71.3         71.4         76.0           71.4         74.0         76.0           71.6         75.0         71.4           71.6         75.0         71.4           71.6         75.0         71.4           71.4         76.0         75.0           71.4         76.0         75.0           71.4         76.0         76.0           71.4         76.0         76.0           71.4         76.0         76.0           71.4         76.0         76.0</td> <td>74.6<br/>73.7<br/>73.6<br/>73.7<br/>74.9<br/>74.9<br/>74.9<br/>74.9<br/>74.9<br/>74.9<br/>74.9<br/>74</td> <td>24.4<br/>74.2<br/>71.5<br/>73.6<br/>73.8<br/>73.8<br/>73.8<br/>74.2<br/>73.8<br/>73.8<br/>74.2<br/>75.1<br/>74.4<br/>75.5<br/>75.4<br/>75.4<br/>75.4<br/>75.5<br/>75.4<br/>75.5<br/>75.4<br/>75.5<br/>75.5</td> <td>20.9<br/>72.8<br/>71.1<br/>72.4<br/>72.6<br/>71.8<br/>72.9<br/>72.3<br/>73.6<br/>73.6<br/>73.6<br/>73.6<br/>73.6<br/>73.7<br/>73.5<br/>73.7<br/>73.5<br/>73.7<br/>73.5<br/>73.7<br/>73.5<br/>73.7<br/>73.5<br/>73.4<br/>73.4<br/>73.4<br/>73.4<br/>73.4<br/>73.4<br/>73.7<br/>73.7</td> <td>74.4<br/>74.5<br/>74.6<br/>75.0<br/>74.6<br/>75.0<br/>74.6<br/>75.0<br/>75.4<br/>75.5<br/>75.6<br/>75.6<br/>75.6<br/>75.6<br/>75.6<br/>75.6<br/>75.6</td> <td>71.2<br/>72.4<br/>72.4<br/>72.8<br/>72.7<br/>71.0<br/>71.8<br/>71.0<br/>71.8<br/>71.0<br/>71.3<br/>71.4<br/>71.0<br/>71.3<br/>71.4<br/>71.0<br/>71.3<br/>71.4<br/>72.4<br/>72.4<br/>72.4<br/>73.4<br/>72.5<br/>73.4<br/>72.5<br/>73.4<br/>72.5<br/>73.4<br/>73.0<br/>73.4<br/>73.0<br/>73.5<br/>73.0<br/>73.5<br/>73.0<br/>73.5<br/>73.0<br/>73.5<br/>73.0<br/>73.5<br/>73.0<br/>73.5<br/>73.0<br/>73.5<br/>73.0<br/>73.5<br/>73.5<br/>73.5<br/>73.5<br/>73.5<br/>73.5<br/>73.5<br/>73.5</td> <td>24.3 点 28.2 年 24.4 年 77.2 24.0 年 24.5 年 77.5 24.5 24.4 年 77.7 24.8 24.8 25.5 年 75.5 1.4 月 75.7 25.7 25.7 25.7 25.7 25.7 25.7 25.7</td> <td>21.3<br/>60.8<br/>71.6<br/>71.7<br/>71.3<br/>72.0<br/>72.8<br/>72.7<br/>72.0<br/>72.8<br/>72.7<br/>72.8<br/>72.7<br/>72.8<br/>72.7<br/>72.8<br/>72.7<br/>72.8<br/>72.7<br/>72.8<br/>72.7<br/>72.8<br/>72.7<br/>72.8<br/>72.7<br/>72.8<br/>72.7<br/>72.9<br/>72.8<br/>72.7<br/>72.0<br/>72.8<br/>72.7<br/>72.0<br/>72.8<br/>72.7<br/>72.9<br/>72.8<br/>72.7<br/>72.9<br/>72.8<br/>72.7<br/>72.9<br/>72.8<br/>72.7<br/>72.9<br/>72.8<br/>72.7<br/>72.9<br/>72.8<br/>72.7<br/>72.9<br/>72.8<br/>72.7<br/>72.9<br/>72.8<br/>72.8<br/>72.7<br/>72.9<br/>72.8<br/>72.8<br/>72.7<br/>72.9<br/>72.8<br/>72.8<br/>72.7<br/>72.8<br/>72.8<br/>72.8<br/>72.8<br/>72.7<br/>72.8<br/>72.8</td> <td>71.5<br/>71.3<br/>71.2<br/>71.4<br/>71.5<br/>71.3<br/>71.3<br/>71.3<br/>71.3<br/>71.3<br/>71.3<br/>71.3<br/>71.3</td> <td>66.5<br/>69.2<br/>66.9<br/>64.9<br/>67.0<br/>69.1<br/>69.1<br/>69.1<br/>69.1<br/>69.1<br/>69.1<br/>69.1<br/>69.1</td> <td>71.8<br/>73.9<br/>73.9<br/>73.9<br/>73.9<br/>73.9<br/>73.9<br/>73.9<br/>73.9</td> <td>66.0<br/>64.9<br/>64.9<br/>64.9<br/>64.9<br/>64.9<br/>64.9<br/>64.1<br/>70.0<br/>60.1<br/>71.2<br/>70.5<br/>70.0<br/>71.0<br/>71.0<br/>71.0<br/>71.0<br/>71.0<br/>71.0<br/>71.0</td> <td>72.3<br/>72.8<br/>72.8<br/>72.8<br/>72.8<br/>72.8<br/>72.8<br/>72.8<br/>72.8</td> <td>68.3<br/>67.4<br/>67.0<br/>67.2<br/>68.3<br/>68.4<br/>70.3<br/>70.1<br/>68.0<br/>71.0<br/>70.3<br/>72.1<br/>68.1<br/>72.5<br/>73.7<br/>72.7<br/>72.9<br/>71.7<br/>72.9<br/>71.7<br/>72.8<br/>72.9<br/>73.0<br/>73.0</td> <td>71.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7</td> <td>5.6 4445 4457 44457 470 470 470 470 470 470 470 470 470 47</td> <td>2 70.4<br/>6 72.3<br/>73.8<br/>73.8<br/>73.8<br/>73.8<br/>72.8<br/>72.8<br/>72.8<br/>72.8<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>72.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75.5<br/>75</td> <td>201<br/>642<br/>71.3<br/>71.2<br/>70.7<br/>70.7<br/>71.3<br/>71.3<br/>71.3<br/>71.3<br/>71.3<br/>71.3<br/>71.3<br/>71</td> <td>14.8<br/>17.5<br/>17.5<br/>17.5<br/>17.5<br/>17.5<br/>17.5<br/>17.5<br/>17.5</td>
<td>73,7<br/>73,8<br/>73,8<br/>73,8<br/>73,8<br/>73,7<br/>73,2<br/>73,4<br/>74,6<br/>74,6<br/>74,7<br/>74,6<br/>75,7<br/>75,7<br/>75,7<br/>75,7<br/>75,7<br/>75,7<br/>75,7<br/>75</td> <td>73.5<br/>73.3<br/>74.4<br/>73.4<br/>73.5<br/>73.5<br/>73.5<br/>73.5<br/>73.5<br/>73.5<br/>73.5<br/>73.5</td> <td>76.4<br/>72.4<br/>72.4<br/>73.8<br/>72.7<br/>75.2<br/>76.3<br/>76.3<br/>76.2<br/>77.2<br/>76.4<br/>77.3<br/>77.4<br/>77.3<br/>77.4<br/>77.3<br/>77.3<br/>77.3<br/>77</td> <td>13.00 2 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7</td> <td>72,7 1<br/>73,8 7<br/>73,8 7<br/>73,9 7<br/>74,9 7<br/>74,9 7<br/>75,8 7<br/>75,9 75,9 75,9 75,9 75,9 75,9 75,9 75,9</td> <td>2.4         75.4           14.1         75.3           14.1         75.3           15.3         75.3           15.4         75.3           15.7         75.2           15.8         74.3           15.8         74.3           15.8         74.3           15.8         74.3           15.8         74.3           15.8         74.3           15.8         74.3           15.8         74.3           15.9         74.3           15.9         74.3           15.0         74.9           15.4         74.6           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5<td></td><td>20<br/>25<br/>30<br/>35<br/>40<br/>45<br/>50<br/>55<br/>60<br/>65<br/>70</td></td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 1         66.1           66.9         68.4           0         68.4           0         64.6         
 0         64.6           0         64.7           0         64.8           0         64.7           0         64.8           0         64.8           0         70.0           0         71.0           0         74.0           0         74.0           0         74.0           0         74.0           0         74.0           1         71.0           4         71.7           7         73.4           7         73.4           7         73.2           6         71.8           7         73.3                                                                                                                                                                                                                                                           | 654<br>645<br>645<br>608<br>672<br>635<br>653<br>671<br>668<br>702<br>668<br>702<br>668<br>702<br>668<br>702<br>668<br>702<br>671<br>708<br>716<br>727<br>708<br>749<br>738<br>754<br>754 | 62.4<br>61.2<br>64.0<br>65.3<br>62.4<br>60.5<br>61.8<br>62.9<br>61.7<br>66.0<br>65.4<br>67.7<br>67.3<br>68.4<br>65.7<br>72.6<br>73.6<br>72.8<br>70.6<br>72.8<br>70.6                                         | 701 1<br>401 39<br>71 5<br>72 2<br>71 0<br>71 8<br>70 0<br>71 8<br>70 0<br>72 2<br>70 0<br>71 8<br>70 0<br>72 2<br>70 0<br>71 8<br>70 0<br>70 | 71.4         71.2         74.3           71.2         74.3         71.7           71.2         71.3         71.7           71.2         71.3         71.7           71.2         71.3         71.3           70.8         72.2         71.4           70.9         71.3         71.4           71.3         71.4         71.3           71.3         71.4         71.4           72.5         74.4         76.0           72.5         74.4         76.0           72.5         74.4         76.0           72.5         74.4         76.0           71.3         71.4         76.0           71.4         74.0         76.0           71.6         75.0         71.4           71.6         75.0         71.4           71.6         75.0         71.4           71.4         76.0         75.0           71.4         76.0         75.0           71.4         76.0         76.0           71.4         76.0         76.0           71.4         76.0         76.0           71.4         76.0         76.0                                                                                                                                                                                  | 74.6<br>73.7<br>73.6<br>73.7<br>74.9<br>74.9<br>74.9<br>74.9<br>74.9<br>74.9<br>74.9<br>74 | 24.4<br>74.2<br>71.5<br>73.6<br>73.8<br>73.8<br>73.8<br>74.2<br>73.8<br>73.8<br>74.2<br>75.1<br>74.4<br>75.5<br>75.4<br>75.4<br>75.4<br>75.5<br>75.4<br>75.5<br>75.4<br>75.5<br>75.5                                                                                                                 | 20.9<br>72.8<br>71.1<br>72.4<br>72.6<br>71.8<br>72.9<br>72.3<br>73.6<br>73.6<br>73.6<br>73.6<br>73.6<br>73.7<br>73.5<br>73.7<br>73.5<br>73.7<br>73.5<br>73.7<br>73.5<br>73.7<br>73.5<br>73.4<br>73.4<br>73.4<br>73.4<br>73.4<br>73.4<br>73.7<br>73.7 | 74.4<br>74.5<br>74.6<br>75.0<br>74.6<br>75.0<br>74.6<br>75.0<br>75.4<br>75.5<br>75.6<br>75.6<br>75.6<br>75.6<br>75.6<br>75.6<br>75.6 | 71.2<br>72.4<br>72.4<br>72.8<br>72.7<br>71.0<br>71.8<br>71.0<br>71.8<br>71.0<br>71.3<br>71.4<br>71.0<br>71.3<br>71.4<br>71.0<br>71.3<br>71.4<br>72.4<br>72.4<br>72.4<br>73.4<br>72.5<br>73.4<br>72.5<br>73.4<br>72.5<br>73.4<br>73.0<br>73.4<br>73.0<br>73.5<br>73.0<br>73.5<br>73.0<br>73.5<br>73.0<br>73.5<br>73.0<br>73.5<br>73.0<br>73.5<br>73.0<br>73.5<br>73.0<br>73.5<br>73.5<br>73.5<br>73.5<br>73.5<br>73.5<br>73.5<br>73.5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 24.3 点 28.2 年 24.4 年 77.2 24.0 年 24.5 年 77.5 24.5 24.4 年 77.7 24.8 24.8 25.5 年 75.5 1.4 月 75.7 25.7 25.7 25.7 25.7 25.7 25.7 25.7 | 21.3<br>60.8<br>71.6<br>71.7<br>71.3<br>72.0<br>72.8<br>72.7<br>72.0<br>72.8<br>72.7<br>72.8<br>72.7<br>72.8<br>72.7<br>72.8<br>72.7<br>72.8<br>72.7<br>72.8<br>72.7<br>72.8<br>72.7<br>72.8<br>72.7<br>72.8<br>72.7<br>72.9<br>72.8<br>72.7<br>72.0<br>72.8<br>72.7<br>72.0<br>72.8<br>72.7<br>72.9<br>72.8<br>72.7<br>72.9<br>72.8<br>72.7<br>72.9<br>72.8<br>72.7<br>72.9<br>72.8<br>72.7<br>72.9<br>72.8<br>72.7<br>72.9<br>72.8<br>72.7<br>72.9<br>72.8<br>72.8<br>72.7<br>72.9<br>72.8<br>72.8<br>72.7<br>72.9<br>72.8<br>72.8<br>72.7<br>72.8<br>72.8<br>72.8<br>72.8<br>72.7<br>72.8<br>72.8 | 71.5<br>71.3<br>71.2<br>71.4<br>71.5<br>71.3<br>71.3<br>71.3<br>71.3<br>71.3<br>71.3<br>71.3<br>71.3 | 66.5<br>69.2<br>66.9<br>64.9<br>67.0<br>69.1<br>69.1<br>69.1<br>69.1<br>69.1<br>69.1<br>69.1<br>69.1                                                                                                                  | 71.8<br>73.9<br>73.9<br>73.9<br>73.9<br>73.9<br>73.9<br>73.9<br>73.9                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 66.0<br>64.9<br>64.9<br>64.9<br>64.9<br>64.9<br>64.9<br>64.1<br>70.0<br>60.1<br>71.2<br>70.5<br>70.0<br>71.0<br>71.0<br>71.0<br>71.0<br>71.0<br>71.0<br>71.0 | 72.3<br>72.8<br>72.8<br>72.8<br>72.8<br>72.8<br>72.8<br>72.8<br>72.8                                                                                                                                                                             
                                      | 68.3<br>67.4<br>67.0<br>67.2<br>68.3<br>68.4<br>70.3<br>70.1<br>68.0<br>71.0<br>70.3<br>72.1<br>68.1<br>72.5<br>73.7<br>72.7<br>72.9<br>71.7<br>72.9<br>71.7<br>72.8<br>72.9<br>73.0<br>73.0 | 71.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.7         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7           70.8         7 | 5.6 4445 4457 44457 470 470 470 470 470 470 470 470 470 47                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 2 70.4<br>6 72.3<br>73.8<br>73.8<br>73.8<br>73.8<br>72.8<br>72.8<br>72.8<br>72.8<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>72.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75.5<br>75 | 201<br>642<br>71.3<br>71.2<br>70.7<br>70.7<br>71.3<br>71.3<br>71.3<br>71.3<br>71.3<br>71.3<br>71.3<br>71 | 14.8<br>17.5<br>17.5<br>17.5<br>17.5<br>17.5<br>17.5<br>17.5<br>17.5                                        | 73,7<br>73,8<br>73,8<br>73,8<br>73,8<br>73,7<br>73,2<br>73,4<br>74,6<br>74,6<br>74,7<br>74,6<br>75,7<br>75,7<br>75,7<br>75,7<br>75,7<br>75,7<br>75,7<br>75                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 73.5<br>73.3<br>74.4<br>73.4<br>73.5<br>73.5<br>73.5<br>73.5<br>73.5<br>73.5<br>73.5<br>73.5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    
                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 76.4<br>72.4<br>72.4<br>73.8<br>72.7<br>75.2<br>76.3<br>76.3<br>76.2<br>77.2<br>76.4<br>77.3<br>77.4<br>77.3<br>77.4<br>77.3<br>77.3<br>77.3<br>77                                                                                                                             | 13.00 2 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 72,7 1<br>73,8 7<br>73,8 7<br>73,9 7<br>74,9 7<br>74,9 7<br>75,8 7<br>75,9 75,9 75,9 75,9 75,9 75,9 75,9 75,9 | 2.4         75.4           14.1         75.3           14.1         75.3           15.3         75.3           15.4         75.3           15.7         75.2           15.8         74.3           15.8         74.3           15.8         74.3           15.8         74.3           15.8         74.3           15.8         74.3           15.8         74.3           15.8         74.3           15.9         74.3           15.9         74.3           15.0         74.9           15.4         74.6           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5         75.3           15.5 <td></td> <td>20<br/>25<br/>30<br/>35<br/>40<br/>45<br/>50<br/>55<br/>60<br/>65<br/>70</td> |   | 20<br>25<br>30<br>35<br>40<br>45<br>50<br>55<br>60<br>65<br>70 |

# Figure 17. Heatmap. Southbound I-15 PM observed (top) versus modeled (bottom) speed contours – incident condition.

(Source: San Diego I-15—Post-Deployment Analysis, Modeling and Simulation (AMS) Reasonableness Assessment and Tool Modification Technical Memorandum—Final, Cambridge Systematics, Texas A&M Transportation Institute, and Southern Methodist University, December 2015, p. 18, unpublished.)

3000 FM         400         400           3000 FM         464         400           400 GM         464         400           400 GM         464         400           500 GM         464         400           500 GM         400         400			No.         No. <th>B         C         C         C         C         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S</th> <th>100 <math>100</math> <math>1000</math> <math>100</math> <math>100</math>         &lt;</th> <th></th> <th>化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化</th>	B         C         C         C         C         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S         S	100 $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $100$ $1000$ $100$ $100$ <		化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化化
B) M         B) M <th< td=""><td>65 年 46 年 46 年 46 年 47 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8</td><td>66         47         47           44         44         27           45         46         27           45         53         23           54         54         27           44         42         27           44         43         27           44         43         28           53         45         32           54         53         28           44         21         34           58         47         28           58         47         28           58         47         28           53         47         28           54         52         28           64         53         47           53         47         28           54         47         28           53         47         28           54         47         28           54         47         28           53         47         28           54         52         28           64         53         53           64         53         53</td><td>bit         7         7         7         7           44         70         74         71         74           30         44         54         71         74           30         44         54         71         74           30         44         54         71         74           30         44         54         71         84           56         44         57         72         72           57         66         64         72         72           57         66         64         72         72           57         66         64         72         72           57         66         64         72         72           57         66         64         72         72           57         66         64         72         72           57         66         64         72         72           58         67         73         73         75           59         66         67         72         72           50         68         72         72           59         67&lt;</td><td>70         80         35         40         55           70         80         35         40         55           71         60         37         34         35           71         30         32         34         35           71         33         36         37         34         35           71         33         36         37         34         36           71         32         34         38         32         34           71         32         34         38         32         34           71         32         34         38         32         34           71         32         34         38         32         35           71         35         45         32         36         32           70         36         43         38         32         36           71         35         45         32         36         32           70         36         43         38         32         36           71         35         45         32         35         32           70         <t< td=""><td>12         10         17         64         61         6           12         13         14         16         16         6           13         13         14         16         16         6         16         1           13         14         16         10         15         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1</td><td>30         60         33         75           54         40         72         74           6         54         40         72         74           6         32         24         72         74           6         32         24         72         74           7         32         24         72         74           6         32         24         72         73           7         35         27         72         73           6         40         72         73         74           6         40         72         73         74           6         40         72         73         74           6         40         72         73         74           6         54         44         72         73           6         54         44         72         73           6         54         44         72         73           7         54         44         72         73           7         54         44         72         73           7         54         51</td><td>В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В</td></t<></td></th<>	65 年 46 年 46 年 46 年 47 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	66         47         47           44         44         27           45         46         27           45         53         23           54         54         27           44         42         27           44         43         27           44         43         28           53         45         32           54         53         28           44         21         34           58         47         28           58         47         28           58         47         28           53         47         28           54         52         28           64         53         47           53         47         28           54         47         28           53         47         28           54         47         28           54         47         28           53         47         28           54         52         28           64         53         53           64         53         53	bit         7         7         7         7           44         70         74         71         74           30         44         54         71         74           30         44         54         71         74           30         44         54         71         74           30         44         54         71         84           56         44         57         72         72           57         66         64         72         72           57         66         64         72         72           57         66         64         72         72           57         66         64         72         72           57         66         64         72         72           57         66         64         72         72           57         66         64         72         72           58         67         73         73         75           59         66         67         72         72           50         68         72         72           59         67<	70         80         35         40         55           70         80         35         40         55           71         60         37         34         35           71         30         32         34         35           71         33         36         37         34         35           71         33         36         37         34         36           71         32         34         38         32         34           71         32         34         38         32         34           71         32         34         38         32         34           71         32         34         38         32         35           71         35         45         32         36         32           70         36         43         38         32         36           71         35         45         32         36         32           70         36         43         38         32         36           71         35         45         32         35         32           70 <t< td=""><td>12         10         17         64         61         6           12         13         14         16         16         6           13         13         14         16         16         6         16         1           13         14         16         10         15         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1</td><td>30         60         33         75           54         40         72         74           6         54         40         72         74           6         32         24         72         74           6         32         24         72         74           7         32         24         72         74           6         32         24         72         73           7         35         27         72         73           6         40         72         73         74           6         40         72         73         74           6         40         72         73         74           6         40         72         73         74           6         54         44         72         73           6         54         44         72         73           6         54         44         72         73           7         54         44         72         73           7         54         44         72         73           7         54         51</td><td>В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В</td></t<>	12         10         17         64         61         6           12         13         14         16         16         6           13         13         14         16         16         6         16         1           13         14         16         10         15         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1	30         60         33         75           54         40         72         74           6         54         40         72         74           6         32         24         72         74           6         32         24         72         74           7         32         24         72         74           6         32         24         72         73           7         35         27         72         73           6         40         72         73         74           6         40         72         73         74           6         40         72         73         74           6         40         72         73         74           6         54         44         72         73           6         54         44         72         73           6         54         44         72         73           7         54         44         72         73           7         54         44         72         73           7         54         51	В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В           В         В

# Figure 18. Heatmap. Northbound I-15 PM observed (top) versus modeled (bottom) speed contours – incident condition.

(Source: San Diego I-15—Post-Deployment Analysis, Modeling and Simulation (AMS) Reasonableness Assessment and Tool Modification Technical Memorandum—Final, Cambridge Systematics, Texas A&M Transportation Institute, and Southern Methodist University, December 2015, p. 19, unpublished.)

#### **Overall Conclusion**

Verifying that the model accurately represents the current traffic conditions in the field is an important component of the Reasonableness Assessment. This effort helps to ensure that the post-ICM deployment baseline model is capable of accurately representing road geometries, demands, and post-deployment operational conditions in 2014 and 2015. The changes made and the lessons learned through this assessment contribute to the continuous improvement of the AMS approach throughout the various stages of the ICM Initiative.

New and more current field data were collected and several adjustments to the model were completed in order to improve the baseline model. The presence of additional information therefore allowed for a more accurate observed dataset to be compared to the model outputs. For the evaluation of the incident case, the model was fine-tuned to represent the incident by correctly adjusting the number of lanes associated with the closure, the passerby speed at the location of the incident, the incident duration, and the vehicle reactions approaching the incident. The model is a fair representation of the incident.

For a typical day with no incident, the overall comparison of total model link flows against the aggregate field volumes showed that the model met the suggested link count calibration criteria. Plus, the overall results of the speed contour comparisons showed that the model is able to sufficiently represent the bottleneck temporal and spatial extents for both southbound and northbound I-15. For an incident day the model is also able to sufficiently represent the bottleneck temporal an incident. This is crucial for representing different non-recurring operational conditions that the analysis was based on which increases the accuracy of the estimated corridor benefits.

Therefore, the model was capable of adequately representing the post-deployment corridor operational conditions in the I-15 Corridor.

Further calibration was conducted on the individual clusters used in alternatives analysis, which is described in greater detail in the Analysis Scenarios section of Chapter 8.

# Chapter 8. Alternatives Analysis and Results

This section first provides an overview of the analysis, modeling and simulation (AMS) efforts associated with the Post-Deployment Alternatives Analysis. The alternatives analysis serves to assess the performance of various components of the Integrated Corridor Management (ICM) system under different operational conditions (e.g., time of day, direction of traffic, duration until the incident was cleared, etc.). The methodologies, tools, and strategies incorporated into the Post-Deployment Alternative scenarios identified for analysis. This AMS work followed the United States Department of Transportation (US DOT)-approved Post-Deployment AMS Plan.

Several facets of analysis results are presented. The pulse survey results from the Volpe Center reveals the percentage of travelers along the I-15 Corridor who made travel changes based on pre-trip or en-route traveler information. Main findings from mobility, reliability, and variability performance measures are highlighted and compared against estimated travel time benefits from pre-deployment AMS of the "as-planned" ICM system. Lastly, the ICM AMS methodology is reviewed as a whole, with several takeaways summarized for future ICM adopters.

# **Analysis Scenarios**

Once the models were refined using the enhancements presented in the previous chapter, the models were used for additional testing and analysis that served to assess the impacts of the implemented ICM deployment. The potential ICM deployment-related alternatives were identified using cluster analysis that grouped together incidents that occurred under operational conditions (e.g., time of day, direction of traffic, length of time until the incident was cleared, etc.) which were more similar to each other, than to those in other groups (clusters). These clusters were then prioritized based on total delay impact. Field observed incidents that occurred in the year after ICM deployment were matched to high-impact clusters sharing similar operational conditions. Feedback and input from the site coordinators and local agencies were used to select the final eight scenarios included in alternatives analysis, along with an additional hypothetical scenario. (Please note, the contents of this chapter have been based on a set of memos from the Evaluation Contractor to FHWA titled "ICM Evaluation – San Diego Site Cluster Analysis – Daily Incident Probability", dated April 14, 2016 and "ICM Evaluation – San Diego Incident Matching", also dated April 14, 2016.)

The AMS team focused on identifying and then representing the "as deployed" system. This includes linking the assumptions in Chapter 5 about how the "with" and "without" cases are differentiated and modeled with the cluster analysis. Alternatives analysis were performed primarily by Transport Simulation Systems (TSS), contractor to the San Diego ICM Demonstration Site (San Diego Association of Governments) in conjunction with Cambridge Systematics, the AMS contractor for the US DOT.

# Analysis Timeframe

Although the ICM deployment on I-15 became operational in March 2013, a major update took place at the beginning of September 2013, and incident/event response capabilities came online in 2015. Wayfinder signs for route guidance on the arterials became active in 2015 – these signs were expected to facilitate routing through alternate arterial routes in response to incident and events. Existing bus rapid transit (BRT) was operating during the peak periods only in 2013; in the summer of 2014 off-peak BRT became operational all day in both directions.

The period after the deployment and testing of Wayfinder signs and additional BRT (mid-2014 to mid-2015) was determined to be the best option for conducting AMS to represent "with ICM" because: 1) the basic ICM system was operational for several months and the system was deemed stable due to early bug fixes; and 2) Wayfinder signs were deployed and alternate routing was in place. The analysis timeframe was consistent with the cluster-representative days shown in Table 8.

# **Cluster Analysis**

In order to group events with similar traffic characteristics during the pre-deployment period along the I-15 corridor, the Evaluation Team conducted a statistical cluster analysis using event occurrences (date and duration), traffic flow rates, average speeds, and travel time data. The San Diego ICM decision support system (DSS) generates response plans based on congestion, and all DSS activations during the evaluation period happened to be incident-related. As such, the objective of the cluster analysis, was to identify the event clusters that fit a certain criteria such as delay impact and occurrence percentage as percent of time period with incidents and percent of total analysis time period. Identified clusters with the most occurrence and impact were used in finding incidents in the post-deployment period that are a close match to those pre-deployment incident types. This was accomplished by measuring the percentage of occurrence of clusters during the pre-deployment period and seeking matches that reflect similar, or ideally, identical incidents in the post-deployment period. The matched incidents and/or events were then used as input parameters in the AMS process.

The archived pre-deployment data were collected between March 1, 2012 and February 28, 2013, and the data sets consisted of the key variables such as the location and type of an event, duration of incidents in minutes, and directional traffic volumes and speeds recorded by the detectors along I-15 in one-hour intervals. The data sets excluded weekend days and Federally-observed holidays; no days of planned special events that would affect traffic conditions significantly were identified for removal. The time periods included in the analyses were AM peak period (6:00 AM – 9:59 AM), Midday (MD) period (10:00 AM – 2:59 PM), and PM peak period (3:00 PM – 6:59 PM). The overnight period from 7:00 PM to 5:59 AM was not of interest for the analyses.

Based on expected impact magnitude, proposed clusters of operational conditions were identified using the following temporal and/or directional variables:

- *Traffic Flow Rate* (vehicles per hour) The temporal and directional traffic flow rate measured in one-hour intervals at the stations along the I-15 corridor and within the study limits.
- Day of the Incident The day on which the incident occurred.

- Day-of-Week The day of the week on which the incident occurred.
- *Time of the Incident* The time-of-day on which the incident occurred.
- Direction of Traffic The direction of lanes on which the incident occurred.
- Duration of Incidents (minutes) The reported number of minutes elapsed from the initial occurrence of the traffic incident to the earliest of the time it was closed or that the scene was cleared according to the data.
- *Travel Time* (minutes) The calculated temporal average directional travel time along the I-15 corridor.

Data sources for the variables listed above include the Regional Integrated Transportation Information System (RITIS), which is maintained by the University of Maryland's Center for Advanced Transportation Technology Laboratory (CATT Lab).

Table 6 presents a summary of identified clusters for the I-15 corridor. After the tabulation of the cluster analysis results, two measures were added to the tables to gain a better understanding of cluster comparisons. These two measures are post-processed values and were not part of the clustering process. The first measure, *single incident delay impact*, is the difference between the average travel time calculated within a particular cluster and the lowest observed (measured) corridor-wide directional travel time regardless of the time period. The observed lowest travel time was recorded as 13.58 minutes for the northbound (December 25, 2012 during the midday peak period) and 14.00 minutes for the southbound direction (November 22, 2012 during the AM peak period). The second measure, *total cluster delay impact* (for each cluster), was calculated as the product of the single incident delay impact and number of days in the cluster.

The "Percent of Analysis Time Period" column shows the percent of days in a year that are represented in each cluster for each direction of travel and directional AM/MD/PM peak period. For the southbound morning peak direction-period, no incidents occur approximately 12.4 percent of the time and for the northbound afternoon peak direction-period, no incidents occur approximately 9.6 percent of the time; the percentages given in this column are by direction-period. For example cluster SB AM 2 represents an occurrence of 37.5 percent for the AM peak period and in the southbound direction only.

Cluster	Duration (min)	Volume (vehicle per hour [vph])	Travel Time (min)	Single Incident Delay Impact (min)	Incidents Per Period	Days in Cluster	Total Cluster Delay Impact (min)	Percent of Time Period with Incidents	Percent of Total Analysis Time Period	Total Days with Incidents in Analysis Period	Total Days in Analysis
SB AM 2	42.89	6,348	16.77	2.77	3.7	39	108.03	42.9%	37.5%	91	104
NB PM 3	46.18	9,034	16.35	2.77	5.5	36	99.72	48.6%	34.6%	94	104
NB MID 4	37.31	7,079	15.54	1.96	2.1	42	82.32	51.2%	40.4%	82	104

# Table 6. Summary of all clusters for all time periods and both directions, ordered by total cluster delay impact.

Cluster	Duration (min)	Volume (vehicle per hour [vph])	Travel Time (min)	Single Incident Delay Impact (min)	Incidents Per Period	Days in Cluster	Total Cluster Delay Impact (min)	Percent of Time Period with Incidents	Percent of Total Analysis Time Period	Total Days with Incidents in Analysis Period	Total Days in Analysis
NB PM 4	44.46	8,870	16.11	2.53	2.1	25	63.25	33.8%	24.0%	94	104
SB AM 1	32.64	6,201	15.72	1.72	1.9	29	49.88	31.9%	27.9%	91	104
NB PM 1	35.00	6,416	16.04	2.46	2.5	17	41.82	23.0%	16.3%	94	104
SB PM 1	36.64	4,773	14.95	0.95	2.6	43	40.85	52.4%	41.3%	86	104
SB MID 3	35.46	4,456	15.19	1.19	2.2	33	39.27	36.3%	31.7%	91	104
NB AM 3	29.81	6,721	14.99	1.41	1.9	27	38.07	28.7%	26.0%	74	104
SB AM 3	41.20	6,038	18.33	4.33	5.9	8	34.64	8.8%	7.7%	91	104
SB MID 5	27.38	4,462	15.76	1.76	2.4	19	33.44	20.9%	18.3%	91	104
NB MID 3	32.29	5,992	16.29	2.71	1.6	12	32.52	14.6%	11.5%	82	104
NB AM 1	34.17	4,767	15.02	1.44	1.7	19	27.36	20.2%	18.3%	74	104
SB AM 9	50.40	5,658	21.81	7.81	9.7	3	23.43	3.3%	2.9%	91	104
NB PM 2	32.53	6,955	16.5	2.92	8.8	8	23.36	10.8%	7.7%	94	104
NB AM 4	23.55	5,657	15.94	2.36	3.6	9	21.24	9.6%	8.7%	74	104
SB PM 3	29.97	5,011	15.17	1.17	3.1	18	21.06	22.0%	17.3%	86	104
NB PM 5	34.71	6,836	19.83	6.25	4.7	3	18.75	4.1%	2.9%	94	104
SB MID 1	32.88	4,847	15.41	1.41	3.9	13	18.33	14.3%	12.5%	91	104
SB AM 4	46.01	6,154	16.91	2.91	10.6	5	14.55	5.5%	4.8%	91	104
NB AM 2	35.13	6,937	15.09	1.51	6.0	9	13.59	9.6%	8.7%	74	104
NB MID 6	30.90	5,903	16.24	2.66	7.4	5	13.30	6.1%	4.8%	82	104
NB MID 2	38.14	5,148	15.77	2.19	4.5	6	13.14	7.3%	5.8%	82	104
SB MID 6	51.14	4,354	16.06	2.06	6.2	6	12.36	6.6%	5.8%	91	104
SB PM 2	26.60	5,409	15.08	1.08	3.2	11	11.88	13.4%	10.6%	86	104
NB PM 8	51.29	6,178	24.05	10.47	7.0	1	10.47	1.4%	1.0%	94	104
NB AM 5	68.75	8,146	15.52	1.94	2.0	5	9.70	5.3%	4.8%	74	104
NB MID 7	30.85	7,565	15.5	1.92	5.6	5	9.60	6.1%	4.8%	82	104
NB MID 1	48.53	4,753	15.49	1.91	2.0	5	9.55	6.1%	4.8%	82	104
SB MID 2	36.86	4,177	15.57	1.57	1.8	6	9.42	6.6%	5.8%	91	104
NB PM 6	38.05	9,156	18.02	4.44	13.0	2	8.88	2.7%	1.9%	94	104
SB PM 5	61.17	4,711	16.17	2.17	2.0	4	8.68	4.9%	3.8%	86	104
NB MID 5	38.38	5,888	15.23	1.65	1.3	4	6.60	4.9%	3.8%	82	104
SB MID 4	30.84	4,481	15.74	1.74	11.0	3	5.22	3.3%	2.9%	91	104
SB PM 4	61.68	4,788	15.21	1.21	9.5	4	4.84	4.9%	3.8%	86	104
SB MID 9	22.17	4,999	15.42	1.42	1.3	3	4.26	3.3%	2.9%	91	104
SB PM 6	39.00	4,888	16.13	2.13	8.0	2	4.26	2.4%	1.9%	86	104
NB AM 6	113.38	6,208	15.69	2.11	4.5	2	4.22	2.1%	1.9%	74	104
SB AM 7	35.44	4,774	15.30	1.30	1.7	3	3.90	3.3%	2.9%	91	104

Table 6. Summary of all clusters for all time periods and both directions, orde	ered by total
cluster delay impact (continuation).	

U.S. Department of Transportation

Office of the Assistant Secretary for Research and Technology

Intelligent Transportation Systems Joint Program Office

Cluster	Duration (min)	Volume (vehicle per hour [vph])	Travel Time (min)	Single Incident Delay Impact (min)	Incidents Per Period	Days in Cluster	Total Cluster Delay Impact (min)	Percent of Time Period with Incidents	Percent of Total Analysis Time Period	Total Days with Incidents in Analysis Period	Total Days in Analysis
NB MID 8	262.83	7,042	15.48	1.90	2.0	2	3.80	2.4%	1.9%	82	104
SB PM 7	693.80	4,662	17.17	3.17	10.0	1	3.17	1.2%	1.0%	86	104
NB AM 7	52.65	4,453	16.62	3.04	17.0	1	3.04	1.1%	1.0%	74	104
SB MID 12	27.73	5,193	15.45	1.45	6.0	2	2.90	2.2%	1.9%	91	104
NB PM 7	733.25	7,778	16.45	2.87	4.0	1	2.87	1.4%	1.0%	94	104
NB MID 9	659.00	6,507	16.33	2.75	4.0	1	2.75	1.2%	1.0%	82	104
SB MID 8	31.00	5,239	14.86	0.86	1.7	3	2.58	3.3%	2.9%	91	104
SB AM 6	229.00	6,350	16.52	2.52	1.0	1	2.52	1.1%	1.0%	91	104
SB PM 9	210.00	5,153	15.24	1.24	1.0	2	2.48	2.4%	1.9%	86	104
NB AM 8	34.75	1,880	15.95	2.37	8.0	1	2.37	1.1%	1.0%	74	104
SB AM 8	32.25	3,417	15.14	1.14	1.5	2	2.28	2.2%	1.9%	91	104
SB MID 10	753.33	4,397	15.88	1.88	3.0	1	1.88	1.1%	1.0%	91	104
SB AM 5	804.00	6,314	15.70	1.70	1.0	1	1.70	1.1%	1.0%	91	104
SB MID 11	203.00	4,651	15.55	1.55	1.0	1	1.55	1.1%	1.0%	91	104
SB MID 7	194.29	4,415	15.43	1.43	7.0	1	1.43	1.1%	1.0%	91	104
NB AM 9	184.00	5,723	14.92	1.34	1.0	1	1.34	1.1%	1.0%	74	104
NB PM 9	189.00	4,620	14.77	1.19	1.0	1	1.19	1.4%	1.0%	94	104
SB PM 8	274.47	5,280	14.95	0.95	15.0	1	0.95	1.2%	1.0%	86	104

Table 6. Summary of all clusters for all time periods and both directions, ordered by total cluster delay impact (continuation).

(Source: ICM Evaluation – San Diego Site Cluster Analysis – Daily Incident Probability, Battelle, 4/14/16, p. 9-10, unpublished.)

# **Incident Matching**

The focus for incident matching was on clusters for the two peak direction-periods (e.g., southboundmorning and northbound-afternoon). The incident matching process followed the prioritized criteria presented in Table 7 below, which shows how close the values must be to be considered a match. For the purposes of the evaluation, it was desirable that incidents from the baseline and post-deployment periods be matched that include variety, both in the cluster types shown above and type of DSS response plan that was implemented— including combinations of plans with and without traffic signal timing plan changes, ramp metering changes, and advanced traveler information systems (ATIS). As such, incident matching was sometimes performed such that an incident with a unique DSS response was matched to a cluster, while in other incidences unrepresented major clusters were matched to a post-deployment incident.

All incidents with an implementable DSS response were examined for a baseline period match. Some of the top direction-period clusters did not have an incident match with an implementable, actionable DSS response, so incidents with an information only DSS response were examined to identify candidate post-deployment incident direction-periods for the remaining high-impact clusters. The same characteristic matching criteria were used.

#### Table 7. Incident matching criteria.

Matching Element	Criteria	Priority
Number of Incidents per period	+/- 1	1
Average Peak Hourly Volume	+/- 5%	2
Average Incident Duration	+/- 20%	3

(Source: ICM Evaluation – San Diego Incident Matching, Battelle, 4/14/16, p. 4, unpublished.)

Table 8 shows the results of the incident matching process, which was used as the AMS scenarios in the alternatives analysis. Incident matches with actionable DSS response plans were identified for eight of the nine selected clusters. These incident matches represent a variety of DSS response plan types and clusters. In total, the best matching incident results include five incidents for the northbound afternoon peak direction-period and three for the southbound morning peak direction-period. The sum of all impacts across these top eight clusters represents the majority of impacts associated with the implementation of ICM in the I-15 corridor.

 Table 8. Summary of representative days/incidents in the most frequent/impactful clusters

 during the AM and PM peak periods.

Scenario	Date of Representative Day	DSS Plan Type Implemented	Total Cluster Day Impact (min)	Percent of Total Time Period	DSS Event ID	DSS Response ID
NB PM 1	6/16/15	Ramps, Signals, ATIS	41.8	16.3%	845922	30617
NB PM 2	6/09/15	Ramps, Signals	23.4	7.7%	842085	30451
NB PM 3	5/05/14	Ramps, Signals, ATIS	99.7	34.6%	853963	31039
NB PM 4	7/07/14	Ramps, Signals, ATIS	63.3	24.0%	639956	19536
NB PM 5	2/19/15	Signals, ATIS	18.7	2.9%	760369	28292
SB AM 1	5/27/15	Signals	49.9	27.9%	817649	30332
SB AM 2	2/09/15	Signals, ATIS	108.0	37.5%	754666	27929
SB AM 3	5/07/15	Ramps, Signals, ATIS	34.6	7.7%	804238	30028
Hypothetical	5/26/15	None. Express Lanes opened.	N/A	N/A	N/A	N/A

(Source: ICM Evaluation - San Diego Incident Matching, Battelle, 4/14/16, p. 5, unpublished.)

The purpose of the ninth scenario – Hypothetical – was to model the impacts of a DSS response for a scenario that did not occur during the evaluation period. The Evaluation, AMS and US DOT teams chose to evaluate the effectiveness of opening the Express Lanes to all traffic as part of the DSS response plans. An incident that warranted this response was selected and its duration and number of lanes closed was similar to the average incident that warranted a response.

It should be noted that a scenario which results in a response plan which opens the Express Lanes to all traffic is generally driven by rare conditions such as major incidents that involve significant disruptions to corridor performance. Under such conditions, the value that ICM brings is that it serves/provides coordinated response plans *in advance* of resorting to opening the Express Lanes. Operations and management of the I-15 ICM system is on-going and the I-15 ICM site team continues

to examine future enhancements to the system including how to enhance the system business rules based on lessons learned and response plan activations including major incidents that may require resorting to opening Express Lanes to all traffic.

Figure 19 shows where each of the nine AMS scenarios would lie on an operational condition dartboard with two axes: (1) time period (AM peak period or PM peak period), and (2) event type (congestion caused by bottleneck or congestion caused by incident). The size of each circle corresponds to the percent of total analysis time period for that scenario, except for the Hypothetical scenario, which does not have a representative percentage.



# Figure 19. Illustration. Operational condition dartboard categorized by time period and event type.

(Source: Cambridge Systematics, Inc., 2016)

The most impactful clusters of operational conditions were analyzed using the AMS tools, and then compared to the "without ICM" alternatives representing the transportation system without ICM turned on (but with pre-ICM corridor management practices in place). These comparisons facilitated the evaluation of impacts of the ICM system on the I-15 corridor. The identification of specific incidents or other events representing individual clusters were closely coordinated between the AMS, Evaluation and Volpe Center survey teams so as to ensure that event start and end times, impacts (such as number of lanes closed), and other characteristics were in complete agreement between the AMS, Evaluation and Survey team efforts.

# Model Calibration for Cluster-Representative Days

An iterative travel demand adjustment process was employed at the start of the analysis of each of the cluster-representative days, so that the model would reasonably represent the travel demand during each particular representative day. This process started by comparing observed versus modeled link volumes in the five links directly upstream of the primary incident during that day. Then

the origin-destination (OD) table was iteratively adjusted so that the sum of the modeled volumes in these links comes within 15 percent of the sum of the observed volumes in these links.

Table 9 below shows that the calibration process was able to bring the total percent error calculated for each cluster-representative day below the threshold value of 15 percent for all 36 hourly time periods. Therefore, the San Diego AMS model was deemed capable of representing the travel demand during each particular representative day. The resulting trip table was then used in modeling both the "with ICM" and "without ICM" scenarios.

Cluster	With Incidents and with ICM					
	4-5 PM	5-6 PM	6-7 PM	7-8 PM		
	1%	10%	6%	2%		
	2-3 PM	3-4 PM	4-5 PM	5-6 PM		
	-5%	1%	-12%	4%		
	2-3 PM	3-4 PM	4-5 PM	5-6 PM		
	-6%	-3%	-5%	2%		
	4-5 PM	5-6 PM	6-7 PM	7-8 PM		
	1%	3%	4%	-6%		
	3-4 PM	4-5 PM	5-6 PM	6-7 PM		
	3%	-1%	2%	-4%		
SB AM 1	6-7 AM	7-8 AM	8-9 AM	9-10 AM		
	-1%	-1%	3%	2%		
SB AM 2	6-7 AM	7-8 AM	8-9 AM	9-10 AM		
	6%	-7%	6%	3%		
SB AM 3	6-7 AM	7-8 AM	8-9 AM	9-10 AM		
	0%	-11%	-5%	-13%		
Hypothetical (based on	6-7 AM	7-8 AM	8-9 AM	9-10 AM		
SB AM 3 cluster)	0%	-11%	-5%	-13%		

### Table 9. Aggregated percent error of modeled vs. observed volume data for each clusterrepresentative day.

(Source: Cambridge Systematics, Inc., 2016.)

# **Traveler Survey Results**

The report titled *"Integrated Corridor Management Initiative: Traveler Response Panel Survey – San Diego"* presents findings from the ICM traveler behavior surveys, a set of panel surveys of I-15 corridor users, conducted before and after the deployment of ICM. The purpose of the surveys was to measure the impacts of the ICM initiative on travelers' use of real-time information (pre-trip and enroute), their travel behavior in the corridor, and their satisfaction with their corridor trips. In addition to surveying drivers about their general behavior in a baseline and endline survey, pulse surveys were administered immediately following incidents in the corridor to obtain a measure travelers' use of traveler information during incident conditions and its impact on their behavior. A survey of transit

riders was also conducted. Key findings are summarized below. (Please note, the contents of this section have been based on the US DOT report "Integrated Corridor Management Initiative: Traveler Response Panel Survey – San Diego", dated July 2016.)

### I-15 Drivers

Awareness and Use of Traveler Information Sources

In both the baseline and endline surveys, respondents were asked about their awareness and use of specific information sources – including websites, apps, alerts, social media, and telephone numbers. For a number of the sources, there is a significant increase in awareness; however this does not translate into increased <u>use</u>. That is, decreases in the percentage who had "never heard of" a source were accompanied by increases in the percentage who had "heard of, but never use" the source. It is worth noting that these shifts may be due, in part, to a learning effect. That is, through the process of being surveyed multiple times, respondents were repeatedly exposed to questions about information sources, and this constituted a form of "learning" about these sources.

In a separate measure that asked respondents to rate how informed they feel (using a seven point scale) about where to check for real time *traffic* information, there is a decrease in the proportion who feel they are uninformed (rating of 1, 2, or 3) – from 30 percent to 20 percent. However, there is no significant increase in the percent who feel informed (rating of 5, 6, or 7); rather more respondents reported being "somewhat" informed. The findings were similar with respect to how informed respondents feel about where to check for real time *transit* information.

The use of specific information sources generally remained stable across the baseline and endline surveys, with the exception of Google Maps (both the website and the app). The percent ever using the Google Maps website increased from 45 percent to 55 percent and similarly for the Google Maps app, the percent using it increased from 46 percent to 56 percent. However, there was no significant change in the frequency with which respondents consulted these or other information sources.

In addition, in the baseline and endline surveys, respondents were asked more generally about the frequency with which they consult real-time travel information for their morning and afternoon peak hour trips in the I-15 corridor– always, nearly always, sometimes, rarely, or never. While there was no change on this measure for morning peak trips, there was a slight increase in the frequency with which respondents consult information for their afternoon trips.

### Travel Behavior in the Corridor

In the baseline and endline surveys, traveler behavior changes were captured only very generally. For a list of possible changes (e.g., minor route changes, completely change route, leave for trip earlier, leave for trip later, switch to transit, telecommute) respondents were asked whether they had made the change – as a result of learning about traffic congestion on their route – in the last month, outside of the last month or never. The question was asked separately for travel behavior changes occurring pre-trip versus en-route. Overall, responses on these measures were consistent across the baseline and endline surveys. In response to learning about traffic congestion *prior to leaving for their trip*, respondents were most likely to start their trip earlier or to make route changes; about one-third of respondents reported making each of those changes in the last month. In addition, about one-fifth reported completely changing their route (17 percent) or starting their trip later (17 percent) in the last month. Relatively few respondents made other types of changes. In fact, in both the baseline and

endline surveys, three-quarters or more of respondents reported "never" having switched mode (e.g. taking transit or carpooling instead of driving), and two-thirds of respondents had never cancelled their trip or telecommuted instead of traveling. With respect to *en-route* changes in travel due to learning about traffic congestion, respondents were again most likely to change their route. While en-route, large majorities have never switched to transit or cancelled their trip – a finding that is consistent in both the baseline and endline surveys.

When asked about changes made in response to information at the trip level (across all pulse surveys), Table 10 and Figure 20 show a small increase in the morning peak in the proportion who made a minor route change based on pre-trip information (increased from 7 percent pre-ICM to 10 percent post-ICM). In addition, three percent fewer post-ICM travelers did nothing in response to congestion during the morning peak after learning from real-time information sources en-route. Table 11 and Figure 21 highlight travel behavior changes during the afternoon peak period. Two percent fewer post-ICM travelers who did nothing in response to congestion during the afternoon peak period after learning from real-time information sources pre-trip. Travelers surveyed by the Volpe Center stated several reasons for not making changes to their travel plans after learning about the delays, including the fact that alternatives were unlikely to reduce their trip home, incidents did not sound too severe, alternatives were not convenient, or they did not need to arrive at their destination by a particular time (more prevalent for travelers during the afternoon peak period).

# I-15 Transit Riders

### Use of Communication Devices and Real-Time Traveler Information

Transit riders are most likely to use their smartphones to acquire real-time traffic or transit information, and fewer respondents cited regular use of the radio, television or highway electronic signs. Indeed, in comparison to drivers, transit riders were less likely to regularly use the radio (which tends to focus on road conditions) or electronic highway signs. Like drivers, transit riders tend to consult Google Maps for their traffic information, as this source dominated both website and app use. Nearly half of transit riders consult the Google Maps app once a week or more often, and 10 percent or fewer consult any other app as frequently. Likewise, 39 percent of respondents consult the Google Maps website once a week or more often compared to 19 percent who consult TV and radio station websites and 14 percent who consult SigAlert. Other websites utilized once a week or more by at least 10 percent of respondents included 511 (10 percent) and MTS (10 percent).

Travel Changes	AM Peak	PM Peak	Reverse Peak Trips
Minor route changes			-
Pre-ICM	7%	12%	7%
Post-ICM	10%	10%	9%
Left earlier			
Pre-ICM	4%	5%	12%
Post-ICM	5%	6%	11%
Left later			
Pre-ICM	3%	4%	<0.5%
Post-ICM	3%	3%	4%
Completely different route			
Pre-ICM	2%	3%	2%
Post-ICM	3%	2%	3%
Changed stops			
Pre-ICM	<0.5%	<0.5%	6%
Post-ICM	1%	1%	<0.5%
Changed to MTS Express/Rapid			
Pre-ICM	<0.5%	<0.5%	<0.5%
Post-ICM	<0.5%	<0.5%	<0.5%
Changed to other transit			
Pre-ICM	<0.5%	<0.5%	<0.5%
Post-ICM	<0.5%	<0.5%	<0.5%
Carpooled			
Pre-ICM	1%	1%	2%
Post-ICM	<0.5%	1%	2%
Other			
Pre-ICM	1%	<0.5%	<0.5%
Post-ICM	<0.5%	<0.5%	<0.5%
No changes			
Pre-ICM	81%	76%	73%
Post-ICM	79%	78%	73%
Sample Size (Pre-ICM)	332	248	145
Sample Size (Post-ICM)	311	235	106

Table 10. I-15 pulse survey results – changes in travel plans based on pre-trip information.

(Source: Integrated Corridor Management Initiative: Traveler Response Panel Survey San Diego - Draft, Volpe National Transportation Systems Center, July 2016.)

Travel Change	AM Peak	PM Peak
Minor route changes		
Pre-ICM	11%	16%
Post-ICM	10%	16%
Completely different route		
Pre-ICM	6%	5%
Post-ICM	4%	3%
Changed stops		
Pre-ICM	1%	1%
Post-ICM	<0.5%	<0.5%
Used MTS Express		
Pre-ICM	<0.5%	<0.5%
Post-ICM	<0.5%	<0.5%
Used other transit		
Pre-ICM	<0.5%	<0.5%
Post-ICM	<0.5%	<0.5%
Other		
Pre-ICM	5%	1%
Post-ICM	6%	3%
No changes		
Pre-ICM	77%	78%
Post-ICM	80%	79%
Sample Size (Pre-ICM)	215	127
Sample Size (Post-ICM)	178	180

Table 11. I-15 pulse survey results – changes in travel plans based on en-route information.

(Source: Integrated Corridor Management Initiative: Traveler Response Panel Survey San Diego - Draft, Volpe National Transportation Systems Center, July 2016.)

About one-in-ten transit respondents said they always (8 percent) check information, and a similar proportion do so nearly always (13 percent) for their transit trips in the corridor; one-third hardly ever do so (34 percent), and another third never check information (32 percent). By comparison, 20 percent of drivers always check information and another 18 percent check nearly always.

Transit riders who reported "never" consulting information were asked why they don't. From a list potential reasons, respondents indicated that they do not check information because they have to use the same route no matter what (48 percent), they are not interested (42 percent), or they typically do not experience delays (39 percent). No respondents said that information is not available, information is not accurate or up-to-date, or information is not detailed enough.



Figure 20. Chart. I-15 pulse survey results – AM peak period travel plan changes. (Source: Cambridge Systematics, Inc., 2016.)





### Impact of Real Time Information on Travel Behavior

Similar to drivers, transit riders were asked a series of questions about the impact of real time traffic and transit information on their travel decisions, both before making a trip as well as during the trip. More specifically, transit riders were asked if they had made any of the following changes – prior to leaving for their trip – as a result of learning about traffic or transit problems:

- Start their trip earlier (29 percent in last month/23 percent not in the last month/46 percent never)
- Choose a different route to get to the transit station (13 percent in last month/19 percent not in the last month/63 percent never)
- Start trip later (20 percent in last month/18 percent not in the last month/58 percent never)
- Choose to drive or carpool instead of taking transit (18 percent in the last month/21 percent not in the last month/57 percent never)
- Choose a different station to get on Metropolitan Transit System (MTS) Rapid/Rapid Express (11 percent in last month/13 percent not in the last month/71 percent never)
- Choose a different station to get off MTS Rapid/Rapid Express (10 percent in last month/15 percent not in the last month/72 percent never)
- Decide to Telecommute (5 percent in last month/13 percent not in the last month/72 percent never)
- Cancel Trip (7 percent in last month/8 percent not in the last month/81 percent never)

For each change, no more than one-third of respondents had made the change in the last month, and a majority of transit riders had "never" made the change (with the exception of starting their trip earlier).

Similarly, respondents were asked if they had ever made any of the following changes while **en-route**, as a result of learning about traffic problems:

- Wait for a later bus due to overcrowding (22 percent in last month/20 percent not in the last month/55 percent never).
- Change route to the transit station (8 percent in last month/16 percent not in the last month/71 percent never).
- Get off MTS Rapid/Rapid Express at a different transit station (5 percent in last month/19 percent not in the last month/73 percent never).
- Use a different station to get on MTS Rapid/Rapid Express (6 percent in last month/13 percent not in the last month/79 percent never).
- Turn around and return to trip start (1 percent in last month/4 percent not in the last month/91 percent never).

While one-fifth of respondents have had to wait for a later train in the last month, relatively few respondents have made any of the other changes. Again, a majority of respondents indicate never having made each change while en-route (with the exception of waiting for a later bus due to

overcrowding). Based on the high level of satisfaction with their transit experience, the findings suggest that transit riders generally do not need to alter their trip behavior, as they generally do not face conditions that would require some change.

# **Analysis Results**

One of the main purposes of conducting the traveler response surveys is to determine the effectiveness of providing improved actionable traveler information. Key findings of the survey results are presented, which indicate the actual changes in travel behavior of I-15 drivers and transit riders attributed to the deployed ICM system.

Next, the estimated mobility and reliability performance measures produced through post-deployment AMS are shown. Daily and annual time savings from improved incident management are discussed. Potential travel time benefits for individual corridor users were converted into annual travel time savings and the "as-deployed" ICM system is stacked up against the "as-planned" ICM system.

### **Performance Measures**

In post-deployment AMS, eight scenarios were analyzed with- and without-ICM. Scenarios included both peak period directions. A ninth scenario was analyzed to compare the impact on mobility and reliability between two with-ICM scenarios, one with Express Lanes using dynamic variable pricing (normal operations) and one with Express Lanes open to all traffic during a severe incident. Since a deterministic microscopic model (Aimsun) was used for the simulation, post-processors were used to calculate the impacts on the reliability of travel time. Mobility measures for travel time, delay, and throughput per scenario can be seen in Table 12, Table 14, and Table 15. Reliability measures are presented in Table 15. The values shown in bold typeface represent results where the deployed ICM system had negative impacts on mobility, travel time reliability or variability.

### Mobility

### Travel Time

Table 12 shows mobility results in terms of daily person hours traveled (PHT). Reductions in PHT and travel time savings are seen in both peak directions (northbound or NB PM and southbound or SB AM). This analysis finding shows a significant travel time benefit resulting from deployment of ICM in the I-15 corridor. The NB PM 3 scenario saved 1,305 person hours traveled (-1.25 percent) in a single four hour window. On the other hand, the Hypothetical scenario shows an *increase* of 1,695 PHT (+2.02 percent) when Express Lanes are open to all traffic. The NB PM 2 and NB PM 5 scenarios also show a slight increase in PHT (+0.04% and +0.25%), which may be attributed to normal variation in the data.

When Express Lanes are under normal operation in the Hypothetical scenario, approximately 300 additional travelers shifted to BRT to avoid additional congestion resulting from the incident and approximately 1,700 additional travelers changed route during the four hour AM peak period to minimize incident impacts. Table 13 breaks down the number of vehicles and travelers that were on the general purpose lanes and Express Lanes both upstream and downstream of the severe incident during the Hypothetical scenario when Express Lanes were under normal operations. Downstream of the event, the I-15 Express Lanes are shown to carry approximately 68 percent more travelers per lane than the general purpose lane (13,735 travelers vs. 8,162 travelers). This results from the

combination of higher speeds and higher occupancies in the Express Lanes. So, when the Express Lanes are opened to all traffic during a major incident, this results in higher overall person hours of travel for the entire I-15 corridor which may decrease the appeal for travelers to shift to transit or HOV vehicles in order to take advantage of the travel time savings experienced on the Express Lanes during normal operations.

	-		Person Hours Traveled						
Scenario	Time Period Reported	Percent of Total Analysis Time Period	with ICM	without ICM	Difference (with ICM – without ICM)	Percent Change [(with ICM – without ICM) / without ICM]			
NB PM 1	4-8 PM	16.3%	78,774	79,320	-546	-0.69%			
NB PM 2	2-6 PM	7.7%	91,612	91,576	36	0.04%			
NB PM 3	2-6 PM	34.6%	103,437	104,742	-1,305	-1.25%			
NB PM 4	4-8 PM	24.0%	75,942	76,365	-424	-0.55%			
NB PM 5	3-7 AM	2.9%	84,570	84,356	214	0.25%			
SB AM 1	6-10 AM	27.9%	78,201	78,451	-250	-0.32%			
SB AM 2	6-10 AM	37.5%	75,413	76,189	-776	-1.02%			
SB AM 3	6-10 AM	7.7%	86,648	87,658	-1,010	-1.15%			
Hypothetical	6-10 AM	N/A	85,762	84,066	1,695	2.02%			

Table 12.	. Mobility	performance	measures –	daily P	Person Hou	irs Traveled.

Note for Hypothetical Scenario: Percent change is calculated using [(with ICM – without ICM) / without ICM] for all scenarios, except for the Hypothetical scenario, where percent change is calculated using [(Express Lanes with ICM open to all - Express Lanes with ICM Normal Operations) / Express Lanes with ICM Normal Operations].

(Source: Cambridge Systematics, Inc., 2016.)

In the Hypothetical scenario when Express Lanes were under normal operations, the average travel time for I-15 southbound travelers in the Express Lanes between Pomerado Road and SR-52 was approximately 2.3 minutes faster compared to travelers in the general purpose lanes (20.5 minutes vs. 22.8 minutes). Table 13 shows that the Express Lanes carried 38 buses, with a total of 1,520 passengers during the 6-10 AM time period of the Hypothetical scenario, resulting in 58.3 personhours of travel savings attributed to bus occupants alone. Travelers in single-occupancy vehicles saved 28.9 person-hours by opting to use the Express Lanes, while passengers in high-occupancy vehicles saved 965.8 person-hours.
Table 13. Mobility performance measures – traveler breakdown when Express Lanes were under normal operations during hypothetical scenario.

Upstream of Event:	General Pur	pose Lanes	Express Lanes	
Miramar Interchange	# of Vehicles	# of Travelers	# of Vehicles	# of Travelers
Single-Occupancy Vehicles	38,973	38,973	735	735
High-Occupancy Vehicles	1,394	3,485	9,216	23,040
Buses	-	-	38	1,520
Total	40,367	42,458	9,989	25,295
Average Vehicle Occupancy (AVO)	1.0	05	2.5	53
Number of Lanes	5		2	
Number of Travelers/Lane	8,492		12,648	

Downstream of Event:	General Pur	pose Lanes	Express Lanes		
Split between I-15 and SR-163	# of Vehicles	# of Travelers	# of Vehicles	# of Travelers	
Single-Occupancy Vehicles	39,097	39,097	754	754	
High-Occupancy Vehicles	686	1,715	10,078	25,195	
Buses	-	-	38	1,520	
Total	39,783	40,812	10,870	27,469	
Average Vehicle Occupancy (AVO)	1.03		2.53		
Number of Lanes	5		2		
Number of Travelers/Lane	8,162		13,735		

(Source: Cambridge Systematics, Inc., 2016.)

#### Delay

Delay results, normally represented by person-hours of delay, are reflected in the person hours traveled results shown in Table 12.

#### Throughput

Table 14 shows mobility results in terms of throughput, using person miles traveled (PMT) as a macroscopic measure of the general mobility of the corridor. The majority of the scenarios in both peak directions show no change, or an increase in PMT, which may be attributed to higher travel speeds resulting from the implementation of ICM. The Hypothetical scenario shows the biggest reduction in PMT – a *decrease* of 9,805 PMT (-0.33%), perhaps linked to the slower travel speeds experienced during the incident during the four hour PM peak period window.

Throughput across an entire period may not show much difference unless latent demand is taken into account. Even if network efficiencies were improved by ICM, it is possible that this did not translate into a significant increase in throughput if there was not much excess demand to begin with. In the case of San Diego I-15 ICM AMS the use of four-hour peak periods contains most of the network congestion (both spatially and temporally) as demonstrated by the model calibration speed contours in Chapter 7 of this report. Adding a reliability threshold or a throughput measure based on peak hourly

or 15-minute passenger throughput across a geographic screen-line could potentially be another measure to use in the future to determine the impact on PMT.

Time Scenario Period <sup>Reported</sup> T		Percent of	Person Miles Traveled				
		Total Analysis Time Period	with ICM	without ICM	Difference (with ICM – without ICM)	Percent Change [(with ICM – without ICM) / without ICM]	
NB PM 1	4-8 PM	16.3%	2,905,819	2,905,328	491	0.02%	
NB PM 2	2-6 PM	7.7%	3,446,769	3,445,499	1,270	0.04%	
NB PM 3	2-6 PM	34.6%	3,405,305	3,404,610	696	0.02%	
NB PM 4	4-8 PM	24.0%	2,921,317	2,917,474	3,843	0.13%	
NB PM 5	3-7 AM	2.9%	3,349,013	3,351,511	-2,498	-0.07%	
SB AM 1	6-10 AM	27.9%	2,972,239	2,971,470	769	0.03%	
SB AM 2	6-10 AM	37.5%	2,932,909	2,933,015	-106	0.00%	
SB AM 3	6-10 AM	7.7%	2,929,091	2,929,228	-137	0.00%	
Hypothetical	6-10 AM	N/A	2,923,030	2,932,835	-9,805	-0.33%	

### Table 14. Mobility performance measures – daily Person Miles Traveled.

Note for Hypothetical Scenario: Percent change is calculated using [(with ICM – without ICM) / without ICM] for all scenarios, except for the Hypothetical scenario, where percent change is calculated using [(Express Lanes with ICM open to all – Express Lanes with ICM Normal Operations) / Express Lanes with ICM Normal Operations].

(Source: Cambridge Systematics, Inc., 2016.)

### Reliability and Variability

The scenarios were aggregated into two categories based on direction and time period to show the trends in mobility and reliability performance measures. Reliability and variability capture the relative predictability of the public's travel time. Unlike mobility, which measures how many people are moving at what rate, the reliability and variability measures focus on how much mobility varies from day to day. Travel time reliability is reported in terms of changes in the Buffer Time and Planning Time Index, while travel time variability is reported in terms of changes in the standard deviation of average travel time. The Planning Time Index is the ratio of the 95<sup>th</sup> percent peak period travel time to the free flow travel time. The Buffer Time represents the additional time (or time cushion) that travelers must add to their average travel time when planning trips to ensure on-time arrival.

Figure 22 shows the impact that the San Diego ICM deployment is expected to have on travel time reliability in terms of changes to the buffer time. Buffer time is calculated as the difference between the 95<sup>th</sup> percentile travel time and the median travel time. For both aggregated scenarios (NB PM and SB AM), the figure shows an improvement (decrease) in the median travel time with ICM; this reinforces

the findings reported in the mobility performance results earlier in this chapter. An improvement (decrease) in the buffer time needed to ensure on-time arrival 95 percent of the time can also be seen in both aggregated scenarios.



Figure 22. Chart. ICM impacts on travel time reliability – buffer time. (Source: Cambridge Systematics, Inc., 2016.)

In Table 15, the standard deviation of travel time (linked to travel time variability) also shows an improvement (decrease) for both aggregated scenarios, following the same trend as the buffer time reliability performance measure. The Planning Time Index for the NB PM aggregated scenario shows the opposite trend. The calculation of the Planning Time Index follows the "Travel Time Reliability" equations (11) and (11a) in Appendix B: Performance Measure Calculation Using Simulation. The free flow travel time used in each individual scenario is the minimum travel time for each Origin-Destination (OD) pair. The free flow travel time used for each OD pair in the aggregated scenario is the minimum travel time for that OD pair among all scenarios within each aggregated scenario. All of the travel time values (i.e., median travel time, average travel time, and 95th percentile travel time) are weighted by the number of travelers in each ODTM, meaning travelers going to the same <u>OD</u>, in the same 15 <u>T</u>ime frame, using the same Mode.

The calculation of the standard deviation of travel time follows the "Variance in Travel Time" equations (14), (14a) and (14b) also in Appendix B. The calculation for standard deviation of travel time includes only completed trips during the four-hour peak period(s) so as to reduce potential bias caused by outliers. The calculation for Planning Time Index includes all trips completed or partially completed during the four-hour peak period. These differences in the trip population sets used along with complexities in establishing free flow travel times across aggregations of clusters (i.e. free flow travel time will differ across different origins and destinations and will also differ across clusters depending on the response plan activated and alternate routes taken to avoid congestion) influence the diverging trend in this situation. In general, travel time reliability and travel time variability are related but not identical - in situations where the "with ICM" and "without ICM" measures are close to each other, it is

possible that the differences in the two measures (Planning Time Index and standard deviation of travel time) go in opposite directions, as they are influenced differently by outliers.

For the NB PM aggregated scenario, it is interesting to see an increase in person miles traveled while person hours traveled decreased. Person miles traveled can increase when vehicles during the four hour PM peak period experience an increase in travel speeds, attributed to reduced congestion.

	Mobility		Reliability				Variability	
Aggregated Scenario	Person Miles Traveled	Person Hours Traveled	Median Travel Time (s)	Buffer Time (s)	95% Travel Time (s)	Planning Time Index	Average Travel Time (s)	Average Travel Time Standard Deviation (s)
NB PM								
Without ICM	3,180,084	90,164	612.3	132.5	744.8	4.41	635.0	168
With ICM	3,181,446	89,418	608.3	126.6	734.9	4.52	629.0	164
Difference	1,362	-745	-4.0	-5.9	-9.9	0.11	-6.0	-4
SB AM								
Without ICM	2,948,173	78,236	622.4	142.5	764.9	3.14	636.7	161
With ICM	2,946,781	77,578	617.2	138.8	756.0	3.08	632.2	158
Difference	-1,392	-658	-5.2	-3.7	-8.9	-0.06	-4.5	-3

# Table 15. Aggregated mobility, reliability, and variability performance results by direction & time period.

(Source: Cambridge Systematics, Inc., 2016.)

#### Response Plan Details for NB PM 3

This section presents the AMS results for one of the representative days (in this case for the NB PM 3 cluster) as an example representing the AMS work that was done for all nine cluster-representative day. NB PM 3 represents congestion caused by a bottleneck during the PM peak period on I-15 at Camino del Norte at Galatyn Parkway, as indicated by the pink segment in Figure 23. The incident matched to this scenario triggered an implementable response plan which intended vehicles to exit at Ted Williams Parkway to Pomerado North and reenter at the Pomerado on-ramp. To aid in this diversion, multiple changeable message signs displayed verbiage indicating "Slowing at Rancho Bernardo. Expect Delays" both before and after the incident and along State Route (SR) 52. Ramp meter rates were adjusted in order to maximize flow on I-15. The arterial traffic signal timing plans were adjusted for 16 signals on Pomerado North to accommodate the increase in traffic demand. As a result, 1,305 person hours of travel were saved overall, compared to a similar incident without an operating ICMS. The majority of travelers along the corridor (51.85 percent) experienced a decrease in travel time.



Figure 23. Diagram. Implementable response plan for peak direction scenario NB PM 3. (Source: Google Maps, Cambridge Systematics, Inc., 2016.)

### Summary

By multiplying the difference in PHT in Table 12 by each real scenario's percentage of total analysis time period and assuming 250 workdays in a year, the implementation of ICM is expected to produce savings of **267,850 annual person hours of travel**, a significant benefit that is generally consistent with the estimated travel time savings produced in the pre-deployment AMS analysis.

The scenarios in the northbound peak direction (NB PM 1, NB PM 2, NB PM 3, NB PM 4, and NB PM 5) represent 85.5 percent of the total time period, with a total of 511,762 travelers, while the scenarios in the southbound peak direction (SB AM 1, SB AM 2, and SB AM 3) represent 73.1 percent of the total time period, with a total of 441,733 travelers. This equates to a **cumulative annual travel time variability improvement of 188,816 hours** (also assuming 250 workdays in a year).

### Emissions, Fuel Consumption, and Cost Estimation

The I-15 Corridor AMS also produced model outputs for use by the Evaluation Contractor to estimate emissions and fuel consumption, associated with the deployment of ICM strategies. The data provided to the Evaluation Contractor included: a) link lengths, link characterization (freeway, major arterial, minor arterial) and average grade for all network links, and b) average hourly directional link volumes

and speeds for the I-15 freeway, Express Lanes, and arterials within the study area. The emissions analysis methodology will incorporate reference values to identify the emissions and fuel consumption rates based on variables, such as facility type, vehicle mix, and travel speed. The emissions and fuel consumption rates will be based on available sources. Emissions will be computed by pollutant, mode, and facility type. Fuel consumption will be computed by fuel type, mode, and facility type.

For the identified ICM strategies and based on input by the Evaluation Contractor, planning-level cost estimates will be prepared for life-cycle costs (capital, operating, and maintenance costs). Within each of the capital, O&M, and annualized cost estimates, the costs are further disaggregated to show the infrastructure and incremental costs. The costs will be estimated for each scenario and a benefit/cost ratio will be assigned to all the individual performance measures. The annualized benefits for each of the measures mentioned above will be calculated using cluster frequencies of occurrence as presented in Table 8.

# **Travel Time Beneficiaries**

Table 16 shows the percentage of travelers along the corridor who are expected to experience improved, worsened or unchanged travel times, as a result of the implemented ICMS. For example, in the NB PM 1 scenario, a DSS plan was implemented which included adaptive freeway ramp metering, arterial traffic signal coordination, and ATIS. As a result, 51.3 percent of travelers experienced a shorter travel time than usual, while 47.41 percent of travelers' travel time increased, and 1.29 percent of travelers saw no change in their travel time. Overall, 3.89 percent of travelers in the NB PM 1 scenario experienced a travel time net gain (overall travel time *reduction*). The table below shows that overall, more travelers experienced travel time benefits in the northbound and southbound peak scenarios (+2.65 percent and +4.01 percent, respectively). This finding is consistent with the finding presented earlier in this chapter showing overall aggregate improvements in travel time as a result of ICM implementation.

Scenario	Time Period Reported	Weighted Percent of Total Analysis Time Period	Improved Travel Time	Worsened Travel Time	Unchanged Travel Time	Improved % - Worsened %
NB PM 1	4-8 PM	19.0%	51.30%	47.41%	1.29%	3.89%
NB PM 2	2-6 PM	9.0%	48.94%	49.85%	1.21%	-0.91%
NB PM 3	2-6 PM	40.5%	51.85%	46.82%	1.33%	5.03%
NB PM 4	4-8 PM	28.1%	49.39%	49.39%	1.22%	-0.01%
NB PM 5	3-7 PM	3.4%	48.65%	49.81%	1.54%	-1.16%
NB PM (weighted avg)		100%	50.68%	48.03%	1.29%	2.65%
SB AM 1	6-10 AM	38.2%	50.02%	48.26%	1.72%	1.76%
SB AM 2	6-10 AM	51.3%	51.81%	46.23%	1.96%	5.58%
SB AM 3	6-10 AM	10.5%	51.36%	46.83%	1.81%	4.52%
SB AM (weighted avg)		100%	51.08%	47.07%	1.85%	4.01%
Hypothetical	6-10 AM	N/A	50.40%	47.92%	1.68%	2.48%
(Source: Cambridge Sv	stematics.	Inc., 2016.)				

#### Table 16. Travel time-based analysis results.

U.S. Department of Transportation

Office of the Assistant Secretary for Research and Technology Intelligent Transportation Systems Joint Program Office

# "As-Planned" vs. "As-Deployed" AMS Results

The main difference in the scope of the pre- and post-deployment analyses was that the time period of analysis was extended for post-deployment analysis. Instead of focusing only on the AM peak period (6-9 AM), both AM (6-10 AM) and PM (3-7 PM) peak periods were simulated. In pre-deployment AMS in order to estimate annual travel time savings, the AM peak period results were doubled to reflect the PM peak period as well. The ICM strategies remained the same in both pre- and post-deployment analysis. Another difference was that in pre-deployment analysis the TransModeler analysis tool was used, whereas in post-deployment analysis the Aimsun tool was used. Post-deployment AMS used incident details exactly as they occurred, while pre-deployment AMS modeled incidents based on aggregate statistical data on incident frequencies, durations and locations.

Pre-deployment AMS analysis of the "as-planned" ICM system estimated an annual savings of 245,594 vehicle hours of travel (Source: Alexiadis, V. and A. Armstrong, Integrated Corridor Management Modeling Results Report: Dallas, Minneapolis, and San Diego. No. FHWA-JPO-12-037. 2012.). This equates to approximately 282,433 person hours of travel, using an average vehicle occupancy rate of 1.15, while post-deployment AMS analysis of the "as-deployed" ICM system estimated a 267,850 person hour annual savings in the peak directions. The "as-deployed" ICM system was estimated to achieve nearly the same level of travel time savings as the "as-planned" ICM system. While this indicates that AMS is a powerful tool capable of simulating realistic mobility results under a set of ICM strategies, "as-deployed" analysis was more detailed, included several model improvements, and yielded additional insights on when and where benefits are accrued.

# **Overall Analysis Findings**

Overall, the I-15 corridor post-deployment AMS results show consistent travel time improvements in the two peak directions as a result of ICM implementation.

Summary of post-deployment AMS key findings:

- For the two peak directions (southbound AM and northbound PM), the expected daily travel time savings are 1,403 person hours of travel; expected annual savings are 267,850 person hours of travel (approximately 3.3% of delay along the corridor using 35 hours of annual freeway delay per traveler per SANDAG's San Diego Site: I-15 ICM Demonstration Project presentation at the American Public Transportation Association TransITech Conference on March 30, 2011 and an average daily traffic volume of 230,000 per Caltrans District 11 Interstate-15 Transportation Concept Summary published in June 2012).
- The Planning Time Index only improves during the southbound morning peak direction. The Buffer Time improves in both peak directions.
- Travel time variability improves during both peak directions; expected cumulative annual variability improvements is 188,816 hours.
- Overall, more travelers experienced travel time benefits in both peak directions, as well as in the hypothetical severe incident scenario. On average (weighted by percent of total time period of individual clusters), +2.65% of travelers experienced travel time benefits in the northbound PM aggregated scenario and +4.01% experienced travel time benefits in the southbound PM aggregated scenario.

- Overall, in six out of the eight analysis scenarios (excluding the hypothetical scenario), more travelers benefited from ICM, compared to the ones who did not.
- A hypothetical AMS exercise examined the potential benefit of opening the Express Lanes to all traffic during a major incident. No travel time benefits were found in AMS resulting from this potential action for the scenario analyzed.

# Benefits of Analysis, Modeling, and Simulation

Whereas the Dallas and San Diego Demonstration Sites followed the same ICM AMS framework, the San Diego I-15 corridor was found to experience approximately 10 times the amount of travel time savings and reliability improvements when compared to the Dallas U.S. 75 corridor. Both sites utilized similar ICM strategies to counteract bottlenecks and/or incidents, such as improved multimodal traveler information, coordinated traffic signal timings, mode shift and route diversion. One of the key differences was how the San Diego ICMS was designed around the concept of real-time adaptive response to changing traffic conditions, unlike the more rigid pre-defined expert rules-based response plans used in Dallas' ICMS. Based on the AMS results from both Demonstration Sites, it is possible that Dallas' fixed playbook system produced limitations which resulted in less significant benefits.

Coordination between and among local and regional agencies promoted data sharing from independently managed systems (transit, freeways, arterials, Express Lanes, incident management), which facilitated a real-time multimodal decision support system. The use of network state prediction in conjunction with real-time simulation and analysis of recommended response plans resulted in an effective decision support system, and was an innovative way to bring performance measures from planning to operations, a fundamental element of Transportation System Management and Operations (TSM&O). The ability of the San Diego ICMS to make automated traffic management decisions (e.g., automatically adjusting signals and ramps when recommended by the simulated evaluations) is the first of its kind in the nation. In the spirit of continuous improvement, 40 alternate route signs were recently installed on surface streets along the I-15 corridor throughout Escondido, Poway, and San Diego, which are used to guide vehicles through surface streets in the event of a major freeway incident.

The AMS methodology and results presented in this report demonstrate the feasibility of reducing congestion using ICM. The observations and benefits of AMS listed below highlight the lessons learned through this initiative.

The AMS process provides an invaluable framework for conducting assessments of the potential impacts and benefits of ICM strategies. The analytical complexity involved in these types of assessment goes far beyond what is typically required for more traditional types of transportation investments. The inclusion of multiple facility types (freeway and arterial) and multiple modes, combined with the potential influence of congestion pricing, complicates the analysis. The focus of the ICM strategies on nontypical operations scenarios (e.g., high demand, incidents, and inclement weather) adds further complexity to the assessment. The AMS procedures provide a pragmatic roadmap to guide practitioners through this complexity while not being too rigid to allow for flexibility in addressing project contingencies.

One major benefit of the ICM AMS methodology for San Diego was that it instigated the use of performance measures to inform and refine the response plans. This allowed AMS to provide insights

through measurable results, enabling stakeholders to determine how well the ICM system is working and whether it is accomplishing its goals. This is a major factor that can help agencies determine which transportation investments are worthwhile.

The ICM AMS methodology also builds in continuous improvement through the availability of new data sources. AMS allows agencies to "see around the corner", producing simulations of possible future conditions, allowing agencies to react proactively. AMS offers the flexibility of trying different combinations of traffic mitigation strategies, opening up an envelope of potential benefits, and can also provide more insight to realizing benefits. While models may take effort to set up initially, these models are not only used once. Managers can integrate the methodology with ICM decision support systems to facilitate predictive, real-time, and scenario-based operational decision-making. Overall, this helps agencies create better, more informed products and services.

For the ICM Demonstration Sites (Dallas and San Diego), the costs of developing and conducting AMS accounted for approximately five percent of the overall deployment budget. If the analysis was successful in better structuring the deployment to increase the efficiency of the ICM by a minimum of five percent, or reduced the risk of a deployment cost overrun of five percent or more, the investment in AMS paid for itself. The partners at the Demonstration Sites felt there was significant value in AMS which greatly outweighed the analysis costs. The AMS costs for the Demonstration Sites were likely proportionately higher than they would be in future analysis, due to the need to develop and refine new analysis methods and procedures. Hopefully, the best practices from this development procedure, highlighted in the Traffic Analysis Toolbox Volume XIII "Integrated Corridor Management Analysis, Modeling, and Simulation Guide", can be leveraged by subsequent practitioners to reduce the costs of conducting these activities.

The Demonstration Sites reported that using AMS not only improved their analysis capabilities for the ICM evaluation, but also served to enhance many existing tools and capabilities that can be applied to analysis of other investments. This analytical capital will enhance future analysis and increase confidence in the models. Some of the improvements reported by the Demonstration Sites included new software modules for analysis of multimodal assignment (transit), congestion pricing, Express Lanes, ramp metering, and real-time decision support systems. The Demonstration Sites also cited improved data quality control methods and enhanced model calibration procedures as examples of the continuous improvement benefits of the AMS process.

The ICM AMS approach is neither inexpensive, nor easy to accomplish. However, the value gained outweighs the expense and pays dividends throughout an ICM initiative. The lessons learned during this ICM initiative can be applied to other initiatives as well (e.g., Connected Vehicle Pilot Deployment Program). These closing thoughts highlight the benefits of successful ICM AMS planning and implementation:

- **Invest in the right strategies.** The methodology offers corridor managers a predictive forecasting capability that they lack today to help them determine which combinations of ICM strategies are likely to be most effective under which conditions.
- Invest with confidence. AMS allows corridor managers to "see around the corner" and discover optimum combinations of strategies as well as conflicts or unintended consequences inherent in certain combinations of strategies that would otherwise be unknowable before implementation.

- Improve the effectiveness/success of implementation. With AMS, corridor managers can understand in advance what questions to ask about their system and potential combinations of strategies to make any implementation more successful.
- AMS provides a long-term capability to corridor managers to **continually improve implementation** of ICM strategies based on experience.

# References

- Alexiadis, Vassili. Integrated Corridor Management Analysis, Modeling, and Simulation for the I-15 Corridor in San Diego, California Post-Deployment Analysis Plan. No. FHWA-JPO-16-393. 2016.
- Alexiadis, Vassili. Integrated Corridor Management Stage 3A Analysis, Modeling, and Simulation for the I-15 Corridor in San Diego, California Pre-Deployment AMS Assessment Report. No. FHWA-JPO-13-007. 2012.
- Alexiadis, Vassili and A. Armstrong. *Integrated Corridor Management Modeling Results Report: Dallas, Minneapolis, and San Diego.* No. FHWA-JPO-12-037. 2012.
- Alexiadis, Vassili, D. Sallman, and A. Armstrong. *Traffic Analysis Toolbox Volume XIII:* Integrated Corridor Management Analysis, Modeling, and Simulation Guide. No. FHWA-JPO-12-074. 2012.
- Dhindsa, Albinder, V. Alexiadis, A. Estrella, et al. Integrated Corridor Management I-15 San Diego, California Analysis Plan. No. FHWA-JPO-10-039. 2010.
- Dowling, R., A. Skabardonis, and V. Alexiadis. *Traffic Analysis Toolbox Volume III: Guidelines for Applying Traffic Microsimulation Modeling Software*. U.S. DOT-HRT-04-040, Federal Highway Administration, July 2004.
- Gonzalez, Paul, et al. Integrated Corridor Management: Implementation Guide and Lessons Learned. No. FHWA-JPO-12-075. 2012.

# **APPENDIX A. List of Acronyms**

AMS	Analysis, Modeling, and Simulation
ATIS	Advanced Traveler Information Systems
ATMS	Advanced Traffic Management System
BRT	Bus Rapid Transit
CATT	Center for Advanced Transportation Technology
CMS	Changeable Message Sign
CHP	California Highway Patrol
CPS	Congestion Pricing System
CO	Carbon Monoxide
CO <sub>2</sub>	Carbon Dioxide
CT-RAMP	Coordinated Travel – Regional Activity Based Modeling Platform
DAR	Direct Access Ramps
DSS	Decision Support System
FHWA	Federal Highway Administration
GPS	Global Positioning System
HC	Hydrocarbon
HOV	High-Occupancy Vehicle
I-15	Interstate 15
ICM	Integrated Corridor Management
ICMS	Integrated Corridor Management System
IMTMS	Intermodal Transportation Management System
ITS	Intelligent Transportation Systems
MD	Mid-Day
MTS	Metropolitan Transit System
NB	Northbound
NCTD	North County Transit District
NO <sub>x</sub>	Nitrogen Oxides
OD	Origin-Destination
PeMS	Performance Measures System
PHT	Person Hours Traveled
PM	Particulate Matter
PMT	Person Miles Traveled
RAMS	Regional Arterial Management System
RITIS	Regional Integrated Transportation Information System
RMIS	Ramp Metering Information System
SANDAG	San Diego Association of Governments
SB	Southbound
SO <sub>2</sub>	Sulfur Dioxide
SR	State Route
SUV	Single-Occupancy Vehicle
SUE	Stochastic User Equilibrium
	Transportation System Management and Operations
TEE	Transpondition System Management and Operations
100	Hanspon Simulation Systems
	USEI Equilibrium
03 001	U.S. Department of Transportation

V/C	Volume/Capacity
VHT	Vehicle Hours Traveled
VMT	Vehicle Miles Traveled
VOC	Volatile Organic Compound

# APPENDIX B. Performance Measure Calculation Using Simulation

This appendix describes the methodology used in calculating various performance measures for the ICM AMS as summarized in this report.

# Calculation Procedures for Key Integrated Corridor Performance Measures from Simulation Outputs

A core element of the Integrated Corridor Management (ICM) initiative is the identification and refinement of a set of key performance measures. These measures represent both the bottom-line for ICM strategy evaluation and define what "good" looks like among key corridor stakeholders. To date, the emphasis on performance-driven corridor management among the participating Pioneer sites has been on measures derived from observed data. In the Analysis, Modeling, and Simulation (AMS) phase of the effort, however, attention has turned to producing comparable measures derived from simulation outputs. This document provides a detailed process by which a set of key national measures of corridor performance can be calculated. It is the intent of the ICM program, and this document, that these processes will be implemented consistently in the three participating AMS sites applying the ICM AMS methodology.

This document provides a detailed description of how measures of *delay*, *travel time reliability* and *throughput* are calculated from simulation outputs. A brief discussion of travel time variance is also provided given that travel time variance measures are used in ICM-related benefit-cost calculations. The algorithmic approaches defined here are software independent, that is, this process can be implemented with outputs from any of the time-variant simulation tools utilized in the three participating ICM AMS sites. The document begins with a discussion of the calculation of travel time, which informs both a calculation of delay as well as travel time reliability. Next, we provide a discussion of how corridor throughput is defined and measured. The document concludes with a discussion of how these measures are used to make comparisons between system performance in the pre-ICM case and in one or more distinct post-ICM cases.

# **Travel Time**

Our basic unit of observation in calculating ICM-related performance measures is a trip i made

between an origin  $^{O}$  , finishing at a destination d , starting within a particular time interval  $^{\tau}$  using mode  $^{m}$ .

We record travel time from a single run of the simulation under operational conditions k for this unit of observation as  $t_i^k = t_{o,d,\tau',m}^k$ . In the case where multiple random seeds are varied, but the operational conditions are identical, this travel time represents an average for a single trip across the multiple runs. Also, note that this discussion of measures assumes that we are calculating measures for a single case (e.g., pre-ICM); later we will address comparisons between cases.

*Operational conditions* here refer to a specific set of simulation settings reflecting a specific travel demand pattern and collection of incidents derived from a cluster analysis of observed traffic count

data and incident data. An example of an operational condition would be an AM peak analysis with 5 percent higher than normal demand and a major arterial incident. Let k be a specific operational

condition and the set of all conditions K . Note that each condition has a probability of occurrence  $P_k$ 

 $\sum_{k} p_{k} = 1$ 

First, for this particular run(s) representing a specific operational condition, we calculate an average travel time for trips between the same o-d pair that begin in a particular time window. Let  $\tau$  represent

this interval, e.g., an interval between 6:30 AM and 6:45 AM and  $\mathbf{I}_{o,d,\tau,m}^{k}$  the set of  $n_{o,d,\tau,m}^{k}$  trips from

*O* to *d* starting in interval  $\tau$  under operational condition *k* using mode *m*. Note that  $\mathbf{I}_{o,d,\tau,m}^{k}$  is a collection of trips and  $n_{o,d,\tau,m}^{k}$  the scalar value indicating the number of trips contained in  $\mathbf{I}_{o,d,\tau,m}^{k}$ .

The set of all  $\tau$  of interest is the set T. For example, we may be interested in consistently calculating performance measures over all trips that begin in the 12 quarter-hour intervals between 6:00 AM and 9:00 AM.

The classification of travel mode may be determined independently at each site, but the breakdown should capture the combination of all modes utilized in making the trip. For example, one may choose to classify non-high-occupancy (HOV)-auto trips as a mode separately from non-HOV-auto/HOV/walk trips to track the performance of travelers utilizing park-and-ride facilities. However, any classification

of modes must be mutually exclusive and collectively exhaustive, that is,  $\bigcup_{m} \mathbf{I}_{o,d,\tau,m}^{k} = \mathbf{I}_{o,d,\tau}^{k}$  and  $\sum_{n} n_{o,d,\tau,m}^{k} = n_{o,d,\tau}^{k}$ 

The average travel time of trips with origin and destination by mode staring in this time interval is:

$$T_{o,d,\tau,m}^{k} = \frac{\sum_{i \in I_{o,d,\tau}^{k}} t_{i}^{k}}{n_{o,d,\tau,m}^{k}}$$
where  $n_{o,d,\tau,m}^{k} > 0$ . Let  $T_{o,d,\tau,m}^{k} = 0$  when  $n_{o,d,\tau,m}^{k} = 0$ .

The calculation of Equation 1 must also include some estimated travel time for trips that cannot reach their destinations by the end of the simulation period. Later in this document, we will discuss the method for estimating travel times for these trips still underway when the simulation ends.

Next, we calculate the average travel time for this same set of trips across all operational conditions, that is,  $\forall k \in K$ . Note that it is possible that we may have trips for some  $o, d, \tau, m$  under some conditions and no trips for the same  $o, d, \tau, m$  under other conditions. Let  $K'_{o,d,\tau,m}, K'_{o,d,\tau,m} \subseteq K$  be the subset of conditions where  $n^k_{o,d,\tau,m} > 0$ .

Equation 2 finds the average travel time by mode for all trips from O to d starting in interval  $\tau$  over all conditions where at least one trip is made,  $k \in K'_{o,d,\tau,m}$ :

$$T_{o,d,\tau,m} = \frac{\sum_{k \in K'_{o,d,\tau,m}} T^{k}_{o,d,\tau,m} p_{k}}{\sum_{k \in K'_{o,d,\tau,m}} p_{k}}$$
(2)

The average number of trips by mode from O to d starting in interval  $\mathcal{T}$  over all conditions  $k \in K$ :  $n_{o,d,\tau,m} = \sum_{k \in K} n_{o,d,\tau,m}^{k} p_{k}$ (2a)

Combining across modes, the average travel time of trips from  $^{O}$  to  $^{d}$  starting in interval  $^{\tau}$  under operational condition  $^{k}$ :

$$T_{o,d,\tau}^{k} = \frac{\sum_{m} T_{o,d,\tau,m}^{k} n_{o,d,\tau,m}^{k}}{n_{o,d,\tau}^{k}}$$
where  $n_{o,d,\tau}^{k} > 0$ . Let  $T_{o,d,\tau}^{k} = 0$  when  $n_{o,d,\tau}^{k} = 0$ . (3)

The average travel time for all trips from O to d starting in interval  $\tau$  under  $K'_{o,d,\tau}$  the subset of and  $n^k_{o,d,\tau} > 0$   $K'_{o,d,\tau} \subseteq K$ .

$$T_{o,d,\tau} = \frac{\sum_{k \in K'_{o,d,\tau}} T_{o,d,\tau}^k p_k}{\sum_{k \in K'_{o,d,\tau}} p_k}$$
(4)

The average number of trips from O to d starting in interval  $\tau$  over all conditions  $k \in K$ :

$$n_{o,d,\tau} = \sum_{k \in K} n_{o,d,\tau}^{\kappa} p_k \tag{4a}$$

Equation 5 defines the trip-weighted average travel time of the system across all  $o, d, \tau$ :

$$\overline{T} = \frac{\sum_{\forall o,d,\tau} T_{o,d,\tau} \, n_{o,d,\tau}}{\sum_{\forall o,d,\tau} n_{o,d,\tau}}$$
(5)

### Delay

Delay can be broadly defined as travel time in excess of some *subjective minimum* travel time threshold. Often, discussions of delay focus solely on roadway-only travel focus on either travel time at posted speeds or 85<sup>th</sup> percentile speeds. Delay for ICM must be defined differently since ICM

explicitly includes multimodal corridor performance. Instead, we directly identify delay at the o,d,m

level by deriving a zero-delay threshold  $T^0_{o,d,m}$ , considering travel times observed across all operating

conditions  $\forall k \in K$  and all time intervals  $\forall \tau \in T$ . The zero-delay threshold for each o-d pair by mode is calculated looking across all operating conditions and all time intervals:

$$T_{o,d,m}^{0} = \min_{k \in K, \tau \in T} \left\{ T_{o,d,\tau,m}^{k} \right\}$$
(6)

In some cases, the cluster analysis will group low-demand, nonincident conditions into a large, highprobability operational condition. In this case, it is possible that a notionally "low" demand pattern will still produce significant congestion in the corridor, particularly in a peak period analysis.

For this reason, the minimum threshold may also be calculated as the travel time derived in the pre-ICM case under a substantially reduced demand pattern with no incidents or weather impacts. The reduced demand pattern should produce enough trips to generate travel time statistics by mode for

every set of trips from O to d starting in interval  $\tau$  (i.e.,  $n_{o,d,\tau,m}^0 > 0 \forall o, d, \tau, m$ ). At the same time, the reduced demand should generate no volume-related congestion in the network.

Alternatively,  $T_{o,d,m}^0$  may be estimated directly from model inputs. For consistency, however, the travel time associated with these thresholds should include expected transfer time between modes and unsaturated signal delay as in the case where a low-demand pattern is used to drive a zero-delay model run.

From our previous calculation of travel time in Equation 1, recall the average travel time of all trips traversing the network from origin O to destination d starting in time interval  $\tau$  using mode m under operational condition k,  $T^k_{o,d,\tau,m}$ . Using zero-delay thresholds  $T^0_{o,d,\tau,m}$ , calculate average trip delay under condition k for each

 $o, d, \tau, m_{\perp}$ 

$$D_{o,d,\tau,m}^{k} = max \left[ T_{o,d,\tau,m}^{k} - T_{o,d,\tau,m}^{0}, 0 \right]$$
(7)

Combining across all operational conditions, calculate the average delay for each  $o, d, \tau, m$  over K'

Combining across modes, the average delay for trips from  $^{O}$  to  $^{d}$  starting in interval  $^{ au}$  :

$$D_{o,d,\tau} = \frac{\sum_{m} D_{o,d,\tau,m} \, n_{o,d,\tau,m}}{n_{o,d,\tau}}$$
(8)

where 
$$n_{o,d,\tau} > 0$$
. Let  $D_{o,d,\tau} = 0$  when  $n_{o,d,\tau} = 0$ .

Systemwide average trip delay (Equation 9):

$$D = \frac{\sum_{\forall o,d,\tau} D_{o,d,\tau} n_{o,d,\tau}}{\sum_{\forall o,d,\tau} n_{o,d,\tau}}$$
(9)

Aggregating this average delay over all trips produces total system delay (Equation 10):

$$\widehat{D} = \sum_{\forall o, d, \tau} D_{o, d, \tau} n_{o, d, \tau}$$
(10)

### Travel Time Reliability

Corridor reliability measures are inherently measures of outlier travel times experienced by a traveler making the same (or similar) trip over many days and operational conditions. We have already defined and organized travel time measures from the simulation with respect to trips from O to d starting in interval  $\mathcal{T}$  over using mode m for all conditions  $k \in K$ . Just as in the case of the subjective notion of delay as travel time in excess of some minimum threshold, the notion of what reliable travel is depends on a *relative maximum* acceptable travel time threshold. For the ICM AMS effort, as in many studies with a travel reliability measure, a threshold based on the 95<sup>th</sup> percentile travel time is

selected. Note that this percentile is calculated considering travel times for similar trips (i.e.,  $o, d, \tau, m$ ) with respect to travel time variation induced by changes in operational conditions  $k \in K$ .

To identify the 95<sup>th</sup> percentile travel time, first we generate an ordered list of travel times for each  $o, d, \tau, m$  across all operating conditions:

$$\mathbf{T}_{o,d,\tau,m} = \left[ T_{o,d,\tau,m}^{1}, T_{o,d,\tau,m}^{2}, \cdots, T_{o,d,\tau,m}^{J} \right] \text{ where } T_{o,d,\tau,m}^{J} \le T_{o,d,\tau,m}^{J+1} \text{ for all } j = 1 \cdots J .$$
(11)

The 95<sup>th</sup> percentile travel time from this list is identified using the probabilities associated with each operational condition.

$$T_{o,d,\tau,m}^{[95]} = T_{o,d,\tau,m}^{j}$$
 where  $\sum_{k=1}^{J} p_k = 0.95$  (11a)

Note the array of travel times  $\mathbf{T}_{o,d,\tau,m}$  represents levels on a linear step-function. This implies that if 17.4 minutes is the travel time associated with an operational condition occupying the 92<sup>nd</sup> through 98<sup>th</sup> travel time percentile, we simply use the 17.4-minute travel time as the 95<sup>th</sup> percentile value. Also note that the specific operational conditions under which the 95<sup>th</sup> percentile travel time is found will

vary among  $o, d, \tau, m$ . For example, a major freeway incident creates congestion and high travel times for trips that originate upstream of the incident location, but creates free flowing and uncongested conditions for trips that originate downstream of the incident location.

Equation 12 defines planning time index for each  ${}^{O,d,\tau,m}$ , the ratio of the 95<sup>th</sup> percentile travel time to the zero-delay travel time for trips from  ${}^{O}$  to  ${}^{d}$  starting in interval  ${}^{\tau}$  using mode  ${}^{m}$  over all conditions  $k \in K$ :

$$\rho_{o,d,\tau,m} = \frac{T_{o,d,\tau,m}^{[95]}}{T_{o,d,\tau,m}^0}$$
(12)

Equation 12a defines planning time index by  $o, d, \tau$  across all modes:

$$\rho_{o,d,\tau} = \frac{\sum_{m} \rho_{o,d,\tau,m} \, n_{o,d,\tau,m}}{n_{o,d,\tau}}$$
(12a)

Average systemwide planning time index considers all  $o, d, \tau$ , weighted average by trip volume:

$$\rho = \frac{\sum_{\forall o,d,\tau} \rho_{o,d,\tau} \, n_{o,d,\tau}}{\sum_{\forall o,d,\tau} n_{o,d,\tau}}$$
(13)

We may also be interested in trip-weighted planning time index within a mode across all  $^{o,d, au}$ :

$$\rho_{m} = \frac{\sum_{\forall o,d,\tau} \rho_{o,d,\tau} n_{o,d,\tau,m}}{\sum_{\forall o,d,\tau} n_{o,d,\tau}}$$
(13a)

### Variance in Travel Time

Variance in travel time can be calculated in a variety of ways. The key here is that some care must be taken to isolate the specific variation of interest. Additionally, as variance is strongly influenced by outliers, in order to eliminate any potential bias introduced into the variance of travel times resulting from the estimation of a fulfilled travel time for incomplete travelers at the end of the simulation period, the variance calculation should be restricted to completed travelers defined as set  $\ddot{\mathbf{I}}_{a,d,\tau}^{k}$  consisting of

 $\ddot{n}_{o,d,\tau}^{k}$  trips. While the inclusion of the fulfilled incomplete travelers' travel times in the other

performance measures may be influenced by the same bias, the nature of the variance calculation magnifies the effects of that potential bias. This effect may be more significant in larger models where the calibration and validation efforts must be focused on the primary corridor or study area.

Given this, the variance in travel time among members of the same origin, destination, and time interval in a single run is:

$$V_{o,d,\tau}^{k} = \frac{\sum_{i \in \mathbf{I}_{o,d,\tau}^{k}} \left( \dot{t}_{i}^{k} - \ddot{T}_{o,d,\tau}^{k} \right)^{2}}{\ddot{n}_{o,d,\tau}^{k} - 1}$$
(14)

Recall  $K'_{o,d,\tau}$ ,  $K'_{o,d,\tau} \subseteq K$  as the subset of conditions where  $\ddot{n}^k_{o,d,\tau} > 0$ . The variance of travel time for each  $o, d, \tau$  under all operation conditions is then defined as:

$$V_{o,d,\tau} = \frac{\sum_{k \in K'_{o,d,\tau}} V_{o,d,\tau}^k P_k}{\sum_{k \in K'_{o,d,\tau}} P_k}$$
(14a)

The average variance among all  $o, d, \tau$  is a weighted average of the variances:

$$V = \frac{\sum_{\forall o,d,\tau} V_{o,d,\tau} \ddot{n}_{o,d,\tau}}{\sum_{\forall o,d,\tau} \ddot{n}_{o,d,\tau}}$$
(14b)

# Throughput

The role of a throughput measure in ICM is to capture the primary product of the transportation system: travel. Particularly in peak periods, the capability of the transportation infrastructure to operate at a high level of efficiency is reduced. One of the goals of ICM is to manage the various networks (freeway, arterial, transit) cooperatively to deliver a higher level of realized system capacity in peak periods. While throughput (e.g., vehicles per lane per hour) is a well-established traffic engineering point measure (that is, in a single location), there is no consensus on a systemwide analog measure. In the ICM AMS effort, we use the term *corridor throughput* to describe a class of measures used to characterize the capability of the integrated transportation system to efficiently and effectively transport travelers. We do not consider freight throughput in these calculations, although this could be revisited at a later date.

In order to support throughput measures, additional trip data need to be generated as simulation outputs. For each trip i made between an origin O, finishing at a destination d, starting at a

particular time  $\tau'$  we obtain from the simulation the travel time  $t_{o,d,\tau'}^{\star}$  and a distance traveled  $s_{o,d,\tau'}^{\star}$ . In some cases, trip-level outputs from the simulation are only available at a vehicle level, so some trips

may have multiple passengers associated with that trip (e.g., in the case of carpool travel). Let  $\chi^{\hat{r}}_{o,d,r'}$  represent the number of travelers associated with a particular trip record.

Passenger-miles traveled (PMT) are accumulated using a process similar to travel time. First, we convert individual trip PMT into an average PMT for trips from origin  $^{O}$  to destination d with a trip start in time interval  $\tau$ .

$$X_{o,d,\tau}^{k} = \frac{\sum_{i \in I_{o,d,\tau}^{k}} s_{i}^{k} x_{i}^{k}}{n_{o,d,\tau}^{k}}$$
(15)

For trips that cannot be completed before the end of the simulation, see the following section for the estimation of total trip distance.

Equation 16 finds the average PMT for all trips from O to d starting in interval  $\tau$  over all operational conditions  $k \in K$ :

$$X_{o,d,\tau} = \sum_{k \in K} X_{o,d,\tau}^k p_k \tag{16}$$

Equation 17 defines the aggregate PMT across all  $o, d, \tau$ :

$$X = \sum_{\forall o,d,\tau} X_{o,d,\tau} \, n_{o,d,\tau} \tag{17}$$

Restricting the calculation of measures to selected cohorts is also relevant to the calculation of delay and travel time reliability measures. Although peak periods vary among the AMS sites in terms of the onset and duration of congestion, a consistent set of trips that contribute to measure calculation (others simply run interference) should be identified. As in the case of the throughput time cut-off point, US DOT may wish to prescribe specific times in the future.

# Estimation of Travel Times and Travel Distance for Incomplete Trips

Trips that cannot complete their trips by the time that the simulation ends are still included in the calculation of all delay and travel time calculations. Our approach is to estimate total travel time including any additional time that would be required to complete the trip given the average speed of travel.

First, let  $\mathbf{\ddot{I}}_{o,d,\tau}^{0}$  be the set of  $\ddot{n}_{o,d,\tau}^{0}$  trips from origin O, destination d starting a trip in time interval  $\tau$  that can be completed under the low-demand operational condition used to identify the zero-delay travel times.

The average distance traveled over these trips is:

$$\ddot{X}^{0}_{o,d,\tau} = \frac{\sum_{i \in \vec{I}^{0}_{o,d,\tau}} s_{i}}{\ddot{n}^{0}_{o,d,\tau}}$$
(24)

Note: If  $\ddot{n}_{o,d,\tau}^0 = 0$  then  $\ddot{X}_{o,d,\tau}^0$  is indeterminate. In this case, find  $\tau'$ , the closest time interval such arg min.

that 
$$\tau' = \frac{\tau' - \tau}{\tau'} |\tau' - \tau|$$
 where  $n_{o,d,\tau'}^0 > 0$ . Approximate  $\ddot{X}_{o,d,\tau}^0$  using  $\ddot{X}_{o,d,\tau'}^0$ .

Next, let  $\mathbf{\tilde{I}}_{o,d,\tau}^{k}$  be the set trips from origin O, destination d starting a trip in time interval  $\tau$  that *cannot* be completed under operational condition k. For all  $i \in \mathbf{\tilde{I}}_{o,d,\tau}^{k}$ , let  $\mathbf{\tilde{x}}_{i}^{k}$  be the distance traveled on the trip i up to the point where the simulation ends, and let  $\mathbf{\tilde{t}}_{i}^{k}$  the travel time on trip i up to the point where the simulation ends for a trip that cannot be completed is expressed in Equation 25:

$$\tilde{v}_i^k = \frac{\tilde{x}_i^k}{\tilde{t}_i^k} \tag{25}$$

Estimated total trip travel time for a trip that cannot be completed before the simulation ends is the accumulated travel time plus the time to travel the remaining distance at average trip speed:

$$t_{i}^{k} = \overline{t}_{i}^{k} + max \left\{ \frac{\left( \ddot{X}_{o,d,\tau}^{0} - \overline{x}_{i}^{k} \right)}{\overline{v}_{i}^{k}}, 0 \right\}$$

$$x_{i}^{k} = max \left\{ \ddot{X}_{o,d,\tau}^{0}, \overline{x}_{i}^{k} \right\}$$

$$(26)$$

$$(27)$$

# **Comparing Pre-ICM and Post-ICM Cases**

All of the travel time and throughput measure calculation procedures defined above are conducted under a single set of simulation settings reflecting a specific set of corridor management policies, technologies and strategies (here referred to as a *case*, but often called an *alternative*). The complete suite of delay, travel time reliability and throughput measures are calculated independently for each case (e.g., Pre-ICM or Post-ICM). Comparisons of the resulting measures are then made to characterize corridor performance under each case.

# **Comparing Observed and Simulated Performance Measures**

These few key measures have been defined in detail for national consistency across all AMS sites. Sites have also identified measures. This document has dealt in detail with the calculation of measures from simulation outputs. However, the calculation of comparable measures using observed data demands an equivalent level of detailed attention. These observed measures will be critical in the AMS effort to validate modeling accuracy and in performance measurement in the demonstration phase. Because of the nature of the simulation output, the modeling analyst is able to resolve and track performance at a level of detail that is not available to an analyst working with field counts, speeds and transit passenger-counter outputs. However, it is the responsibility of the site and the AMS contractor to ensure that these measures are similar in intent, if not in precise calculation. In many cases, the simulation tools or their basic outputs can be manipulated to produce measures quite comparable with field data. An example of this is in throughput calculation, where a site may wish to pursue a screenline passenger throughput measure from field data. In addition to the system-level throughput measures detailed above, the simulation model can be configured to produce passenger-weighted counts across the same screenline to match the field throughput measure.

U.S. Department of Transportation ITS Joint Program Office-HOIT 1200 New Jersey Avenue, SE Washington, DC 20590

Toll-Free "Help Line" 866-367-7487 www.its.dot.gov

FHWA-JPO-16-403

